Genetic Analysis of Prostate Cancer



Genetic Analysis of Prostate Cancer

Dirk van Alewijk

Financial support for the printing of this thesis was obtained by kind contributions from: *Dutch Cancer Society - Koningin Wilhelmina Fonds Erasmus University Rotterdam Department of Pathology - Josephine Nefkens Institute PamGene International B.V.*

Cover: A ceiling view of "La Grande Arche" - La Défense in Paris, illustrating a different point of view regarding the inactivating mechanisms of tumour suppressor genes in prostate cancer. During my traineeship in Paris, I appreciated the elegance and splendour of how artwork can be embedded into architectural designs, which initiated my fascination for detail photography. I have tried to use this to give varying insights to each of the different chapter covers.

ISBN: 90-9017578-4 Printed by Optima Grafische Communicatie, Rotterdam

Genetic Analysis of Prostate Cancer

Genetische analyse van prostaatkanker

Proefschrift

ter verkrijging van de graad van doctor aan de Erasmus Universiteit Rotterdam op gezag van de Rector Magnificus

Prof.dr. S.W.J. Lamberts

en volgens besluit van het College voor Promoties. De openbare verdediging zal plaatsvinden op

woensdag 17 december 2003 om 11.45 uur

door

Dirk Cornelis Jerrefaas Gelde van Alewijk Geboren te Tilburg Promotiecommissie:

Promotoren:	Prof.dr.ir. J. Trapman Prof.dr. J.W. Oosterhuis
Overige leden:	Prof.dr. Th.H. van der Kwast Prof.dr. C.H. Bangma Prof.dr. C.J. Cornelisse

Deze studie werd bewerkt binnen de afdeling Pathologie van de Faculteit der Geneeskunde en Gezondheidswetenschappen, Erasmus Universiteit te Rotterdam. Het onderzoek werd financieel gesteund door de Nederlandse Kanker Bestrijding -Koningin Wilhelmina Fonds (project nr. EUR96-1375).

Voor "mijn vrouwen"

INDEX

List of abbreviations		
List of gene	abbreviations	9
Chapter 1	General Introduction	11
. 1.1	Prostate cancer	13
	1.1.1 Morphology, function, and development of the prostate	13
	1.1.2 The epidemiology of prostate cancer	14
	1.1.3 Clinicopathological aspects of prostate cancer	15
	1.1.3.1 Histological typing of prostate cancer	15
	1.1.3.2 Diagnosis and therapy of prostate cancer	16
1.2	Genetics of tumour development	17
	1.2.1 Oncogenes and tumour suppressor genes	17
	1.2.2 A genetic model of tumour development	18
1.3	Genetic alterations in prostate cancer	19
	1.3.1 Methods to detect genes involved in cancer	19
	1.3.2 Hereditary prostate cancer	
	1.3.2.1 Definition	
	1.3.2.2 Hereditary prostate cancer loci and genes	23
	1.3.3 Sporadic prostate cancer	24
	1.3.3.1 Genome-wide search for sporadic prostate cancer loci	24
	1.3.3.2 Analysis of individual chromosomes	
	1.3.3.2.1 Chromosome 8	
	1.3.3.2.2 Chromosome 10	35
1.4	Scope of this thesis	43
Chapter 2	Frequent inactivation of <i>PTEN</i> in prostate cancer cell lines and xenografts	45
Chapter 3	Chromosome 10 alterations in prostate cancer xenografts and cell lines:	
cano	didate tumour suppressor genes and the PTEN flanking genes	57
Chapter 4	Identification of a homozygous deletion at 8n12-21 in a human prostate	
cand	cer xenograft	79
Chapter5	Genetic analysis of chromosome interval 8p12-p21 in prostate cancer	
xenc PC1	33 disrupting the Werner Syndrome gene	93
Chantor F	Preliminary data on corponing for homozygous deletions, and	
	ression and mutation analysis of candidate tumour suppressor goods on	
chro	mosome 8n12-n21 in prostate cancer	115
Child		
Chapter 7	General Discussion	125

Appendix		131
A1	Table I. HPC loci and candidate genes	133
A2	Table II. Details on chromosomal losses observed by CGH analysis in PrCa	134
A3	Table III. Details on chromosomal gains observed by CGH analysis in PrCa	
A4	Table IV. Candidate tumour suppressor genes on chromosome 8 analysed in PrCa	136
A5	Table V. Candidate tumour suppressor genes on chromosome 10 analysed in PrCa	138
A6	A high-resolution physical and transcript map of chromosome 8p12-p21	140
References		149
Summary / S	amenvatting	167
Dankwoord		176
List of publica	tions	179
Curriculum Vi	tae	181

List of abbreviations

AR	Androgen Receptor
BPH	Benign Prostate Hyperplasia
CGH	Comparative Genome Hybridization
DRE	Digital Rectal Examination
EST	Expressed Sequence Tag
FISH	Fluorescence in situ Hybridization
FITC	Fluoresceine Isothiocyanate
HD	Homozygous Deletion
HPC	Hereditary prostate cancer
JPS	Juvenile Polyposis Syndrome
LOH	Loss of Heterozygosity
LUTS	Lower Urinal Tract Symptoms
MMR	Mismatch Repair
PCR	Polymerase Chain Reaction
PIN	Prostatic Intraepithelial Neoplasia
PrCa	prostate cancer
PSA	prostate specific antigen
PSP94	Prostatic Secretory Protein 94
RCC	Renal Cell Carcinoma
RDA	Representational Difference Analysis
RFLP	Restriction Fragment Length Polymorphism
RT	Reverse Transcriptase
SCID	Severe Combined Immunodeficient
SKY	Multicolour Spectral Karyotyping
SNP	Single Nucleotide Polymorphisms
SSCP	Single Stranded Conformation Polymorphism
STS	Sequence Tagged Site
TNM	Tumour, Nodes and Metastasis
TSG	Tumour Suppressor Gene
TRAMP	Transgenic Adenocarcinoma of Mouse Prostate
TRUS	Transrectal Ultrasound
TUR	Transurethral resection
UGM	Urogenital sinus Mesenchyme
WS	Werner Syndrome

List of gene abbreviations

1D12A	pre-T/NK cell associated protein
AK057533	
AKT1	v-akt murine thymoma viral oncogene homologue 1 (RAC / PKB)
AKT3	v-akt murine thymoma viral oncogene homologue 3
ANXA7	Annexin A7 (ANX7)
APC	Adenomatosis Polyposis Coli
AR	Androgen Recentor
RAD	BCI 2-Antagonist of cell Death
BCL2	B-cell CLL / wmphoma 2
BC-PC	Brain Cancer and Prostate Cancer
BDCA2	Broact Cancer 2
CASPO	press Calicel 2
CASFS	CD44 antigon
CD44 CDC42	Coll Division Ovela 40
00042	Cell Division Cycle 42
CDHI	Cadherin T (E-cadherin)
CDKN1A	Cyclin-Dependent Kinase inhibitor 1A (WAF1 / P21 / CIP1)
CDKIN1B	Cyclin-Dependent Kinase inhibitor 1B (P27 / KIP1)
CDKN2A	Cyclin-Dependent Kinase inhibitor 2A (P16/MTS1)
CLU	Clusterin
COPEB	Core Promoter Element Binding protein (<i>KLF6</i>)
CREB	cAMP Responsive Element Binding protein
CUGBP2	CUG triplet repeat, RNA Binding Protein 2
CYP17A1	Cytochrome P450, family 17, subfamily A, polypeptide 1 (CYP17)
CYP3A4	Cytochrome P450, family 3, subfamily A, polypeptide 4
DCC	Deleted in Colon Cancer
DMBT1	Deleted in Malignant Brain Tumours 1
EIF3S3	Eukaryotic translation Initiation Factor 3, Subunit 3 gamma, 40kDa
ELAC2	elaC homolog 2
eNOS	Endothelial Nitric Oxide Synthase
EPHX2	Epoxide Hydrolase 2
FLJ11218	hypothetical protein FLJ11218
FLJ30656	hypothetical protein FLJ30656
FOXO3A	Forkhead box O3A (FKHRL1)
FRAP1	FK506 binding protein 12-Bapamycin Associated Protein 1 (<i>mTOB</i>)
GATA3	GATA hinding protein 3
GNRH1	Gonadotropin-Beleasing Hormone 1 (I HBH)
GSK3	Glycogen Synthase Kinase 3 alpha (beta)
GSTM1	Glutathione S-Transferase M1
GTE2E2	General Transcription Eactor IIE polypentide 2 heta 3/kDa
HRAS	v-Ha-ras Hanvey Bat Sarcoma viral oncorrene homologue
HSD2B1	Hydroxy_dolta_5_Storoid Dobydrogonaso, 3 Bota, and storoid dolta_isomoraso 1
100001	Hydroxy delta 5 Staroid Dehydroganasa, 3 Data- and staroid delta isomarasa 2
H3D3D2 UDC1	Hydroxy-della-5-Steroid Dellydrogenase, 5 Dela- and Steroid della-isomerase 2
	Hereditary Prostate Cancer 1
	Hereditary Prostate Cancer 2
HPCX	Hereditary Prostate Cancer X
IKK	IkappaB Kinase
KRAS	v-Ki-ras2 Kirsten Rat Sarcoma 2 viral oncogene homologue
LAPSER1	
LZTS1	Leucine Zipper, putative Tumour Suppressor 1 (FEZ1)
MADH4	Mothers Against Decapentaplegic Homologue 4 (SMAD4 / DPC4)
MDM2	Mouse Double Minute 2 homologue
MINPP1	Multiple Inositol Polyphosphate histidine Phosphatase, 1
MSMB	Microseminoprotein, Beta- (PSP-94)

MSR1	Macrophage Scavenger Receptor 1
MXI1	MAX Interacting protein 1
MYC	v-myc avian Myelocytomatosis viral oncogene homologue
N33	Putative prostate cancer tumor suppressor
NEFL	Neurofilament, Light polypeptide 68kDa
NKX3-1	NK3 transcription factor related, locus 1
NRAS	Neuroblastoma RAS viral (v-ras) oncogene homologue
PAP	Prostatic Acid Phosphatase
PAPSS2	3'-Phosphoadenosine 5'-Phosphosulfate Synthase 2
PDGF	Platelet Derived Growth Factor alpha (beta)
PDGFRL	Platelet Derived Growth Factor Receptor-Like (PRLTS)
PG1	Putative prostate cancer
PI3K	Phosphoinositide-3-Kinase, class 2, alpha polypeptide (beta)
POLB	DNA Polymerase Beta
PTEN	Phosphatase and Tensin homologue (MMAC1 / TEP1)
PTK2	Protein Tyrosine Kinase 2 (<i>FAK2</i>)
RB1	Retinoblastoma 1
RBPMS	RNA Binding Protein with Multiple Splicing
RNASEL	Ribonuclease L (2',5'-oligoisoadenylate synthetase-dependent)
SHC	(SRC Homology 2 domain Containing) transforming protein 1
SKP2	S-phase Kinase-associated Protein 2
SRD5A2	Steroid-5-alpha-Reductase, Alpha polypeptide 2
TGFB1	Transforming Growth Factor, Beta 1
TNFRSF6	Tumour Necrosis Factor Receptor Superfamily, member 6
TP53	Tumour Protein p53 (<i>P53</i>)
TP73	Tumour protein p73 (<i>P73</i>)
UNC5D	Unc-5 homologue D
VDR	Vitamin D (1,25- dihydroxyvitamin D3) Receptor
WNT	Wingless-type MMTV integration site family
WRN	Werner Syndrome

CHAPTER 1

GENERAL INTRODUCTION



1.1 PROSTATE CANCER

1.1.1 Morphology, function, and development of the prostate

The human prostate has the size of a chestnut and envelops the urethra as it exits the bladder, below the bladder neck. It is the largest of the male accessory sex glands, which also include the seminal vesicles, and bulbourethral gland. The prostate is composed of glandular structures, which are tightly fused within a common capsule. The prostate can be subdivided in a transitional, a central, and a peripheral zone (Figure 1.1) (McNeal 1997). Histologically, the glandular structures are complex arrays of luminal structures. Except for the larger ducts near the urethra, the ductal-acinar system is lined by highly differentiated secretory columnar epithelial cells layered on undifferentiated nonsecretory basal epithelial cells, thus forming a continuous layer adjacent to the basement membrane (Figure 1.2). The glandular structures are supported by stroma. The stromal compartment encompasses all cellular and extracellular elements outside the epithelial basement membrane and includes smooth muscle cells, blood vessels, lymphatic tissues, nerves, and fibroblasts embedded in a loose collagenous matrix (reviewed by Cunha et al. 1987)(McNeal 1997).



Figure 1.1. Anatomy of the human prostate that shows the location of the peripheral zone (PZ), transition zone (TZ), and bladder neck (bn) (McNeal 1997).



Figure 1.2. Hematoxylin-eosin stained tissue section of a normal prostate. Columnar secretory epithelial cells (arrows) are layered on nonsecretory basal epithelial cells (arrow heads), which are scattered along the basement membrane.

14 | Chapter 1

The function of the prostate is to add components to the ejaculate. Prostatic secretions contain many different proteins. A major component is prostate specific antigen (PSA), which belongs to the family of kallikrein-like proteases. Two other major secreted proteins are prostatic acid phosphatase (PAP), and microseminoprotein beta (MSMB / PSP-94; Lilia and Abrahamsson 1988). PSA is known to be involved in semen liquefaction (Lilia 1985; Lundwall and Lilia 1987), but the physiological functions of PAP and MSMB are still unknown. Other products of the prostate are kallikreins, plasminogenactivator, pepsinogen II, metalloproteases, caseinolytic and gelatinolytic activities, and $Zn-\alpha 2$ -Glycoprotein (reviewed by Wilson 1995).

The development of the prostate occurs via mesenchymal-epithelial interactions in which the urogenital sinus mesenchyme (UGM) is thought to induce ductal morphogenesis, and epithelial growth. This prostatic development is induced by androgens. In early development, androgenic effects on epithelial development are mainly elicited via androgen receptors (AR) in the mesenchymal cells. The stromal-epithelial interactions in the prostate continue into adulthood (reviewed by Cunha 1994; Marker et al. 2003).

1.1.2 The epidemiology of prostate cancer

Prostate cancer (PrCa) is the most frequently diagnosed cancer in men, and the second leading cause of male cancer death in Western countries (Greenlee et al. 2001). Out of all cancers in men, the incidence of PrCa increases most rapidly with age (Carter and Coffey 1990; Jacobsen et al. 1995; Potosky et al. 1995; Merrill et al. 1997; Post et al. 1998). From autopsy studies and from studies of radical cystoprostatectomy specimens removed for bladder cancer, it is known that approximately 10% of men in their fifties to 70% of men in their eighties, harbour microscopic foci of well or moderately differentiated adenocarcinoma in the prostate, although there is no clinical evidence of PrCa (latent PrCa; reviewed by Gittes 1991; Scardino et al. 1992; Matzkin et al. 1994; Ohori and Scardino 1994). Extrapolated to 50-year-old American men, it is estimated that about one in four latent PrCas becomes clinically manifest (Carter and Coffey 1990; Coffey 1993), whereas about one of three patients with clinical PrCa eventually die of the disease (Scardino et al. 1992).

Although the incidence of latent PrCa might be comparable worldwide, there are large variations in clinical incidence and mortality among different geographical and ethnic populations (Carter and Coffey 1990). The highest incidence rate was found in African-American men in the United States, followed by Caucasian men in Northern America, Western Europe and Australia. The lowest rates were found among Chinese and Japanese men (Coffey 1993). Some of these population differences may be attributed to life style and dietary factors (Hanley et al. 1995; Uchida et al. 1995).

1.1.3 Clinicopathological aspects of prostate cancer

1.1.3.1 Histological typing of prostate cancer

Most PrCa are adenocarcinomas that arise in the peripheral zone of the prostate. It is generally accepted that PrCa originates from stem cells or progenitor cells of the luminal epithelial cells (De Marzo et al. 1998). The clinical stages that are distinguished in PrCa are pre-malignant, locally confined, infiltrating, and metastatic disease (Figure 1.3). The pre-malignant stage is characterized by severe cytological aberrations in the luminal epithelial cells with preservation of the glandular architecture. These changes are also known as prostatic intraepithelial neoplasia (PIN), and are subdivided into a low- and high-grade form. High-grade PIN is considered to be the immediate precursor of locally confined PrCa (Bostwick et al. 1997). Local prostate cancer is characterized by invasive tumour growth without penetration through the prostatic capsule. In locally advanced cancer, the tumour infiltrates the tissues surrounding the prostate after penetration of the capsule. When PrCa metastasizes, secondary deposits are found in lymphatic and bone tissue. In late stages, PrCa metastases can be found in skin, lung, and liver tissues. Two commonly used clinical staging systems for PrCa are the Whitemore-Jewlett classification (Catalona and Avioli 1987; Gittes 1991), and the TNM (tumour, nodes and metastasis) staging system (Schroder et al. 1992).



Figure 1.3. Morphological phases of PrCa progression.

1.1.3.2 Diagnosis and therapy of prostate cancer

Until the late eighties in the 20th century PrCa was clinically established by digital rectal examination (DRE) in patients visiting the physician with clinical symptoms. About 15 years ago, transrectal ultrasound (TRUS) imaging and the serum prostate specific antigen (PSA) test were introduced. The final diagnosis is made on the basis of histological examination of a prostate biopsy. The serum PSA test resulted in the detection of PrCa at an earlier stage. However, it has to be realized that a positive PSA test can not only be due to PrCa, but many other explanations are possible. Before the introduction of the serum-PSA-test, a patient with PrCa would typically present in an advanced stage with micturation problems, lower urinary tract symptoms (LUTS), and / or complaints caused by metastases, for instance bone pain. The screening with the serum PSA test resulted in a dramatic increase in the incidence of diagnosed PrCa.

Treatment decisions for PrCa depend on tumour characteristics (stage, pathological grade, and serum PSA levels), the patient's life expectancy, and quality versus quantity of life trade-offs. Locally confined PrCa can be surgically removed by radical prostatectomy, or may be treated by radiation therapy. A third option is observation, also known as "watchful waiting". Infiltrative or metastatic PrCa can be treated by endocrine therapy. This treatment modality includes surgical castration, chemical castration (e.g. gonadotropin-releasing hormone 1 (GNRH1) analogues), or therapy with anti-androgens. Endocrine therapy is based on the function of androgens in prostate growth and maintenance of its structural integrity. Androgen depletion dramatically affects normal prostate development, secretory function, maintenance of structural integrity, and ultimately results in apoptosis (Cunha et al. 1987; Lee 1997; Gao and Isaacs 1998). The majority of PrCas show a partial or complete response to endocrine therapy. However, after a mean period of less than 2 years, PrCa inevitably progresses to an androgen-independent stage. Transurethral resection (TUR) can be

carried out as a palliative treatment to diminish micturation complaints and lower urinary tract symptoms. Radiation therapy may be administered for symptomatic metastases.

1.2 GENETICS OF TUMOUR DEVELOPMENT

1.2.1 Oncogenes and tumour suppressor genes

Cell growth, programmed cell death (apoptosis), and differentiation are strictly controlled by the interplay of complex molecular mechanisms (Hanahan and Weinberg 2000). Deregulation of these mechanisms can lead to aberrant cell growth, and eventually to the formation of a cancer. During the last decades, many theories regarding the molecular mechanisms of tumourigenesis have been proposed. Fundamentally, all theories postulate that cancer is a genetic disease. In a tumour cell, genes involved in the regulation of cell growth, apoptosis, differentiation and repair of DNA damage are mutated such that the encoded protein no longer functions properly. However, it is becoming more and more clear that altered gene expression also contributes to tumour growth.

Historically, genes involved in tumourigenesis have been subdivided into oncogenes and tumour suppressor genes (reviewed by Bishop 1987; Hahn and Weinberg 2002). Oncogenes are derived from cellular genes (proto-oncogenes) that have become oncogenic by activating mutations or by overexpression. If activated, (proto-)oncogenes function in cell growth stimulation or inhibition of apoptosis. Overexpression of an oncogene might be caused by amplification of the gene or by gene rearrangement (e.g. chromosomal translocation). In the latter situation, the gene is expressed under control of a different promoter. The aberrant allele of an oncogene is functionally dominant over the wild-type allele. The physiological function of many tumour suppressor genes is inhibition of cell growth or stimulation of apoptosis. In a tumour, a tumour suppressor gene might be inactivated by mutation, deletion or by down-regulation of expression by promoter methylation or other mechanisms. Classically, the wild-type allele is functionally dominant over an inactivated allele. Therefore, both alleles must be inactivated to produce an oncogenic effect. This hypothesis is generally known as Knudson's "two hit" theory (Knudson 1971). More recent insights in the malfunctioning of tumour suppressor genes provide evidence that also loss of one functional allele (haplo-insufficiency) and epigenetics can contribute to tumourigenesis.

Although most human cancers result from somatic genetic alterations in tumour cells, rare cancers can run in families with a hereditary pattern. In these families individuals are prone to develop tumours because of an inherited genetic alteration.

1.2.2 A genetic model of tumour development

Tumourigenesis is a multi-step process, as first proposed by Armitage and Doll (1954). The very first molecular events that underlie the initiation and progressive growth of human tumours were identified in the late eighties of the last century (reviewed by Bishop 1987; Weinberg 1989; Hanahan and Weinberg 2000). These observations constituted an essential contribution to a multi-step model for tumourigenesis. The first model describing a cascade of genetic changes was proposed for colorectal carcinogenesis (Fearon et al. 1990). This multi-step model described the sequential molecular and morphological transition of normal colon epithelium via hyperproliferating epithelium, benign adenoma, and malignant carcinoma to metastatic colon cancer. Furthermore, Fearon and Vogelstein (1990) assumed that (1) cancer involves the mutational activation of oncogenes and the inactivation of tumour suppressor genes, (2) the mutational activation or inactivation of four to five genes is required, and (3) although the genetic alterations may occur in a preferred sequence, the accumulation of changes rather than their order determines the biological properties of the tumour. These stepwise genetic alterations are currently well accepted as a general model for tumourigenesis.

The colorectal tumour progression model has been modified by Kinzler and Vogelstein (1997 and 1998), and Lengauer et al. (1998). They proposed that tumour suppressor genes could be sub-divided into three functionally distinct groups. Genes in these 3 groups are called: gatekeepers, caretakers, and landscapers. Gatekeepers are genes directly involved in neoplastic growth and prevent cancer by inhibiting cell growth or promoting apoptosis. Examples of this group of tumour suppressor genes are *RB1*, and *APC*. Restoration of a missing gatekeeper gene can result in suppression of neoplastic growth. Caretakers are indirectly involved in neoplastic growth. The normal function of these genes is to prevent DNA damage, chromosomal instability, and abnormal genetic recombination. Many genes with such a function are known nowadays

(Wood et al. 2001). Representatives of this group are the mismatch repair genes. Inactivation of certain caretaker genes results in a continuously increasing mutation rate in individual genes (domino effect) and thus genetic instability. Restoration of a defective caretaker function to cancer cells has no direct effect on neoplastic growth, because its loss has resulted in secondary mutations. Thus, inactivation of a caretakergene is not an immediate trigger to neoplastic growth. It is the increased mutation rate that results in a higher probability of mutational inactivation of gatekeeper genes.

Cancer research has predominantly focussed on cancer cells and their genomic properties. However, the importance of intercellular signalling between diverse cell types in cancer is also being recognized. Tumours are complex tissues in which mutant cells may have recruited normal non-neoplastic cells to serve as active collaborators in their neoplastic growth (Skobe and Fusenig 1998). Genes with a role in this partnership are designated "landscapers". An example of this group of tumour suppressor genes is *MADH4 (SMAD4 / DPC4)*, which showed germline mutations in some JPS patients. Surprisingly, bi-allelic *MADH4* inactivation was only found in the stromal cells of the hamartomatous polyps of these patients (Koyama et al. 1999). Landscaper genes are thought to promote the gatekeeper- or caretaker-pathways in at least some tumour types (Jacoby et al. 1997).

1.3 GENETIC ALTERATIONS IN PROSTATE CANCER

The colorectal cancer models described above might be exemplary to tumour development in other human epithelial tumours, including PrCa. In PrCa, however, many of the molecular mechanisms underlying tumour development remain to be discovered. The focus of this thesis was to identify tumour suppressor genes that are involved in PrCa.

1.3.1 Methods to detect genes involved in cancer

In general, the genetic strategy to identify genes involved in cancer is (1) genome-wide screening, (2) chromosome mapping, (3) gene mapping, (4) mutation analysis of candidate genes, and (5) functional screening of candidate genes. Genes involved in hereditary and sporadic cancers are identified by different approaches. Hereditary cancer genes are genome-wide and fine-mapped by genetic linkage

analysis. This positional cloning technique is a powerful tool for the identification of inherited disease genes based solely on their location within the genome. Many currently known cancer susceptibility genes have originally been identified by linkage analysis of cancer-prone families. In a linkage analysis, a genotype of such families is generated using polymorphic markers scattered throughout the genome. Subsequent correlation of the genotype with the segregation pattern of the disease within the family may reveal linkage of the phenotype with a particular chromosomal locus.

Genes involved in sporadic cancers are genome-wide mapped by characteristic large chromosomal alterations (losses, gains, and translocations) in the tumour genomes. Chromosomal losses are an indication for the localization of a tumour suppressor gene, whereas gains and translocations are an indication for the location of an oncogene. These larger chromosomal alterations may be identified by karyotyping (Brothman et al. 1994), multicolour spectral karyotyping (SKY; Schrock et al. 1996; Speicher et al. 1996), comparative genomic hybridisation (CGH; Kallioniemi et al. 1992), and array CGH (Pinkel et al. 1998). Karyotyping provides a complete scan of banded mitotic chromosomes. It is suitable for the identification of all kinds of chromosomal alterations, including aneuploidy and iso-chromosome formation. However, the technique requires metaphase cells, and the identification of chromosomal aberrations may thus necessitate culturing of the tumour cells. Apart from difficulties like slowly growing cultures and cell selection, the induction of genetic alterations during culturing may lead to misinterpretation of the results. SKY is a modernized karyotyping showing each chromosome in a specific colour, which is particularly helpful for the identification of chromosomal translocations. For the identification of both losses and gains preference is given to CGH, in which normal and tumour DNA are each labelled with a different fluorescent dye and then hybridised to a normal metaphase chromosome spread. For CGH, DNA from frozen tumour samples as well as formalin-fixed, paraffin-embedded tumours can be utilized. Larger chromosomal aberrations can easily be characterized, but alterations like translocations, aneuploidy, and iso-chromosome formation can not be identified because CGH is based on the utilization of total tumour DNA. Nowadays, array CGH becomes a robust alternative for CGH analysis. Although allelotypical differences still can not be distinguished by this new technology it maps losses and gains more precisely than CGH.

Chromosomal regions of loss and gain can be mapped by chromosome transfer and genetically mapped in detail by analysis of allelic imbalances. Chromosome transfer provides a host cell with a complete set of genes from an added chromosome or chromosomal region, including the putative tumour suppressor gene or oncogene that induces the host-cell to lose or gain tumourigenic characteristics (Ichikawa et al. 1991). Chromosome transfer is a functional approach for the mapping of cancer genes to larger chromosomal regions. A potential problem of the technique is that transferred chromosome transfer can be misinterpreted due to the properties of the *in vitro* system used or the occurrence of mutations or deletions of multiple genes in the host tumour cell line.

Analysis of allelic imbalances does not easily discriminate between losses and gains of genomic DNA in the tumour cell. It can be carried out by analysis of restriction fragment length polymorphisms (RFLP; Vogelstein et al. 1989), analysis of polymorphic microsatellites (Weber and May 1989), or single nucleotide polymorphisms (SNP). Initially, analysis of allelic imbalance was done by RFLP analysis, which is a Southern blotting detection method for specific restricted genomic fragments. RFLP analysis is laborious and requires large amounts of tumour DNA sample. Much faster alternatives that require smaller amounts of sample DNA are microsatellite and SNP analysis. These two techniques are PCR-based detection methods of polymorphic mono-, di- and tri-nucleotide repeats (so-called microsatellites), and of single nucleotide polymorphisms (SNPs), respectively. SNPs can also be identified by oligo-hybridisation, allowing a high-throughput approach using microarrays. Allelic imbalance of informative markers specify the boundaries of a region of importance. Besides application in analysis of allelic imbalance, microsatellite analysis can also be informative in detecting genetic instability, which can be visualized by unstable microsatellite lengths (MSI, microsatellite instability).

A next step in fine-mapping regions of genetic alterations can be (fluorescent) *in situ* hybridisation ((F)ISH; Pinkel et al. 1986; Brothman et al. 1999) or screening for homozygous deletions. ISH is based on hybridisation of labelled DNA probes to chromosome spreads, cells in metaphase or interphase nuclei in histological sections. FISH is very powerful in detection of high-level amplifications.

Homozygous deletions are rare, but if found, they can map a tumour suppressor gene to a region of less than 1 Mbp. Homozygous deletions can be

22 | Chapter 1

identified in a defined chromosomal area by the absence of a PCR product, utilizing a large set of markers (Kamb et al. 1994; Hahn et al. 1996). A genome-wide search for homozygous deletions can be done by representational difference analysis (RDA). RDA identifies a small genomic fragment derived from normal DNA, which is not subtracted after reiterated hybridisation steps against tumour DNA (Lisitsyn and Wigler 1993). A search for homozygous deletions in DNA from primary tumour specimens can be severely complicated by the presence of contaminating DNA from normal cells. Nevertheless, a homozygous deletion narrows down a tumour suppressor gene region enormously.

Conclusive evidence for the identification of a tumour suppressor gene or oncogene can be given by mutational analyses and functional studies. In particular, for tumour suppressor genes and mismatch repair (MMR) genes, strongest evidence is provided by sequence analysis, showing a deletion, frame-shift or nonsense mutation. Functional evidence can be obtained by gene transfer or anti-sense RNA (RNAi) approaches. However, the efficacy of the introduced or inactivated gene will depend on the properties of the host cell line studied.

Alteration of gene expression may be a direct result from a mutation in a cancer gene. However, it also can be the result of an epigenetic event such as promoter hypermethylation. Therefore, down- or up-regulation of a candidate tumour gene provides indirect evidence for a role in cancer as it might be the result and not the cause of tumourigenesis.

1.3.2 Hereditary prostate cancer

1.3.2.1 Definition

Characteristic patterns for inherited cancer syndromes are familial aggregation of tumours and early disease onset (Bishop and Kiemeney 1997). Such patterns are found in nearly one quarter of all PrCa (reviewed by Carter et al. 1993; Narod 1999; Bratt 2000a, b; Karayi et al. 2000). By definition PrCa is familial if two first-degree relatives, father and son or brothers, develop the disease. Hereditary prostate carcinoma (HPC) refers to a subtype of familial PrCa in which there are either (1) at least three first degree cases of PrCa, (2) at least three successive generations of either maternal or paternal lineages with PrCa, or (3) a cluster of two relatives diagnosed with PrCa before the age of 55 years (Carter et al. 1993). Ninety percent of PrCa cases are believed to be sporadic, the remaining ten percent displays a hereditary component (Carter et al. 1993; Keetch et al. 1996; Bratt et al. 1999).

1.3.2.2 Hereditary prostate cancer loci and genes

Various genome-wide scans of DNA samples from HPC families demonstrated linkage to loci on chromosomes 1p36, 1q42.2-q43, 1q24-q25, 8p22-p23, 11, 16q23, 17p11, 20q13, and Xq27-q28 (see Appendix A1 - Table I for references and locus names). Most linkages were confirmed in other studies. Some groups however, were unable to demonstrate linkage to a specific locus in their set of high-risk families. It is difficult to determine to which extent confounding factors, like late age of onset, lack of distinguishing features between the hereditary and sporadic forms of the disease, and the difficulty in identifying HPC families, might have affected these investigations (Smith et al. 1996).

Besides the age of onset, a few clinical features appear characteristic for HPC families. In families with linkage to 1p36, PrCa was found to aggregate with brain cancer (Gibbs et al. 1999; Xu et al. 2001b). In a subset of Swedish families, Gronberg et al. (2000) observed significant aggregation of PrCa with breast carcinoma and/or gastric carcinoma (Gronberg et al. 2000). In families with linkage to 1q24-q25, significantly more aggressive PrCa were diagnosed (Goode et al. 2001).

Although many HPC loci have been described, only a few high-penetrant susceptible genes were proposed (see Appendix A1 - Table I). In two HPC families that showed linkage to the HPC1 locus at chromosome 1q24-q25, a nonsense mutation and a mutation in an initiation codon of the *RNASEL* gene were found to segregate (Carpten et al. 2002). In three HPC families that showed linkage to the HPC2 locus at chromosome 17p11, a frame-shift and two missense mutations of the *ELAC2* gene were found to segregate (Rokman et al. 2001; Tavtigian et al. 2001; Wang et al. 2001). Besides these families, no additional *RNASEL* and *ELAC2* mutations or polymorphisms were found in HPC1 or HPC2 families, respectively. Subsequent studies of *ELAC2* failed to demonstrate an increased risk of a polymorphism described before (Rokman et al. 2001; Suarez et al. 2001; Vesprini et al. 2001; Wang et al. 2001; Xu et al. 2001a). Therefore, the relative importance of the *ELAC2* polymorphisms remains under debate. Other proposed high-penetrant susceptible genes are *HSD3B1* and *B2*, *TP73*, *PG1*, *LZTS1*, and *MSR1*, but the roles

of these genes are also disputable because the functional significance of the identified polymorphisms is not known.

HPC has also been associated with polymorphisms in genes involved in steroid hormone metabolism and steroid hormone signal transduction, including AR, SRD5A2, and cytochrome P450 isoforms (CYP17A1 and CYP3A4), vitamin D metabolism including VDR, and carcinogen metabolism including GSTM1 (Reviewed by Coughlin and Hall 2002). These polymorphisms, however, are not highly penetrant alleles in families at high risk for PrCa.

In summary, prostate cancer etiology involves several genetic loci, but so far no gene has been identified that accounts for a large proportion of susceptibility to the disease (reviewed by Nwosu et al. 2001).

1.3.3 Sporadic prostate cancer

1.3.3.1 Genome-wide search for sporadic prostate cancer loci

In the genome-wide search for chromosomal alterations in PrCa an important contribution was made by CGH analysis (Visakorpi et al. 1995; Cher et al. 1996; Nupponen et al. 1998b; Alers et al. 2000). Differences were identified between various clinical stages and histological grades of PrCa, implicating that certain chromosomal regions have their specific role in prostate tumourigenesis. Although CGH is not suitable to specify in detail a region of loss or gain, it provides a strong indication of the chromosomal alterations in the various stages of PrCa. Nevertheless, several discrepancies were found between the CGH-studies described above (see Appendix A2 and A3 - Tables II and III for more details). These differences may be due to the contamination of tumour DNA with DNA from normal cells, a small number of samples analysed, and by differences in interpretation of the data. For example, the chromosomal regions 1p36, 9q34, 11q13, 19, and 22q, are known to be problematic in CGH studies (Kirchhoff et al. 1998), therefore, the data of Nupponen et al. (1998a) describing loss of 1p36, 19 and 22q are of some concern.

In primary PrCa, chromosomal losses were found to predominate over gains in a 5 to 1 ratio (Visakorpi et al. 1995). Most frequently, deletions were found for chromosome arms 6q, 8p and 13q (Visakorpi et al. 1995; Alers et al. 2000). In metastatic PrCa a higher frequency of these three losses, and losses of 2q, 5q, 10q, and 16q were found. Furthermore, gains of 1q, 3q, 4q, 7q, 8q, 11p and 17q were described (Cher et al. 1996; Alers et al. 2000). In local recurrences, that typically develop during endocrine therapy, a higher frequency of 8p and 16q losses, additional losses of 1p, 15q, 17p, 19, and 22q, and a higher frequency of 8q gain and gains of 7p, Xp, and Xq were found (Visakorpi et al. 1995; Nupponen et al. 1998b). The results of the studies described above are summarized in Figure 1.4A and 1.4B (more details are listed in Appendix A2 and A3 - Tables II and III). As shown in Figure 1.4 and Appendix A2 and A3 – Tables II and III, loss of 2q, 5q, and 6q, and gain of 1q, 3q, 4q, 9q, 11p, and 17q are more frequently found in metastases than in local recurrences.

So far, little attention has been given to chromosomal alterations in PIN lesions (Qian et al. 1998; Zitzelsberger et al. 2001). Although losses of 8p and 13q, and gains of 7 and 8q were found in PIN, these studies are not included in Figure 1.4 and Appendix A2 and A3 – Table II and III, because a limited number of samples was studied and because inconsistencies were described that were not seen in any other stage of PrCa. These discrepancies are likely due to the small number of patients in each study, and by small tissue samples that necessitated PCR amplification.

Genome-wide searches in PrCa have also been performed by allelic imbalance analysis (Cunningham et al. 1996). Cunningham et al. (1996) analysed DNAs from primary PrCa with a set of polymorphic markers scattered along all chromosomal arms. They reported an increased allelic imbalance for the same regions as described above in primary PrCa. Except for 8p, the frequencies found were less than those in the CGH studies. In addition, Cunningham et al. (1996) reported frequent allelic imbalance on 18q. Observed differences in frequency may be due to polymorphic markers with low heterozygosity, or to markers mapping outside the region of loss identified by CGH. If a marker maps in a region that is too small to be detected by CGH, allelotyping detects an additional genomic alteration. Furthermore, it is impossible to monitor by CGH loss of one allele combined with gain of the second copy.









1.3.3.2 Analysis of individual chromosomes

Deletion of 8p is one of the most common chromosomal alterations in PrCa. Loss of this chromosomal arm occurs already in early stage PrCa and increases during tumour progression to approximately 60-75% in advanced cancers. Deletion of 10q is observed in metastatic PrCa in particular, where it can be as high as 35-40%. Because this thesis focuses on chromosomes 8p and 10q, the remaining part of this chapter is limited to these two chromosomes.

1.3.3.2.1 Chromosome 8

The picture of 8p loss in PrCa that emerged from genome-wide CGH studies was supported by many independent studies using FISH analysis (Macoska et al. 1994; Oba et al. 2001), and analysis of allelic imbalance (Kunimi et al. 1991; Bova et al. 1993; Chang et al. 1994; MacGrogan et al. 1994; Trapman et al. 1994; Macoska et al. 1995; Suzuki et al. 1995; Vocke et al. 1996; Prasad et al. 1998, and reviewed in Isaacs 1995; Kallioniemi and Visakorpi 1996; Roylance et al. 1997; Abate-Shen and Shen 2000). Specifically, allelic imbalance studies demonstrated the presence of several separate regions of loss on this chromosomal arm. Comparison of these data indicated the presence of two, possibly three, regions of allelic loss at 8p, suggesting that this chromosomal arm contains several tumour suppressor genes that are of importance for PrCa tumourigenesis. The identified regions include 8p22-p23, 8p12-p21, and 8p11-p12 (Figure 1.5).



arranged by their chromosomal localisation (April 2003 freeze UCSC database).

Gain of 8q is the most common chromosomal amplification in PrCa (Visakorpi et al. 1995; Cher et al. 1996; Nupponen et al. 1998b). Gain of this chromosomal arm is frequently seen in combination with loss of 8p. Gain of 8q occurs occasionally in early stage PrCa and increases during tumour progression to 75%. Nupponen et al. (1998b) identified two independently amplified regions on the q-arm, 8q21 and 8q23-qter, suggesting the presence of at least two PrCa-related oncogenes on 8q. The latter region was narrowed to 8q24 by Alers et al. (2000). Gain of either region on 8q was shown to correlate with tumour progression and poor prognosis (Van Den Berg et al. 1995; Sato et al. 1999; Alers et al. 2000), but the precise contribution of each region awaits the identification of the target genes. One of the strongest candidates is *MYC*, which maps at 8q24.2 (Reiter et al. 2000). A second candidate is *EIF3S3*, which maps at 8q24.1 (Saramaki et al. 2001). Detailed description of these target genes is beyond the scope of this thesis.

Part of the 8p losses and 8q gains in PrCa were found to occur through isochromosome 8q formation (Webb et al. 1996; Virgin et al. 1999; Macoska et al. 2000). However, there is also evidence for other mechanisms. In primary PrCa, loss of 8p was found more frequently than gain of 8q (Visakorpi et al. 1995; Alers et al. 2000), implicating that loss of 8p occurs earlier in prostate tumourigenesis than gain of 8q. Additional evidence was provided by Sato et al. (1999), who found frequent loss of 8p (24%) without gain of 8q in a large cohort of high-grade, advanced, non-metastatic PrCa (Sato et al. 1999).

Loss of 8p is not only common in PrCa, but also in many other tumour types, including carcinomas of the bladder (Kallioniemi et al. 1995; Takle and Knowles 1996; Wagner et al. 1997), breast (Kerangueven et al. 1997; Nishizaki et al. 1997; Anbazhagan et al. 1998), colon (Cunningham et al. 1993; Fujiwara et al. 1993; Chang et al. 1994), head and neck (Wu et al. 1997; El-Naggar et al. 1998), kidney (Schoenberg et al. 1995; Schullerus et al. 1999), lung (Ohata et al. 1993; Fujiwara et al. 1994; Lerebours et al. 1999), liver (Fujiwara et al. 1994; Pineau et al. 1999; Wang et al. 1999), ovary (Wright et al. 1998), stomach (Yustein et al. 1999; Baffa et al. 2000), and uterus (Fujino et al. 1994; Ahmed et al. 2000). Not unexpected, most of these regions of loss partially or completely overlap with the regions depicted in Figure 1.5. However, comparison of all data results in a complicated and rather confusing pattern. Besides experimental failures, these differences may reflect tissue specific

tumour suppressor genes. In this overview, only the smallest common deleted regions in PrCa will therefore be taken into account for further analysis.

The various regions of 8p loss were also found to correlate with tumour grade and progression (Matsuyama et al. 1994; Macoska et al. 1995; Suzuki et al. 1995; Jenkins et al. 1998; Sato et al. 1999; Matsuyama et al. 2001; Oba et al. 2001). Specifically, losses of 8p22 and 8p21.3 occur in early stage PrCa and correlate with tumour grade, indicating that these regions may harbour a tumour suppressor gene. On the other hand, deletion of 8p21.1-p21.2 occurs mainly in advanced prostate cancer, showing capsular penetration or positive nodal metastases, and was found to correlate significantly with tumour progression. In particular, deletion of 8p21.1-p21.2 was more frequently observed in PrCa cases with lymph node metastases than those without. These results implicate that this specific region may harbour a "metastatic" suppressor gene. No clinical pathological parameters were found to associate with deletions on 8p11-p12.

Functional evidence for a tumour suppressor gene at 8p involved in PrCa was initially obtained by chromosome transfer studies. Transfer of human chromosome 8p into a rat prostate cancer cell line, resulted in suppression of its metastatic capacity (Ichikawa et al. 1994; Nihei et al. 1996). Recently, Nihei et al. (2002) limited a tumour suppressor gene region to a surprisingly small region of 60 Kbp at 8p12-p21 in a chromosome transfer study. Although the human genome sequence of this region is known, the authors did not mention any candidate gene within this region. Genes located in and near this region are *RBPMS*, *1D12A*, *AK057533*, and *GTF2E2* (April 2003 freeze UCSC database). So far, none of these genes has been analysed in PrCa. No other small regions were identified in PrCa by chromosome transfer.

In a human colorectal cancer cell line, transfer of chromosome 8p22-p23 resulted in a less tumourigenic phenotype (Gustafson et al. 1996). Furthermore, transfer of 8p11.1-q11.1 restored the DNA-repair defect of murine SCID fibroblast cell lines SCVA2 and SCVA4 in response to ionising radiation (Kurimasa et al. 1994). Although both studies demonstrated reduction of tumourigenicity in a non-PrCa derived model system, the transferred tumour suppressor genes may also be involved in PrCa.

8p22-p23 The best effort to date to localize a tumour suppressor gene at 8p22-p23 was the identification of a 730-970 Kb homozygous deletion in a PrCa lymph node

32 | Chapter 1

metastasis (Bova et al. 1993; Bova et al. 1996). A candidate tumour suppressor gene located within this homozygously deleted region is the N33 gene (Macgrogan et al. 1996). Although no mutations were found, N33 transcripts were undetectable in cell lines derived from lung, liver and colon cancers (see Appendix A1 - Table I). Absence of expression was strongly correlated to hypermethylation of a CpG island in the promoter region of the N33 gene, suggesting epigenetic inactivation of the gene. However, reintroduction of N33 cDNA in these colorectal cancer cell lines did not change their growth or tumourigenic properties (MacGrogan and Bookstein 1997). In the PrCa derived cell lines DU-145, TSU-PR1, LNCaP, and PPC-1, N33 inactivation was not detected. A second homozygous deletion (440-860 Kb) that partly overlapped with the deletion described by Bova et al. (1993) was found in a pancreatic tumour cell line (Levy et al. 1999). This overlap narrowed down the tumour suppressor gene region to less than 600 Kb and was found to contain, apart from N33, at least four ESTs that might represent a tumour suppressor gene (Bova et al. 1996; Macgrogan et al. 1996; Levy et al. 1999; Arbieva et al. 2000). To date however, no novel candidate gene has been identified in this region. Levy et al. (1999) speculated that this region of 8p22 simply is a fragile site without any biological significance for cancer cells, implicating a tumour suppressor gene elsewhere on 8p22-p23 in PrCa.

PDGFRL (platelet-derived growth factor receptor-like) is another candidate tumour suppressor gene at 8p22 (Fujiwara et al. 1995; Bova et al. 1996). However, only one missense mutation was found in a large series of prostate tumours (see Appendix A4 - Table IV), implicating a minor role for PDGFRL in PrCa. Consistent with this low mutation frequency, mutations in *PDGFRL* were not detected in the cohort of 43 prostate tumour DNA samples studied by us (Chapter 6; Van Alewijk et al., unpublished results).

A more recently identified candidate tumour suppressor gene at 8p22 is *LZTS1* (Ishii et al. 1999), which showed mutations in primary oesophageal cancers and in the prostate cancer cell line PC-3 (see Appendix A4 - Table IV). Intriguingly, *LZTS1* mRNA expression was undetectable in over 60% of epithelial tumours from various anatomical sites. Ishii et al. (1999) therefore suggested a role for LZTS1 in multiple human tumour types, including PrCa. We found down regulation of *LZTS1* expression in one xenograft, but no mutations in our cohort of 43 prostate tumour DNA samples and 15 PrCa derived xenografts and cell lines, including PC-3 (see Appendix A4 - Table IV). Although our results implicate a minor role for LZTS1 in PrCa, some support

for LZTS1 was provided by Cabeza-Arvelaiz et al. (2001). They demonstrated that transfer of YAC and BAC clones containing the *LZTS1* gene into rat prostate AT6.2 cells reduced their colony-forming efficiency. Subsequent experiments showed that over-expression of *LZTS1* cDNA inhibited colony-formation in soft agar of AT6.2, HEK-293, and LNCaP cells (Cabeza-Arvelaiz et al. 2001).

8p12-p21 Allelic imbalance studies strongly suggested that 8p12-p21 harbours a tumour suppressor gene involved in PrCa. The overall judgement of this region, however, is complicated by the heterogeneity of allelic losses (Figure 1.4). This may indicate that 8p12-p21 contains more than one tumour suppressor gene. The best evidence to date for a tumour suppressor gene in this region again is the identification of homozygous deletions. Four homozygous deletions were identified in PrCa (Kagan et al. 1995; Prasad et al. 1998; Van Alewijk et al. 1999). Kagan et al. (1995) reported two huge overlapping homozygous deletions, of 5 and 21 Mb, around the *NEFL* locus. The significance of these two homozygous deletions is unclear because of their size, but also because they are roughly defined, and because none of the deletions was flanked by a region of allelic loss. A homozygous deletion however, needs to be defined in further detail. The only nearby located genes are *FLJ30656* and *UNC5D* (April 2003 freeze UCSC database).

The fourth homozygous deletion on 8p12-p21 was reported by our group (Chapter 4 of this thesis; Van Alewijk et al. 1999). We identified a 890 Kb homozygous deletion in a PrCa derived xenograft (Chapter 5; Van Alewijk et al., submitted). This work became feasible after the construction of a continuous high-density physical and transcript map of 8p12-p21, encompassing the region between the markers D8S87 and D8S133 (Appendix A6 of this thesis). The interest for this specific region arose from previous results, showing frequent (69%) allelic loss in PrCa DNA samples (Trapman et al. 1994). Despite the availability of many markers, no second (overlapping) homozygous deletion was found in PrCa derived xenografts and cell lines. The Werner syndrome gene (*WRN*), which is disrupted by the homozygous deletion (Chapter 4; Van Alewijk et al. 1999), encodes a protein with DNA helicase and exonuclease activity involved in maintaining the integrity of the genome (Yu et al. 1996a; Gray et al. 1997; Suzuki et al. 1999; Hickson et al., 2003). Werner syndrome

predisposition to rare cancers, including soft tissue sarcoma, thyroid cancer, meningioma, and melanoma (Epstein et al. 1966; Goto et al. 1996; Yu et al. 1996a; Yu et al. 1997). However, no mutations or aberrant *WRN* gene expression patterns were identified in our set of PrCa derived xenografts and cell lines (Chapter 5; Van Alewijk et al., submitted). These results suggest that functional loss of two copies of *WRN* plays a minor role in PrCa, although it may be that the loss of the *WRN* gene contributed to the growth of xenograft PC133. Except for *WRN*, no other bona fide gene has been detected in the homozygous deletion region (see also Chapter 5 of this thesis).

The location and known function of clusterin (CLU; Fink et al. 1993), DNA polymerase β (POLB; Cannizzaro et al. 1988; Dib et al. 1995), and epoxide hydrolase 2 (EPHX2; Larsson et al. 1995), made these genes good candidates for tumour suppressor genes at 8p12-p21 (see Appendix A4 - Table IV). To test this hypothesis, we analysed 15 PrCa derived xenografts and cell lines, and 43 PrCa tumour samples for mutations and altered expression (Van Alewijk et al., unpublished results). In CLU, apart from silent polymorphisms, no sequence variations were identified. Although 2 out of 12 PrCa samples were reported to have a mutation in POLB (Dobashi et al. 1994), no ORF mutations were found in our cohort of 43 prostate tumour DNA samples (Chapter 6; Van Alewijk et al., unpublished results). Thus, we found no clear evidence for a role of any of these two genes in PrCa. Expression of EPHX2 was down-regulated in 2 samples. Furthermore, 6 other samples had retained an allelic variant with a unique amino acid substitution, which was also found in their normal tissue. Although 8p12-p21 has not been recognized as an HPC region, these results suggest a predisposition for PrCa due to EPHX2 polymorphisms. Yet, there is no further evidence for a role of EPHX2 in PrCa.

To date, the most promising candidate tumour suppressor gene at 8p12-p21 is the homeobox gene *NKX3-1* (He et al. 1997). *NKX3-1* encodes a homeodomaincontaining protein related to the Drosophila *NK-3* gene family. Northern blot analysis revealed that NKX3-1 had a unique tissue expression pattern. The 3.5 Kb *NKX3-1* transcript was abundantly expressed in the prostate, at lower levels in the testis, and absent from all other tissues tested (He et al. 1997). Furthermore, expression was seen in the hormone-responsive, AR-positive PrCa cell line LNCaP, and was markedly increased upon androgen stimulation. *NKX3-1* was not expressed in either of the two AR-negative cell lines PC-3 and DU-145 (He et al. 1997). Supported by frequent allelic
loss of 8p12-p21 in PrCa and the lack of expression in PC-3 and DU-145, He et al. (1997) suggested that *NKX3-1* was a candidate tumour suppressor gene. So far, no *NKX3-1* gene mutations or homozygous deletions have been identified in PrCa (Voeller et al. 1997; Ornstein et al. 2001; Chapter 6; Van Alewijk et al., unpublished results). However, we found low *NKX3-1* expression in 4 out of 15 PrCa derived xenografts and cell lines (PC133, PC135, PC324, and PC-3) (Chapter 6; Van Alewijk et al., unpublished results). All 4 xenografts and cell lines are hormone independent growing tumours.

The NKX-like transcription factors are implicated in many aspects of cell type specification and maintenance of the functions of differentiated tissues. Sciavolino et al. (1997) demonstrated that Nkx3-1 was involved in murine prostate organogenesis (Sciavolino et al. 1997). In line with expectations, homozygous Nkx3-1 knockout mice showed defects in prostatic ductal morphogenesis and secretory protein production (Bhatia-Gaur et al. 1999). Notably, not only homozygous but also heterozygous Nkx3-1 mutant mice displayed prostatic epithelial hyperplasia and dysplasia that increased in severity with aging (Bhatia-Gaur et al. 1999; Kim et al. 2002a), suggesting that loss of a single Nkx3-1 allele may be sufficient to initiate PIN-like lesions. Due to the role of Nkx3-1 in prostate organogenesis, conventional knockouts do not represent lifelike tumourigenesis. Conditional knockouts simulate a more realistic situation inducing a tumourigenic effect in normal matured tissue. Conditional Nkx3-1 inactivation in adult mice induced hyperplasia and apparent PIN lesions in the prostate (Abdulkadir et al. 2002). Evidence for actual growth-suppressing activity of Nkx3-1 was recently provided by retroviral gene transfer of exogenous Nkx3-1 into a human (PC-3) and rodent (AT6) PrCA derived cell line (Kim et al. 2002a).

1.3.3.2.2 Chromosome 10

The picture of 10q loss in PrCa that was obtained by genome-wide CGH studies was supported by many allelic imbalance analyses. Several of these studies found additional losses of 10p, which were not seen by CGH analysis (Gray et al. 1995; Ittmann 1996; Komiya et al. 1996; Trybus et al. 1996; Cairns et al. 1997; Feilotter et al. 1998; Chapter 3; Hermans et al., submitted). Comparison of the data revealed for both the 10p- and 10q-arm at least one, possibly two, regions of loss, suggesting that each chromosome 10 arm contains one or more PrCa tumour

suppressor genes. The identified regions are 10p12, 10p14-pter, 10q23, and 10q25q26 (Figure 1.6).

Losses of 10p and 10q are not restricted to PrCa. High frequency loss of these chromosomal arms has also been described in bladder cancer (Cappellen et al. 1997), endometrial carcinoma (Peiffer et al. 1995; Nagase et al. 1997), hepatocellular carcinoma (Fujiwara et al. 2000), glioblastoma (Karlbom et al. 1993; Rasheed et al. 1995), malignant melanoma (Herbst et al. 1994; Reiffenberger 1999), meningioma (Rempel et al. 1993), non-Hodgkin's lymphoma (Speaks et al. 1992), renal cell carcinoma (RCC; Morita et al. 1991), and small lung cell cancer (Kim et al. 1998).

In many studies, losses of 10p and 10q were found to correlate with advanced clinical stages of PrCa, particularly metastases and local recurrences (Ittmann 1996; Komiya et al. 1996; Trybus et al. 1996; Dong et al. 1998; Ittmann 1998; Alers et al. 2000; Ozen et al. 2000; Srivastava et al. 2001; Leube et al. 2002).





10p12 and 10p14 Functional evidence for prostate tumour suppressor genes at 10p was initially obtained by chromosome transfer studies. Transfer of human chromosome 10p to human prostate cancer cell lines, resulted in suppression of their growth capacity (Murakami et al. 1996; Sanchez et al. 1996). Fukahara et al. (2001) narrowed down this tumour suppressor gene region to 1.2 Mb at 10p14 (Fukuhara et al. 2001). The authors did not refer to any candidate gene within this region. Genes located in or near this region are *GATA3* and *CUGBP2* (April 2003 freeze UCSC database). So far, none of these genes has been analysed in PrCa. Transfer of chromosome 10p14-p15 into a human glioblastoma cell line resulted in a markedly suppression of its colony forming ability in soft agar (Kon et al. 1998).

Within the overlapping region of loss at 10p12, no candidate tumour suppressor gene has been identified (Figure 1.6). The more telomeric loss at 10p14 encompasses the *COPEB* gene. This gene was found to be mutated in 55% of PrCa (Narla et al. 2001). The authors also showed that wild-type COPEB up-regulates CDKN1A in a p53-independent manner and significantly reduces cell proliferation, whereas tumour-derived *COPEB* mutants did not. Despite the high mutation frequency, the tumour suppressor gene functioning of *COPEB* has so far not been confirmed in any other study. Neither did we find any structural nor expression alterations in the panel PrCa derived xenografts and cell lines (Chapter 3; Hermans et al., submitted).

10q23 Transfer of 10q into a rat prostate cancer cell line resulted in suppression of its metastatic capacity (Nihei et al. 1995). No smaller regions of 10q were identified by chromosome transfer.

Shortly after the first allelic imbalance studies narrowed down the region of loss to 10q23 (see Fig. 1.8), the *PTEN* (phosphatase and tensin homologue deleted on chromosome 10) tumour suppressor gene was identified within this region (Li and Sun 1997; Li et al. 1997; Steck et al. 1997). Especially the finding of homozygous deletions has been pivotal in the identification of *PTEN* (Li et al. 1997; Steck et al. 1997). *PTEN* is also known as *MMAC1* (mutated in multiple advanced cancers (Steck et al. 1997) and *TEP1* (*TGF-β*-regulated and epithelial cell-enriched phosphatase (Li and Sun 1997).

The initial molecular cloning studies reported *PTEN* mutations in a large fraction of glioblastoma multiforme cell lines, xenografts, and primary tumours, as well as in smaller samples of breast and prostate cancers (Li et al. 1997; Steck et

al. 1997). Subsequent analyses confirmed that homozygotic inactivation of PTEN occurs in a large percentage of glioblastomas (at least 30% of primary tumours and 50-60% of cell lines) but not in lower-grade (i.e., less advanced) glial tumours. *PTEN* mutations also are very common in melanoma cell lines (>50%), advanced PrCa, and endometrial carcinomas (30-50%). Although *PTEN* mutations are predominantly found in advanced glial and prostate tumours, mutations occur with equal frequency at all stages of endometrial carcinogenesis. A significant percentage (~10%) of breast cancer cell lines has inactivated *PTEN*. *PTEN* mutations are rare in sporadic breast tumours, independent of severity. Although germ-line *PTEN* mutations are not a frequent cause of familial breast cancer. Occasional *PTEN* mutations have been reported in head and neck cancers and in thyroid cancers (Cantley and Neel 1999).

Germ-line mutations in *PTEN* have been detected in Cowden disease, Lhermitte-Duclos disease, and Bannayan-Zonana syndrome, all of which are rare diseases with an autosomal dominant inheritance pattern. These syndromes are associated with hamartomas of the skin, thyroid, breast, oral mucosa, and intestine, and a predisposition to cancer (Cantley and Neel 1999).

Thus far, *PTEN* is the most frequently mutated tumour suppressor gene in PrCa. Li et al. (1997) reported homozygous deletions of *PTEN* in 2 out of 4 PrCa cell lines. We found a high frequency (60%) of *PTEN* deletions (5 out of 15) and mutations (4 out of 15) in PrCa derived xenografts and cell lines (Chapter 2; Vlietstra et al. 1998). *PTEN* mutations are most common in metastatic PrCa (30-60%; Cairns et al. 1997; Suzuki et al. 1998b), but less frequent in primary PrCa, (5-15%; Cairns et al. 1997; Teng et al. 1997; Feilotter et al. 1998; Whang et al. 1998). This implies that complete PTEN inactivation occurs mainly in advanced stage PrCa, concordant with 10q loss.

PTEN encodes a widely expressed 5.5-kb mRNA. The protein consists of 403 amino acids, it contains a catalytic domain of lipid and protein phosphatases and shows homology to the cytoskeletal proteins tensin and auxilin (Li and Sun 1997; Li et al. 1997; Steck et al. 1997). PTEN is involved in regulation of cell proliferation, cell survival, cell size, and chemotaxis (Figure 1.9; reviewed by Comer and Parent 2002).



Figure 1.9. Mechanisms of PTEN action.

Although PTK2 (protein tyrosine kinase 2) and the adaptor protein Shc have been postulated as targets of the weak in vivo protein phosphatase activity of PTEN (Gu et al. 1998; Tamura et al. 1998), the most prominent function of PTEN is the negative control of phosphoinositidyl 3-kinase (PI3K) signaling by dephosphorylation of the phospholipid PIP3 (phosphoinositol-3-phosphate; Maehama and Dixon 1998; Myers et al. 1998; see for recent reviews Cantley and Neel 1999; Dahia 2000). A prominent downstream target of PIP3 is AKT/PKB, which subsequently phosphorylates a large variety of target proteins, including BAD, CASP9, CREB, eNOS, FOXO3A, GSK3, IKK, MDM2, FRAP1, and CDKN1A. These targets have a characteristic R-x-R-xx-S/T-F/L motif that is phosphorylated at the serine or threonine residues by activated AKT. Several of the AKT target proteins are involved in more than one of the biological functions of PTEN. The functioning of AKT is further complicated by cross-talk of some of its downstream targets with other tumour related pathways (e.g. TP53-, and WNTpathway). In addition to the regulating mechanisms as mentioned above PTEN is thought to affect cell migration. Initially, FAK and Shc have been described as mediators for migration. However, evidence is growing that other downstream targets of Pi3k, like Rac / Cdc42, which might function upstream of Akt, have a more prominent role in cell migration (Higuchi et al. 2001). Little is known about the expression of PTEN in PrCa. Downregulation of PTEN expression was shown by McMenamin et al. (1999).

Recently, Yang et al. (2002) described an inverse correlation between PTEN (+) and CDKIN1B (+) expression, and the expression of the F-box protein Skp2 (-) (Yang et al. 2002).

Functional evidence for a role of PTEN in PrCa was primarily provided by Nihei et al. (1995), who showed that transfer of chromosome 10, particularily 10cen-q23 (including PTEN), suppressed the metastatic ability of a rat prostate cancer cell line (Nihei et al. 1995). Ever since, several studies provided direct and indirect evidence for a role of PTEN in PrCa. Transfer of PTEN cDNA into PC-3 cells induced G1 cellcycle arrest, enhanced apoptosis, and reverted their invasive phenotype (Persad et al. 2000; Kotelevets et al. 2001). Accordingly, adenovirus-mediated expression of PTEN was found to inhibit the *in vitro* growth capability of PC-3 cells, primarily by blocking cell cycle progression. In vivo these transfected cells did not lose their tumourigenicity, but showed a significant reduction in tumour size and complete loss of their metastatic ability (Davies et al. 2002). In LNCaP cells, adenovirus-mediated expression of PTEN or cDNA transfection was found to negatively regulate the PI3K/AKT pathway and enhance apoptosis (Wu et al. 1998; Yuan and Whang 2002). Furthermore, PTEN was found to induce chemosensitivity in LNCaP cells by suppression of BCL-2 expression. Specifically, the lipid-phosphatase activity of PTEN was found to be required for the inhibition of BCL-2. This inhibiting activity was blocked by overexpression of a constitutively active form of AKT (Huang et al. 2001).

Intriguingly, part of the losses at 10q23 included the *PTEN* gene, yet mutations were not identified in the remaining *PTEN* allele (Cairns et al. 1997; Teng et al. 1997; Feilotter et al. 1998; Whang et al. 1998). These observations suggested that a second tumour suppressor gene located nearby *PTEN* might be involved in PrCa. Alternatively, this could point to *PTEN* haplo-insufficiency in tumour growth. Nearby candidate tumour suppressor genes are *MINPP1*, *PAPSS2*, and *TNFRSF6* (see Appendix A5 - Table V). *MINPP1*, whose function overlaps with that of PTEN, had a missense mutation in a follicular thyroid tumour (Gimm et al. 2001). Gimm et al. (2001) therefore suggested a role for *MINPP1* in the pathogenesis of at least a subset of malignant follicular thyroid tumours. In a large inbred Pakistani family with severe chondrodysplasia, a nonsense mutation was found in *PAPSS2* (ul Haque et al. 1998). *TNFRSF6* was found to be inactivated by DNA methylation in a limited number of PrCa and advanced bladder carcinomas (Santourlidis et al. 2001). Additionally, specific inherited *TNFRSF6* mutations were correlated with a high risk factor for non-

Hodgkin and Hodgkin lymphomas (Straus et al. 2001). We screened 13 genes mapping in a 3 Mbp region around *PTEN* for expression alterations (Chapter 3; Hermans et al., submitted). Additionally, the genes *MINPP1, PAPSS2, FLJ11218* and *TNFRSF6*, were selected on the basis of altered expression, frequent deletion or by virtue of their function (see also Appendix A5 - Table V) for a mutation screen in our panel of xenografts and cell lines. Especially, *PAPSS2* and *FLJ11218* appeared interesting due to altered expressions, polymorphisms, a few missense mutations and a nonsense mutation. However, the functional relevance of these alterations in PrCa remains to be established.

Homozygous *Pten* knockout mice are embryonically lethal (Di Cristofano et al. 1998b; Suzuki et al. 1998a; Podsypanina et al. 1999). Heterozygous *Pten* knockout mice showed hyperplastic and dysplastic changes in the prostate, skin, and colon, which are characteristic for Cowden disease, Lhermitte-Duclos disease, and Bannayan-Zonana syndrome (Di Cristofano et al. 1998a). These mice also spontaneously developed germ cell, gonadostromal, thyroid, colon, T-cell and breast tumours. Analysis of PrCa progression in transgenic adenocarcinoma of mouse prostate (TRAMP) model mice crossed with *Pten* (+/-) heterozygous mice, revealed that haplo-insufficiency of the *Pten* gene promoted the progression of PrCa (Kwabi-Addo et al. 2001). Evidence is growing that not only in PrCa but also in other human malignancies loss of one *PTEN* allele can contribute to tumourigenesis (Velickovic et al. 2002; Byun et al. 2003).

10q25-q26 A fourth region of frequent loss of chromosome 10 includes 10q25-q26, which indicates a role for an additional 10q gene in PrCa (Cairns et al. 1997; Feilotter et al. 1998). This chromosomal arm harbours the candidate tumour suppressor genes *MXI1* (10q25.2), and *DMBT1* (10q26.1), which have been suggested to be involved in PrCa or other tumours (*MXI*: Eagle et al. 1995; Wechsler et al. 1996; Prochownik et al. 1998) (*DMBT1*: Mollenhauer et al. 1997; Somerville et al. 1998; Mori et al. 1999; Takeshita et al. 1999; Wu et al. 1999). In accordance with Gray et al. (1995), Kawamata et al. (1996), and Kuczyk et al. (1998), who doubted the role of *MXI1* in PrCa (Gray et al. 1995; Kawamata et al. 1996; Kuczyk et al. 1998), we did not find any altered expression or structure of *MXI1* (Chapter 3; Hermans et al., submitted). In one of our xenografts, analysis of *DMBT1* revealed an internal homozygous deletion, including a repetitive region from which several *DMBT1* variants are derived (Mollenhauer et al. 1999;

Chapter 3; Hermans et al., submitted). Although this deletion resulted in a shortened protein, its functional effect is as yet unknown (Mollenhauer et al. 1999). Down-regulation of *DMBT1* expression was found in lung, oesophageal, gastric and colon cancers, implicating a general role in cancer (Mori et al. 1999; Takeshita et al. 1999). In addition to these studies, we found expressional down-regulation in 2 samples of our panel of xenografts and cell lines (Chapter 3; Hermans et al., submitted).

1.4 SCOPE OF THIS THESIS

The aim of this thesis was the isolation and characterization of tumour suppressor genes located on chromosome 8p and 10g. Chapter 1 is a general introduction of PrCa, genetics of cancer, and genetic alterations in PrCa. Chapter 2 describes the expression and structural analysis of the PTEN gene in PrCa xenografts and cell lines. Chapter 3 describes the genetic analysis of chromosome 10. It also describes a refined characterization of the previously identified homozygous deletions disrupting the PTEN gene (Chapter 2), including the expression and structural analysis of genes flanking PTEN. Finally, chapter 3 describes the expression and structural analysis of the candidate tumour suppressor genes MXI1 and DMBT1 that are located on 10q. Chapter 4 describes the identification and characterization of a homozygous deletion at 8p12-p21 in xenograft PC133. Chapter 5 describes the genetic analysis of chromosome 8 in PrCa xenografts and cell lines using CGH and allelotyping. It further describes the screen for overlapping homozygous deletions and search for genes in the previously identified homozygous deletion. In addition, this chapter describes the expression and structural analysis of WRN. Chapter 6 describes the high-density screen for an homozygous deletion elsewhere on 8p12-p21 and the expression and structural analysis of the candidate tumour suppressor genes LZTS1, NKX3-1, and EXPHX2 that are located on 8p. Chapter 7 discusses the results described in this thesis, its implications for PrCa and future directions of research.

CHAPTER 2

FREQUENT INACTIVATION OF *PTEN* IN PROSTATE CANCER CELL LINES AND XENOGRAFTS

Jame

Remko J. Vlietstra, Dirk C.J.G. van Alewijk, Karin G.L. Hermans, Gert J. van Steenbrugge, and Jan Trapman

Cancer Res 1998; 58: 2720-2723

ABSTRACT

Loss of chromosome 10q is a frequently observed genetic defect in prostate cancer. Recently, the *PTEN/MMAC1* tumor suppressor gene was identified and mapped to chromosome 10q23.3. We studied *PTEN* structure and expression in four *in vitro* cell lines and eleven *in vivo* xenografts derived from six primary and nine metastatic human prostate cancers. DNA samples were allelotyped for eight polymorphic markers within and surrounding the *PTEN* gene. Additionally, the nine *PTEN* exons were tested for deletions. In five samples (PC3, PC133, PCEW, PC295, PC324) homozygous deletions of (parts of) the *PTEN* gene were detected. PC295 contained a small homozygous deletion, encompassing *PTEN* exon 5. In two DNAs (PC82 and PC346), nonsense mutations were found, and in two (LNCaP and PC374), frame-shift mutations were found. Missense mutations were not detected. *PTEN* mRNA expression was clearly observed in all cell lines and xenografts without large homozygous deletions. The high frequency (60%) of *PTEN* mutations and deletions indicates a significant role of this tumor suppressor gene in the pathogenesis of prostate cancer.

INTRODUCTION

Prostate cancer is the most frequently diagnosed tumor in men in the United States and in western and northern Europe, and the second leading cause of male cancer death (Parker et al. 1997). The molecular events leading to the development and the progressive growth of prostate cancer are poorly understood. The most frequent chromosomal aberrations are losses of chromosomes 8p, 10q, 13q and 16q (Bergerheim et al. 1991; Visakorpi et al. 1995; Cher et al. 1996), indicating the localization of tumor suppressor genes at these chromosomal sites. Detailed allelotyping implicated deletion of chromosome 10 region 10q23-25 in prostate cancer (Gray et al. 1995; Komiya et al. 1996; Li et al. 1997).

Recently, the *PTEN* gene, also known as *MMAC1* or *TEP1*, which is located at 10q23.3, has been found to be frequently mutated or deleted in glioblastomas (Li and Sun 1997; Li et al. 1997; Steck et al. 1997). *PTEN* encodes a dual-specific phosphatase and shows homology to the cytoskeletal proteins tensin and auxilin (Li and Sun 1997; Li et al. 1997; Myers et al. 1997; Steck et al. 1997). Structural analysis revealed mutations

of *PTEN* in many different tumor types. A high proportion of mutations was not only demonstrated in glioblastomas (Li et al. 1997; Rasheed et al. 1997; Steck et al. 1997; Wang et al. 1997) but also in endometrial carcinomas (Risinger et al. 1997; Tashiro et al. 1997). Germ-line mutations in *PTEN* have been detected in Cowden disease, an autosomal dominant cancer predisposition syndrome, associated with an increased risk of breast, skin, and thyroid cancer (Liaw et al. 1997; Nelen et al. 1997).

PTEN has also been implicated in prostate cancer. Li et al. (Li et al. 1997) described two homozygous deletions and one frame-shift mutation in prostate cancer cell lines. More recently, *PTEN* mutations and deletions were reported in metastatic prostate cancers (Cairns et al. 1997; Suzuki et al. 1998b). In primary tumors, *PTEN* mutations were less frequent (Cairns et al. 1997; Teng et al. 1997).

Detailed molecular genetic analysis of prostate cancer DNA and gene expression is complicated due to the contamination by normal cells. In this study, we demonstrate a high proportion of *PTEN* mutations and deletions in fifteen prostate cancer xenografts and cell lines. We characterized these mutations, and studied *PTEN* mRNA expression.

MATERIALS AND METHODS

Prostate Tumor Cell Lines and Xenografts.

The *in vitro* growing cell lines LNCaP, PC3, DU145 and TSU were cultured under standard conditions. *In vivo* xenografts PC82, PCEW, PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346 and PC374 were propagated in male nude mice (Noordzij et al. 1996; Van Weerden et al. 1996; and references therein).

DNA Preparation.

DNA from cell lines was isolated according to standard procedures (Sambrook et al. 1989). Genomic DNA from xenografts was isolated from 5 consecutive 5-µm cryostat tissue sections by overnight proteinase K incubation at 55°C, followed by phenol extraction and ethanol precipitation. DNA pellets were dissolved in TE [10 mM Tris.HCl (pH7.8)-1 mM EDTA].

Allelotyping.

DNAs from prostate cancer cell lines and xenografts were allelotyped by PCR amplification of eight polymorphic markers within or flanking the *PTEN* locus at chromosome 10q23.3: *D10S1687*, *D10S579*, *D10S215*, *PTENCA*, *AFMa086WG9*, *D10S541*, *D10S1753* and *D10S583*. *PTENCA* primers were from Ref. 12; other markers are described in Genome Data Base or in the Genethon database. PCR amplifications, using Taq polymerase (Promega, Madison, WI) were for 30 cycles of 1 min at 94°C, 1.5 min at 50°C or 55°C and 1.5 min at 72°C in a 15 µl reaction volume, containing 1.5 mM MgCl₂ and 1 µCi [alpha-³²P]dATP (Amersham, Buckinghamshire, UK). The radio-labelled PCR products were separated over a 6% polyacrylamide sequencing gel.

Screening for Homozygous Deletions of PTEN.

PTEN exons and flanking sequences were PCR amplified according to standard protocols: 30 cycles of 45 s at 94°C, 45 s at 50°C (exons 1 to 6) or 55°C (exons 7 to 9) and 45 s at 72°C in the presence of 1.5 mM (exons 1 to 6) or 2.0 mM MgCl₂ (exons 7 to 9). Exons 5 and 8 were amplified as two overlapping fragments. Primers were from (Steck et al. 1997) (1R, 5F1, 5R2, 7F and 7R, 8R2), and (Guldberg et al. 1997) (1F, 2F and 2R, 3F and 3R, 4F and 4R, 5R1 and F2, 6F and 6R, 9F and 9R). Primers 8R1 (5-CTTGTCATTATCTGCACGCT-3) and 8F2 (5-GAAAATGGAAGTCTATGTG-3) are novel. Control PCRs were with primers 791B9L-A (5-GAAGGTGGCAGTCTGATCTC-3) and 791B9L-B (5-GCAACTGGTTGAAACATACTC-3), which amplify a 410 bp fragment at chromosome 8p12-p21. Amplified products were separated over a 2% agarose gel.

RT-PCR Analysis of PTEN Expression.

RNA was isolated by standard guanidium isothiocyanate (cell lines) or LiCl (xenografts) protocols (Sambrook et al. 1989). cDNA synthesis, followed by PCR amplifications with PTEN-specific primers were performed on 500 ng RNA in the Access RT-PCR system (Promega) according to the instructions of the manufacturer. Primers for cDNA synthesis follows: 9R-cDNA, 5used were as GGATGTGAACCAGTATATCACAA-3 (Fig. 3A); and 7R-cDNA, 5-CCGTCGTGTGGGTCCTGAATTAA-3 (Fig. 3B). cDNA synthesis was for 45 min at 48°C; amplification: 35 cycles of 30 sec at 94°C, 30 sec at 60°C, and 1 min at 68°C. Primer combinations for PCR amplification of cDNAs: 9R-cDNA and 8F-cDNA (5-7R-cDNA AGCCAACCGATACTTTTCTCC-3), and and 1F-cDNA (5CCACCAGCAGCTTCTGCCATCTCT-3). Control RT-PCRs were carried out under the same conditions, using RNA polymerase II-specific primers: PolF-cDNA (5-GCTGAGAGAGCCAAGGATAT-3) and PolR-cDNA (5-CACCACCTCTTCCTCCTCTT-3). RT-PCR products were separated over a 2% agarose gel.

PCR-SSCP Analysis.

Fragments for PCR-SSCP analysis were obtained for all exons of the *PTEN* gene, utilizing primer sets and PCR conditions described above. PCR reactions were in a 15- μ l volume in the presence of 1 μ Ci [alpha-³²P]dATP. Appropriate aliquots of the radiolabelled PCR products were separated over a 6% non-denaturing polyacrylamide gel containing 5 or 10% glycerol at 7 W overnight at room temperature.

Structural Analysis.

RT-PCR fragments and amplified exons were purified over QIAquick spin columns (Qiagen, Hilden, Germany), cloned into pGEM-T Easy (Promega) and sequenced according to the dideoxy chain termination method.

RESULTS

Allelotyping of Prostate Cancer Cell Lines and Xenografts for Chromosome 10q23.

Genomic DNAs from 11 prostate cancer xenografts and 4 *in vitro* propagated cell lines were allelotyped for eight highly polymorphic markers, spanning ~9 cM around the *PTEN* locus at chromosome 10q23 (*D10S1687*, *D10S579*, *D10S215*, *PTENCA*, *AFMa086wg9*, *D10S541*, *D10S1753*, and *D10S583*). The results are summarized in Fig. 1. For most markers, especially those mapping at 114 cM, only one amplified band was detected, suggesting hemizygosity. In DNAs from three xenografts (PC133, PC324 and PCEW) and one cell line (PC3), two or more polymorphic markers were completely deleted (see also (Li et al. 1997) for PC3 deletion). All four samples were negative for *AFMa086wg9*, which is situated at 114 cM, between exons 2 and 3 of the *PTEN* gene. Therefore, these tumors are expected to be completely or partially defective of *PTEN*. The homozygous deletion in PC3 is relatively large (>5 cM), other homozygous deletions seem to be much smaller.

Homozygous Deletion of the PTEN Gene.

To confirm *PTEN* losses and to determine more precisely the borders of the homozygous deletions, all nine exons of the *PTEN* gene, including flanking sequences, were individually amplified. Examples of exons 5 and 8 amplifications are shown in Fig. 2A; results are summarized in Fig. 2B. The four tumors that lacked polymorphic marker *AFMa086wg9* showed complete or partial deletion of *PTEN* (Fig. 2A, *Lanes 1,3,7* and *13*). In PC133 and PCEW, *PTEN* was completely deleted; in PC324, *PTEN* exons 2-9, and in PC3 exons 3-9 were absent (Fig. 2B). Importantly, PC295 contained a very small homozygous deletion, encompassing only *PTEN* exon 5, which could not be detected by allelotyping (Fig. 2A, *Lane 5*).

10576 1051 00051 000500000000														
position —	113cM	114cM	114cM	114cM	114cM	114cM	119cM	122cM						
PC 3	1	1	1	1	0	0	0	1						
PC 82	2	1	1	1	1	1	2	2						
PC 133	1	1	1	0	0	0	1	1						
PC 135	2	1	1	1	1	1	1	2						
PC 295	1	1	1	1	1	1	1	1						
PC 310	2	1	2	2	1	2	2	2						
PC 324	2	1	1	1	0	0	2	1						
PC 329	2	1	2	1	1	2	1	1						
PC 339	2	2	2	1	1	1	1	1						
PC 346	2	1	1	1	1	1	2	2						
PC 374	1	1	1	1	1	1	1	1						
DU 145	2	1	1	1	1	1	1	1						
PCEW	2	0	0	0	0	1	1	1						
LNCaP	1	1	1	1	1	1	1	1						
TSU	1	1	1	1	1	1	1	1						

Figure 1. Allelotyping of prostate cancer cell lines and xenografts at chromosome 10q23.3. Markers are described in more detail in Materials and Methods. Numbers between brackets indicate heterozygosity indices. Heterozygosity indices are not available for *PTEN CA* and *AFMa086WG9*. Genetic map positions of markers are from contig 10.7 of the Whitehead Institute map. The number of bands detected are indicated in each box (0, 1 or 2). Homozygous deletions are marked by shaded boxes.



Figure 2. Deletion analysis of *PTEN* in prostate cancer cell lines and xenografts. (A) Agarose gel electrophoresis of PCR amplified exons 5 and 8. The control is YAC end fragment 791B9L, located at chromosome 8p12-p21. (B) Overview of the results obtained for all *PTEN* exons. Deleted exons are indicated by a minus in the shaded boxes.

Expression of PTEN.

Because of the absence of contaminating normal cells of human origin, xenografts and cell lines are preferable sources for the study of PTEN mRNA expression. In all RNA preparations from cell lines and xenografts without homozygous deletion, PTEN mRNA was easily detectable by RT-PCR, using human PTEN specific primers for amplification (Fig. 3A). In PC133 and PCEW, PTEN mRNA expression could not be visualized (Fig. 3A lanes 4, 15). However, PC3 and PC324, which also lack the PTEN gene showed a faint amplified band of the appropriate length (620 bp), which hybridized to a PTEN specific probe (data not shown). Sequencing of the RT-PCR product from PC324 and PC3 revealed that it was not identical to PTEN, but to a highly homologous processed PTEN pseudogene, located in a duplicated region of chromosome 7 to chromosome 9, near the T-cell receptor beta locus (GenBank, AF029308) (data not shown). This band was not detectable in controls, in which reverse transcriptase was omitted from the reaction mixture (data not shown). Therefore, we concluded that in PC324 and PC3 the pseudogene was expressed at a low level. Because PTEN pseudogene expression was low, or undetectable in the four tumors with chromosome 10q23 homozygous deletions, we presumed that the RT-PCR product from PTEN positive tumors was derived from the original PTEN gene. This was checked for PC295 (see below) and PC135. PTEN mRNA expression in PC135 is relatively low.

Digestions by slected restriction enzymes showed that most, if not all of the fragment amplified by RT-PCR was derived from wild type *PTEN* (data not shown).

Sequencing of a 530 bp exon 1 to exon 7 RT-PCR product of PC295 *PTEN* mRNA, confirmed the complete absence of exon 5 sequences in the transcript (Fig. 3B, and data not shown). The co-amplified 760 bp fragment was not derived from the human *PTEN* mRNA or the pseudogene, but from *PTEN* mRNA of mouse cells present in the transplanted tumor (data not shown).



Figure 3. *PTEN* mRNA expression in prostate cancer cell lines and xenografts. (A) Agarose gel analysis of a 620 bp RT-PCR amplified fragment. Primers used are located in exons 8 and 9, and are specific for human *PTEN*. (B) *PTEN* RT-PCR analysis of PC295 RNA. Primers used are located in exons 1 and 7. PC310 RNA was used as a control. Primers detect both human and mouse *PTEN* mRNA.

Structural Analysis of the PTEN Gene.

From *PTEN* positive DNAs, all nine exons were analyzed for mutations by PCR-SSCP. Aberrant SSCP bands were sequenced. In four DNA samples, *PTEN* mutations were unambiguously established (PC82, PC346, PC374 and LNCaP). The exon 1, codon 6 AAA to A frame-shift mutation in LNCaP has been described previously (Li et al. 1997; Steck et al. 1997). The other three mutations are depicted in Fig. 4. PC82 *PTEN* contained a CAA to TAA nonsense mutation at codon 87; PC346 a CGA130TGA nonsense mutation, and PC374 a TAT76T frame-shift, directly resulting in a TGA stop codon at position 76. All mutations will lead to the synthesis of a truncated protein, lacking the phosphatase domain. In none of the sequenced samples the corresponding wild type sequence was detected, confirming that the second *PTEN* allele was deleted. Both LNCaP and TSU *PTEN* were found to contain GGT(Gly) instead of GGC(Gly) at codon 44. This presumed rare polymorphism could not be detected in 34 control DNAs from healthy individuals (data not shown). The previously described ATG134TTG missense mutation in DU145 *PTEN* could not be confirmed (Li et al. 1997).



Figure 4. Sequence analysis of *PTEN* mutations in prostate cancer xenografts. In PC82 and PC346 nonsense mutations were detected; PC374 shows a TA deletion at codon 76, resulting in a TGA stop codon.

DISCUSSION

In this study we describe five homozygous deletions, two frame-shift mutations and two nonsense mutations in fifteen prostate cancer cell lines and xenografts. Homozygous deletions were established by the complete absence of polymorphic markers in the *PTEN* region and by failure of PTEN exon amplifications. RT-PCR showed *PTEN* expression in all tumors, in which the gene was not deleted.

The four tumors containing mutated *PTEN* (PC82, PC346, PC374 and LNCaP) showed one amplified band for all five polymorphic markers at 114 cM, suggesting deletion of one chromosome 10 copy at these sites. Absence of wild type *PTEN* at the mutated site confirmed this observation. Complete or almost complete deletion of both *PTEN* copies was detected in five tumor samples. The homozygous deletion in PC3 is large [>5cM, Fig. 1; see also (Li et al. 1997)], that in PC295 is probably less than 20 Kbp as judged from preliminary data of the lengths of introns 4 and 5.⁵ In PC133, PC324 and PCEW, two or more markers at 114 cM are deleted. Minimal deleted regions in these tumors are estimated to be at least 100 Kbp.

Five out of six tumors, in which we could not find *PTEN* homozygous deletions or mutations (PC135, PC329, PC339, TSU and DU145), showed apparent allelic losses for the markers *PTENCA* and *AFMAa086WG9*, which are located within the *PTEN* gene. Three of these DNAs even showed one band for all five markers that map at 114 cM (Fig. 1). Although it cannot be completely ruled out, that a small proportion of point mutations in *PTEN* was missed, it is also possible that in these tumors a second *PTEN* deletion or mutation has not occurred, or that a gene different from *PTEN* is mutated. For

tumors with a large deletion, the *MXI1* gene at 10q24-25 might be a candidate, as suggested previously (Eagle et al. 1995). However, others exclude a major role of *MXI1* in prostate cancer (Gray et al. 1995).

Additionally to deletions and mutations, down-regulation of expression by promoter methylation, mutation or other processes can be a mechanism of gene inactivation. Xenografts and cell lines have the advantage that mRNA expression can easily be monitored. The RT-PCR experiments (Fig. 3A) clearly showed that down-regulation of *PTEN* expression is not a common mechanism of *PTEN* inactivation. An exception might be PC135, which shows a considerably lower *PTEN* mRNA level than the other samples. Additional experimental evidence must be collected to prove the physiological implication of this observation. The processed *PTEN* pseudogene, which most likely is expressed at a low level complicates expression studies. Previously, it has mistakenly been described as a mutated *PTEN* gene in breast cancer (Rhei et al. 1997).

Mutations or deletions were found in six out of nine xenografts/cell lines derived from metastatic sites and in three out of six, derived from primary tumors, obtained by prostatectomy or transurethral resection. Original tumor DNAs were available for PC346 (primary tumor) and PC374 (metastasis). In both samples the mutations as shown in Fig. 4 could be confirmed (data not shown), indicating that they were not introduced during propagation in nude mice.

During the course of our study, *PTEN* mutations were reported in primary and metastatic prostate cancer tissues (Cairns et al. 1997; Teng et al. 1997; Suzuki et al. 1998b). In the study of Cairns et al. (Cairns et al. 1997), in 3/60 primary cancers and in 7/20 lymph node metastases mutations or deletions were found. Teng et al. (Teng et al. 1997) could not detect *PTEN* mutations in six primary tumors; Suzuki et al. (Suzuki et al. 1998b) described *PTEN* abnormalities in twelve of nineteen tumors obtained from metastatic sites during autopsy. These numbers might be underestimations, because small deletions would have been missed. Like found in the present study, deletions, point mutations and frame-shifts were detected. In prostate cancer, *PTEN* mutations lead, almost without exception, to the synthesis of a truncated protein. The CGA130TGA nonsense mutation and the TAT76T frame-shift have now been reported in different studies in both glioblastomas and prostate cancer, indicating hot spots of mutation (Rasheed et al. 1997; Steck et al. 1997; Teng et al. 1997; Wang et al. 1997 and this study).

56 | Chapter 2

Most frequently, *PTEN* aberrations were found in DNAs from metastatic disease (Cairns et al. 1997; Suzuki et al. 1998b and this study). However, *PTEN* abnormalities do also occur at the primary tumor site (Cairns et al. 1997). Not unexpected, in the cell lines and xenografts derived from the primary tumor site, as studied here, the percentage of *PTEN* abnormalities is much higher than that found at the primary tumor site by Cairns et al. (Cairns et al. 1997) (3/6 and 3/60, respectively). The difference might be explained by a growth advantage of the original xenografted material in case of the absence of functional *PTEN*.

Because *PTEN* is frequently completely absent or severely truncated, immunohistochemical staining, utilizing specific antibodies could be applied to address in more detail the question whether *PTEN* is a tumor progression marker. Specific antibodies are also additional tools for the detection of the complete spectrum of *PTEN* mutations and deletions in patient samples, including small deletions such as found in PC295. Both complete absence and aberrant cellular distribution of the PTEN protein can be expected.

So far, *PTEN* is the most widely mutated tumor suppressor gene in prostate cancer. Therefore, further elucidation of its function in prostate cancer is of utmost importance.

ACKNOWLEDGMENTS

We thank Dr Leen Blok, Mrs Angelique van der Made and Mrs Nicole Groen for RNA preparations.

int line

and inte

CHAPTER 3

CHROMOSOME 10 ALTERATIONS IN PROSTATE CANCER XENOGRAFTS AND CELL LINES: CANDIDATE TUMOUR SUPPRESSOR GENES AND THE *PTEN* FLANKING GENES

100

Karin G. Hermans, Dirk C. Van Alewijk, Wytske van Weerden, and Jan Trapman

Submitted for publication

international in

ABSTRACT

Detailed genetic studies are a prerequisite in deciphering the molecular mechanisms of tumour cell growth. Xenografts and cell lines are excellent starting materials for such studies. We examined 11 xenografts and 4 cell lines derived from human prostate cancers for chromosome 10 alterations. Comparative Genomic Hybridisation (CGH) revealed a pattern of loss of distal 10p (3/15), gain of proximal 10p and proximal 10g (5/15), and loss of distal 10g (6/15). Allelotype analysis confirmed the CGH data in all samples, which contained 10p deletions (PC310, PC324 and PC3) and in 5 samples with large 10g deletions (PC133, PC295, PC339, PC374, PC3). Previously, we identified PTEN inactivation by homozygous deletion or mutation in 9 of the 15 xenografts and cell lines (Vlietstra et al. 1998). The present study excludes MXI1 and DMBT1 on distal 10g, and KLF6 on distal 10p as important tumour suppressor genes in prostate cancer. Remarkably, study of the PTEN flanking region by allelotype analysis frequently predicted a very small region of allelic loss of 5.8 Mbp or less with or without PTEN inactivation. These regions were not detected by CGH. Sizes of homozygous deletions around PTEN ranged from approximately 1.2 Mbp (PC133) to less than 30 Kbp (PTEN exon 5 in PC295). In both PTEN positive and PTEN negative samples, 16 genes mapping in the PTEN region, were investigated for deletions and expression levels. Loss of 1 or 2 copies of PTEN was almost always accompanied by loss of the distal flanking gene FLJ11218 and the proximal flanking genes MINPP1, PAPSS2 and FLJ14600. Furthermore, differential expression was detected for FLJ11218 and PAPSS2. Four genes were selected for mutation analysis, MINPP1, which like PTEN metabolises phospholipids, the sulfatase PAPSS2, the tumour necrosis factor receptor TNFRSF6 and FLJ11218. Complete deletion or inactivating mutation of PAPSS2 was found in at least 3 samples. MINPP1 was deleted in PCEW, but not mutated in other samples. TNFRFS6 structure was normal in all samples. Additional to 4 homozygous deletions, 1 missense mutation was detected in FLJ11218. In conclusion, our data provide evidence for inactivation of both PTEN alleles as the major genetic defect on chromosome 10 in prostate cancer cell lines and xenografts. PTEN haplo-insufficiency is observed in 1 or 2 samples. PTEN inactivation is in part of the samples accompanied by loss of one MINPP1 allele, loss of one copy, mutation or low expression of PAPSS2 and most frequently with mono- or bi-allelic loss, or low expression of FLJ11218.

INTRODUCTION

Prostate cancer is the most frequently diagnosed cancer and the second leading cause of male cancer death in Western and Northern Europe, North America and Australia (Greenlee et al. 2001). At present, an adequate therapy of metastatic prostate cancer is not available. In order to identify novel therapeutic targets, knowledge of the major molecular alterations is urgently needed.

In prostate cancer, most frequent deletions were found for 6q, 8p, 13q and 16q, indicating the localization of tumour suppressor genes on these chromosomal arms (Visakorpi et al. 1995; Cher et al. 1996; Nupponen et al. 1998b; Alers et al. 2000). Less frequent chromosomal losses were found for 5q, 10q and 17p. The most frequently gained chromosome arm was 8q, followed by 7p, 7q and 20q.

Loss of 10q is generally considered as a late step in prostate cancer progression. Allelic imbalance studies indicated separate regions at 10q22-q26 to be affected, suggesting the inactivation of more than one tumour suppressor gene (Gray et al. 1995; Ittmann 1996; Komiya et al. 1996; Trybus et al. 1996; Cairns et al. 1997; Feilotter et al. 1998; Ittmann 1998; Leube et al. 2002). Loss of 10q is not unique for prostate cancer. Frequent loss of distal 10q has also been described in renal cell carcinoma (Morita et al. 1991), non-Hodgkin's lymphoma (Speaks et al. 1992), glioblastoma (James et al. 1988; Fujimoto et al. 1989), meningioma (Rempel et al. 1993), malignant melanoma (Reifenberger et al. 2000), small lung cell cancer (Kim et al. 1998), bladder cancer (Cappellen et al. 1997), and endometrial carcinoma (Peiffer et al. 1995; Nagase et al. 1996).

The *PTEN* gene at 10q23.3, which encodes a lipid and protein phosphatase, is the most frequently altered tumour suppressor gene in prostate cancer (Cairns et al. 1997; Teng et al. 1997; Feilotter et al. 1998; Gray et al. 1998; Vlietstra et al. 1998; Wang et al. 1998; McMenamin et al. 1999). Complete *PTEN* inactivation was detected at varying frequency in primary tumours and in up to 60% of metastases, cell lines and xenografts. *PTEN* is even more frequently implicated in glioblastoma (Li et al. 1997; Rasheed et al. 1997; Steck et al. 1997; Teng et al. 1997; Wang et al. 1997) and endometrial carcinoma (Tashiro et al. 1997), and to a lesser extent in many other tumours. Less is known about 10p alterations in prostate cancer. Variable frequencies of loss of distal 10p in prostate cancer have been described (Ittmann 1996; Trybus et al. 1996; Fukuhara et al. 2001; Narla et al. 2001). Additional to *PTEN*, 10q harbors the

candidate tumour suppressor genes *MXI1* at 10q25.2 and *DMBT1* at 10q26.2 (Eagle et al. 1995; Mollenhauer et al. 1997). Recently, mutation of *KLF6* on 10p15 in prostate cancer has been described (Narla et al. 2001; Chen et al. 2003).

Xenografts and cell lines are powerful tools in the search for genetic alterations in human cancer. They are available in unlimited quantities and, importantly, they lack normal cells of human origin, which simplifies the analysis of chromosomal alterations, and structural alterations and expression levels of individual genes. Previously, we described frequent *PTEN* inactivation in prostate cancer xenografts and cell lines (Vlietstra et al. 1998). In the present study we analyse the role of chromosome 10 in prostate cancer by CGH and allelotype analyses. We present data on the expression and structure of the candidate tumour suppressor genes *MXI1*, *DMBT1* and *KLF6*. In addition, we studied the structure, deletion and expression of *PTEN* flanking genes. Furthermore, we address the issue of *PTEN* haplo-insufficiency in prostate cancer.

MATERIALS AND METHODS

Prostate Cancer derived Cell Lines and Xenografts

The *in vitro* growing cell lines LNCaP, PC-3, DU-145 and TSU were cultured under standard conditions. The *in vivo* xenografts PC82, PCEW, PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346 and PC374 were propagated on male nude mice (Hoehn et al. 1980; Hoehn et al. 1984; Van Weerden et al. 1996).

DNA and RNA preparation

Genomic DNA from cell lines and xenografts was isolated utilizing the Puregene system from Gentra Systems (Minneapolis, MN) according to the procedure described by the manufacturer.

Cell line RNA was isolated by the guanidium isothiocyanate procedure; xenograft RNA was isolated by the LiCl protocol (Sambrook and Russell, 2001). mRNAs from fetal brain and normal prostate tissue were purchased from Clontech Labs (Palo Alto, CA).

Comparative Genomic Hybridization

CGH was performed essentially as described (Kallioniemi et al. 1992). In brief, tumour DNA and normal male reference DNA samples were labeled by nick translation

(Nick translation system, Life Technologies, Rockville, MD) with bio-dUTP (Roche Diagnostics, Almere, The Netherlands) and digoxigenin (Roche Diagnostics), respectively. Labelled DNA samples (200 ng each) and 15µg COT-1 DNA was ethanol-precipitated and dissolved in 10 µl hybridisation mix (50% formamid, 0.1% Tween-20, and 10% dextran sulfate in 2xSSC at pH7.0). The probe mixture was denatured (10 min, 72°C), pre-hybridised (30 min, 37°C) and hybridized to normal male chromosome spreads (72 h, 37°C). Next, slides were washed, and fluorescent detection of the biotin-and dioxigenin-labelled DNA probes was by fluorescein isothiocyanate (FITC)-conjugated avidin (Vector Labs, Burlingame, CA) and anti-digoxigenin-rhodamine (Roche Diagnostics) staining, respectively. Chromosomes were DAPI counterstained (4'6'-diamidino-2-phenylindole) (Sigma, St. Louis, MO) in Vectashield anti-fade solution (Vector Labs).

Images were acquired with an epifluorescent microscope equipped with a cooled CCD camera (Photometrics Inc., Tuscon, AZ), a triple-band pass beam splitter emission filters (P-1 filter set, Chroma Technology, Brattleboro, VT), and a Quips XL image analysis system (version 3.1 Vysis Inc., Downers Grove, IL). Chromosomal regions were scored as lost if the mean green to red ratio was below 0.85 and gained if this ratio was above 1.15. Eight or more metaphases were analysed per sample.

PCR and PCR-SSCP

Standard PCR amplifications utilizing *Taq* polymerase (Promega, Madison, WI) included 35 cycles of 1 min at 95°C, 1 min at 50°C or 55°C, and 1 min at 72°C. For allelotyping and PCR-SSCP, 1 μ Ci [alpha-³²P]dATP (Amersham, Buckinghamshire, UK) was added to a 15 μ I reaction mix. Amplified, radio-labelled polymorphic microsatellite markers were separated on a sequence gel. SSCPs of radio-labeled gene specific PCR products were analysed on a 6% non-denaturing polyacrylamide gel containing 10% glycerol. Gels were run at 7W overnight at room temperature. For detection of homozygous deletions, PCR amplifications were performed in a 50 μ I reaction volume. Amplified fragments were separated on a 2% agarose gel.

Allelotype analysis and screening for homozygous deletions

Polymorphic microsatellite markers applied for allelotype analysis and screening for homozygous deletions at 10p were: D10S602, D10S1745, CA237H5A, CA237H5B, D10S591, D10S1729, D10S189, D10S547, D10S191, D10S595, D10S197, D10S193), and at 10q: D10S220, D10S581, D10S537, D10S1688, D10S1730, D10S1686, D10S1687, CA163M19, D10S579, D10S215, D10S1765, AFMa086WG9, D10S541, CA13J3, CA80H5, D10S1753, D10S583, D10S1680, D10S1726, D10S192, D10S187, D10S209, D10S217. PCR primer sequences can be found in Genome Database (http://gdbwww.gdb.org/). Other primer sets are: CA237H5A: gcagagcagccttcagtaat and cacttggcaaactacagtgc; CA237H5B: caagagcatgagtcccattg and gaaccaatcagtcaccaagc; CA163M19: gttttgcccagttgaagtca and teetteeceaactattetate; CA13J3: gattageaeaacactgggtag and accetetggggaagtactat; CA80H5: accagattggatgtgcatgc and caaccagcagtatctgtcac. Positions of markers on chromosome 10 were derived from the April 2003 freeze of the UCSC human genome map (http://genome.ucsc.edu).

Primer sets utilized in screening for homozygous deletions of KLF6, MINPP1, PAPSS2, FLJ14600, FLJ11218, LIPF, DKFZp761K1824, ACTA2, TNFRSF6, CH25H, LIPA, IFIT2, IFIT4, IFIT1, RI58, PANK, MPHOSPH1 and MXI1 are available upon request.

The 74K, 36K, G14Ext, G14 and 60K primer sets for detection of homozygous deletions in *DMBT1* by PCR on genomic DNA are published previously (Mollenhauer et al. 1997). For more detailed analysis of the homozygous deletion in *DMBT1* by PCR-SSCP the primers DMBTME39-F (5'-ACTTCAGAGGTAGGAGGGT-3') and DMBTME39-R (AGGTAGAGAGTGAGCCCTAG-3') were utilized.

mRNA expression

Analysis of mRNA expression was performed by semi-quantitative RT-PCR. cDNA was synthesized on 1 μ g RNA template utilizing 200 U M-MuLV-RT (Life Technologies) and a T₁₂-site primer (5-GCATGCGAATTCGGATCCT₁₂-3) in a buffer, containing 10 mM DTT, 1 mM dNTPs, and 40 U RNAsin (Promega) for 1 h at 37°C. RNA polymerase II was utilized as a control. Specific cDNA fragments were amplified by standard PCR. Gene specific RT-PCR primers are available upon request.

Structures of candidate tumour suppressor genes and PTEN flanking genes

For PCR-SSCP analysis, fragments of appropriate sizes were amplified of all exons of *MINPP1, PAPSS2, TNFRSF6, FLJ11218, MXI1* and *KLF6*. Primer sequences are available upon request. Selected amplified fragments were purified over QIAquick spin columns (Qiagen, Hilden, Germany), cloned into pGEM-T Easy (Promega), and sequenced according to the dideoxy chain termination method (Sambrook and Russell, 2001).

RESULTS

Alterations of chromosome 10 are among the most frequent chromosomal changes in prostate cancer as determined by allelotype analysis (Gray et al. 1995; Komiya et al. 1996; Trybus et al. 1996; Ittmann 1998; Leube et al. 2002). However, this is not the case for several CGH studies (Nupponen et al. 1998b; Alers et al. 2000). To increase our insight into the role of chromosome 10 in prostate cancer, we studied overall chromosome 10 alterations in prostate cancer xenografts and cell lines by both CGH and allelotype analysis.

Comparative Genomic Hybridisation

Eight out of 15 xenografts and cell lines showed gain or loss of specific parts of chromosome 10 by CGH (Figure 1). Deletion of the distal region of 10p was found in PC310 (p13-pter), PC324 (p14-pter) and PC3 (p13-pter). Small changes at the telomeres were not taken into account, because of limited reliability. The majority of amplifications were found around the centromere: PC135 (q11.2-q22), PC324 (p11.2-p12 and q21-q24), PC339 (p11.2-p14), PC374 (p12-q21) and PC3 (p11.2-q22). Losses of distal 10q were most frequent. They were present in PC133 (q22-qter), PC135 (q23-qter), PC295 (q21-q23), PC339 (q23-qter), PC374 (q22-q25) and PC3 (q23-qter). In PCEW, PC82, PC329, PC346, LNCaP, TSU and DU145 no clear chromosome 10 alterations were detected by CGH. Summarizing, chromosome 10 was found to be frequently altered in prostate cancer xenografts and cell lines with a characteristic pattern of loss of distal 10p, gain of proximal 10p and proximal 10q, and loss of distal 10q.



Figure 1. Chromosome 10 alterations identified by comparative genomic hybridisation in prostate cancer xenografts and cell lines. A bar to the right of a chromosome ideogram indicates gain, a bar to the left indicates loss of the chromosomal region.

Allelotype analysis of chromosome 10

To obtain more detailed information on chromosome 10 alterations, the 15 genomic DNAs were screened for 35 polymorphic microsatellite markers along both chromosomal arms. A high marker density was chosen in a small region around the *PTEN* locus, which maps at 89.8 Mbp from the top of the p-arm, and around *KLF6* at 3.9 Mbp. The results are summarized in Figure 2. The previously published *PTEN* alterations are indicated at the bottom of Figure 2 (see also Vlietstra et al. 1998). Positions of markers were taken from the UCSC genome browser.

Because matching normal DNA samples were not available for comparison, 5 consecutive mono-allelic bands of highly polymorphic markers was taken as indicative for loss of one copy of the corresponding chromosomal region. According to this

definition, complete loss of one copy of 10p was found for TSU. Loss of distal 10p, including *KLF6*, was detected in PC310, PC324 and PC3. Large regions of apparent loss at 10q were present in PC133, PC295, PC339, PC374 and PC3. A remarkable large number of small regions of loss of 10q23.3 were detected in samples with or without complete *PTEN* inactivation (PCEW, PC82, PC135, PC324, PC346, LNCaP and DU145). Homozygous deletions and mutation of *PTEN* were found both in small and large regions of allelic loss (PCEW, PC82, PC324, PC346 and LNCaP, and PC133, PC295, PC374 and PC3, respectively; see Figure 2). In PC346, PC374, LNCaP and DU145, several polymorphic markers showed microsatellite instability (MSI), which limited the accuracy of allelotype analysis. Gains, which might be scored by allelic imbalance, were not taken into account.

Comparison of comparative genomic hybridisation and allelotype analysis

In general, large 10q deletions, as detected by CGH matched with allelotyping experiments. However, in contrast to allelotype analysis, CGH showed amplification of proximal 10q and deletion of distal 10q in PC135. CGH did not clearly detect loss of proximal 10q in PC339. The TSU cell line showed complete loss of one copy of chromosome 10 by allelotype studies and no loss by CGH. The large number of small mono-allelic regions or homozygous deletion of the *PTEN* locus explains the limited accuracy of CGH of 10q.

Marker	Heterozygosity	and	Mbp (Santa Cruz*)	CEW	PC82	oC133	PC135	PC295	oC310	oC324	oC329	C339	oC346	C374	ŝ	NCaP	ISU	JU145
D10S602	73	10p15.3	22	2	2	1	2	1	1	1	1	2	2	2	1		1	MSI
D10S1745	85	10p15.3	2.5	2	2	1	2	2	1	1	2	1	2	MSI	1	MSI	1	MSI
CA237H5A	-	10p151	3.9	1	2	1	1	2	i.	1	1	2	2	1	1	2	1	2
CA237H5B	-	10p15.1	4.0	2	2	1	2	2	1	1	2	2	1	MSI	1	- MSI	1	MSI
D10S591	71	10p151	4.5	1	1	2	1	2	i.	1	2	2	1	2	1	MSI	1	1
D10S1729	72	10p151	4.9	2	2	1	1	2	i.	1	2	1	2	- MSI	1	2	1	MSI
D10S189	73	10p14	6.9	2	1	1	2	1	1	1	2	1	2	2	1	2	1	1
D10S547	74	10p14	10.7	1	1	2	2	1	i.	2	2	2	2	2	1	2	1	1
D10S191	81	10p13	14 7	1	1	1	2	2	i.	2	2	2	2	- MSI	1	- MSI	1	MSI
D10S595	85	10p12 31	20.8	2	2	2	2	2	2	2	2	2	2	2	i	MSI	÷	MSI
D10S197	75	10p12.01	26.7	1	2	2	2	2	2	2	1	2	MSI	MSI	1	2	1	MSI
D10S193	81	10p11.1	30.7	2	1	2	2	2	2	2	2	1	2	2	i	2	÷	MSI
D10S220	84	10g11.23	52.2	1	2	1	1	2	2	1	2	1	2	2	2	1	1	MSI
D10S581	80	10g213	65.7	2	1	1	1	2	1	1	1	1	1	2	1	2	1	MSI
D10S537	83	10g221	72.3	2	2	1	i.	1	2	2	2	1	MSI	1	i	MSI	1	MSI
D10S1688	86	10q22.1	72.5	1	1	2	i	i	1	2	2	1	2	1	1	MSI	1	1
D10S1730	83	10022.3	78.8	1	2	2	i.	1	2	1	1	1	MSI	MSI	i	2	1	MSI
D10S1686	86	10g23.1	85.7	2	2	1	2	÷	2	2	2	1	2	MSI	i	2	÷	MSI
D10S1687	81	10q20.1	88.8	2	2	÷.	2	÷	2	2	2	2	2	1	4	1	÷	2
CA163M19	-	10q23.2	89.1	1	2	÷.	2	÷	2	2	2	2	1	÷	1	÷.	÷	MSI
D109579	59	10q20.2	89.5	0	1	÷.	1	÷	1	1	1	2	4	÷	4	÷.	÷	1
D100375	81	1002331	80.6	0	4	÷	÷	÷	2	÷	2	2	4	4		÷	÷	4
D105215	83	10q23.31	89.7	0	i I	0	L i	4	2	÷.	2	2	4	4	1	÷.	÷	4
AEM2086WG0		10023.31	80.8	0	1	0	4	4	1	0	1	1	1	1	0	ı.	4	4
D109541	78	10023.31	09.0	0	1	0	4	4	2	0	2	4	4	1	0		4	4
C 413 13	70	10023.31	90.1	2	2	0	4	4	1	1	2	4	2	1	1		4	4
CAROLLE	-	10023.01	01.5	2	4	1	2	4	2	2	1	4	2	2			4	2
D10S1753	- 74	10023.31	91.5	1	2	4	1	4	2	2	1	4	2	2		1	4	1
D1051733	0/	10923.31	92.5	4	2	4	2	4	2	4	4	4	2	4			4	
D105363	04 02	10923.33	94.5	2	2	4	2	2	2	4	2	4		4			4	I MCI
D1031080	76	10423.33	100.0	2	4	4	0	2	2	-	-	4	MOI				4	MOI
D1051720	70	10q24.2	100.0	2	2	4	2	2	2	2	2	4	MGI	11/101		10101	4	MOI
D105192	70	10q24.31	102.0	2	2	4	2	1	2	2	2	4					4	IVISI
D105187	04 74	10q25.3	100.0	2	2	4	1	2	2	2	2	4	2	10101			4	IVISI
D105209	74	10q26.12	122.4	2	2			2	2	1	2		2			2		IVISI
D105217	81	10q26.2	129.5	1	2	20	I	2	2	2	2		10151	10151		2		10151
April 2003 fi	eeze	Genome	; prov	vser	003	50												
	PTEN																	
PTEN		10q23.31	89.8	-	-	-	+	-	+	-	+	+	-	-	-	-	+	+

Figure 2. Chromosome 10 alterations identified by allelotype analysis in prostate cancer xenografts and cell lines. If two bands of different length were detected, two allelic forms were retained in the DNA. One band indicates the presence of one allele or two alleles of identical length. One and two allelic forms are represented by "1" and "2", respectively. The homozygous deletions in PCEW, PC133, PC295, PC324, and PC-3 are represented by "0". The status of the *PTEN* gene is shown at the bottom of the figure. (+) indicates wild-type *PTEN*; (-) indicates inactivated *PTEN*.

Analysis of candidate tumour suppressor genes

Chromosomal studies (see Figures 1 and 2; Ittmann 1996; Komiya et al. 1996; Trybus et al. 1996; Leube et al. 2002) indicated that PTEN might not be the only tumour suppressor gene on chromosome 10. Distal to PTEN the candidate tumour suppressor genes MXI1 (10q25.2) and DMBT1 (10q26.2) have been mapped (Edelhoff et al. 1994; Shapiro et al. 1994; Mollenhauer et al. 1997). MXI1 antagonizes MYC in modulation of gene expression and tumorigenesis (Lahoz et al. 1994). DMBT1 easily recombines in cancer cells and might play a role in immune defense and epithelial cell differentiation (Mollenhauer et al. 1997; Mollenhauer et al. 2000). We examined the expression and structure of both genes in the 15 prostate cancer cell lines and xenografts. No homozygous deletions of MXI1 were detected, and MXI1 mRNA was present in all RNA samples (data not shown). Structural analysis revealed polymorphisms, but no somatic mutations in the MXI1 gene. One polymorphism was in the open reading frame (GGC-GGT, G23G), all others were in intron sequences. In the DMBT1 gene, an intragenic homozygous deletion of the markers G14EXT and G14 was found in PC135 (Figure 3A). This homozygous deletion was further examined by PCR-SSCP, utilizing a primer set, which amplified the repeat units in DMBT1 (Figure 3B; Mollenhauer et al. 1999). Sequencing of the amplified fragments combined with the presence of marker 36K (Figure 3A) indicated that the maximum size of the homozygous deletion in DMBT1 is between exons 16 and 27 (Figure 3C). No deletions were detected in the other 14 DNA samples. Semi-quantitative RT-PCR indicated low expression of DMBT1 in PC82 and PCEW (data not shown).

Recently, inactivating mutations in *KLF6* on 10p15 has been described in prostate cancer (Narla et al. 2001; Chen et al. 2003). However, examination of *KLF6* in the 15 xenografts and cell lines revealed clear expression and absence of homozygous deletions or inactivating mutations (data not shown).



Figure 3. Characterization of the homozygous deletion in *DMBT1* in PC135. (A) Agarose gel analysis of the indicated amplified *DMBT1* genomic fragments. (B) PCR-SSCP of the repeat units of *DMBT1* in prostate cancer xenografts and cell lines. (C) Schematic representation of the deleted segment in *DMBT1*.

Loss of the PTEN flanking region at 10q23.3

In mouse prostate cancer models not only complete *PTEN* inactivation, but also *PTEN* haplo-insufficiency has been implicated in tumorigenisis (Di Cristofano et al. 2001; Kwabi-Addo et al. 2001; Kim et al. 2002b; You et al. 2002). The role of *PTEN* haplo-insufficiency in human prostate cancer is not well defined. Investigation of the *PTEN* locus (Figure 2), compared with *PTEN* inactivation data (Figure 2, bottom; Vlietstra et al. 1998) learned that the xenografts PC310 and PC329 contain two wild-type *PTEN* alleles. In 9 DNA samples both *PTEN* alleles were inactivated by homozygous deletion (PCEW, PC133, PC295, PC324 and PC3) or deletion of one allele combined with a point mutation (PC82, PC346, PC374 and LNCaP). Loss of 1 copy of *PTEN* might have occurred in PC135, PC339 and DU145 (Figure 2). TSU was not taken into account,

because all chromosome 10 markers showed 1 allelic form. More detailed studies also excluded PC339 (see below). So, *PTEN* haplo-insufficiency might be present in 2 out of 15 xenograft and cell line DNAs.

Flanking genes might be complementary or independent candidate genes involved in prostate cancer. We searched 16 genes flanking *PTEN* for homozygous deletions, and the borders of small regions of allelic loss or homozygous deletion were accurately mapped. The order of genes and candidate genes was taken from the UCSC genome browser (see also Figure 5A). Figure 4 illustrates the genes at 10q23.3, which were completely deleted in the xenografts and cell lines, as determined by PCR. The first distal *PTEN* flanking gene, *FLJ11218*, was completely or partially deleted in all samples with *PTEN* deletion, except for PC295. *MINPP1, PAPSS2, FLJ14600* and *LIPF* were deleted in part of the samples. None of the DNAs without complete deletion of *PTEN* contained a homozygous deletion of one or more of 16 flanking genes (data not shown; see for genes investigated Materials and methods and Figure 5A).



Figure 4. Genes inactivated by homozygous deletion in the prostate cancer xenografts PCEW, PC133, PC295 and PC324 and the cell line PC3. The figure indicates agarose gel electrophoresis of amplified exons of the indicated genes. The positions of the genes on chromosome 10 in Mbp from the top are indicated at the right.
Figure 5 summarizes the calculations of the lengths of homozygous deletions and mono-allelic regions in the PTEN region. These calculations not only took into account the outcome of microsatellite repeat analyses, but also single nucleotide polymorphisms in MINPP1, PAPSS2, FLJ11218 and TNFRSF6 genes (see Materials and Methods). The homozygous deletions in PCEW, PC133, PC295, PC324 and PC3 ranged in size from more than 1.2 Mbp (PC133) to less than 30 Kbp (PC295) (Figure 5B, see also Figures 2 and 4). In both PC324 and PCEW, the telomeric border of the deletion is in intron 5 of FLJ11218. In PCEW, the deletions in both copies of 10g are small and almost identical in lengths. In PC324, the mono-allelic region around PTEN is less than 2 Mbp. The monoallelic regions in PC82 and PC346, which both contain a PTEN point mutation, and in DU145 and PC135, which do not show structural alterations of PTEN, are also small (Figures 2 and 5C,D). So, in many samples small mono-allelic regions around the PTEN locus could be precisely determined. All samples had in common that they lost one or two copies of PTEN. Most DNAs lost one or two copies of FLJ11218 and one copy of MINPP1, PAPSS2 and FLJ14600. Most likely, PC339 is an exception, because the mono-allelic region starts in FLJ11218, distal from PTEN.



Figure 5. Schematic representation of the homozygous deletions and mono-allelic regions flanking the *PTEN* locus in prostate cancer xenografts and cell lines. (A) Genes flanking *PTEN*. Data are from the April 2003 freeze UCSC gene map. (B) Homozygous deletions (open bars) and mono-allelic regions (gray bar) in PC295, PC324, PCEW, PC133 and PC3. Grey blocks indicate unknown border (C) Mono-allelic regions (gray bars) in PC82, PC346, PC374 and LNCaP. A cross indicates *PTEN* inactivation by point mutation. (D) Mono-allelic regions in PC135, DU145, PC339 and TSU, which lack complete *PTEN* inactivation.

Expression and structure of genes flanking the PTEN locus

To investigate further a possible contribution of PTEN flanking genes in prostate cancer, the expression patterns of 14 genes in 3 Mbp flanking the PTEN locus, bordered by the polymorphic markers D10S1687 and D10S1753 (see Figure 5A) were analysed by semi guantitative RT-PCR The results are summarized in Table I. Two 1 exon genes, DKFZp761K184 and CH25H were omitted. MINPP1, PAPSS2 and FLJ11218 expressions are shown as examples in Figure 6. A mixed pattern of expression profiles was found. Obviously, in all examples except for PTEN in PC295, homozygous deletion of a gene correlated with absence of expression. In the semi-quantitative RT-PCR no clear-cut correlation was found between the copy number and expression level of a gene, indicating gene-copy independent regulatory mechanisms to be more important. The stomach-specific gene LIPF was hardly expressed in the prostate tumour cells. Most other genes showed a rather stable expression level. Potentially interesting variable expression patterns were seen for PAPSS2, FLJ11218, and the interferon-regulated gene family IFIT1, 2, 4, RI58. High expression of the latter gene family almost perfectly correlated with androgen independence of a xenograft or cell line (PC82 is the exception; see Table I). PAPSS2 not only showed absence of expression due to homozygous deletion of the gene (PCEW), but expression was also undetectable or very low in PC133, PC324, PC329 and LNCaP. FLJ11218 expression was not only absent in the 4 samples, where the gene was completely deleted, but also absent or very low in PC135, PC346, LNCaP and TSU.



Figure 6. Expression of MINPP1, PAPSS2 and FLJ11218 mRNA in prostate cancer xenografts and cell lines. Expression was monitored by agarose gel electrophoresis of semi-quantitative RT-PCR products, indicated by "+" or "-" reverse transcriptase. RNA pol II was used as a control.

Table I

Expression of the PTEN flanking genes in prostate cancer xenografts and cell lines

	Xenografts											Cell lines			
	PCEW	PC82	PC133	PC135	PC295	PC310	PC324	PC329	PC339	PC346	PC374	PC3	LNCaP	TSU	DU145
Androgen dependent mRNA	+	+	-	-	+	+	-	+	-	+	+/-	-	+	-	-
MINPP1	-	+++	+++	+++	++	+++	+++	+++	+++	+++	+++	+++	++	+++	+++
PAPSS2	-	+++	+/	+++	+++	+++	-	+/-	+++	+	++	++	+/	+++	+++
FLJ14600	-	+++	+	+++	+++	+++	+++	+++	++	+++	+++	+++	+++	+++	+++
PTEN	-	+++	-	+	+++	+++	-	+++	++	+++	+++	-	+++	++	+++
FLJ11218	-	+++	-	-	+++	++	-	+	++	-	+	-	+/	-	++
LIPF	+/-	+	-	-	+	-	-	-	-	-	+/-	-	-	-	-
ACTA2	++	++	-	++	++	++	++	++	++	-	++	++	++	++	++
TNFRSF6	+++	+++	+++	+++	+++	+++	+++	+++	+++	+	+++	+++	+++	+++	+++
LIPA	+++	+++	+++	+++	+++	+++	+++	+++	+++	+/-	+++	+++	+++	+++	+++
IFIT2	+/-	+	+	++	+/-	+	+++	+/-	+	+/-	+++	+++	+/-	+	++
IFIT4	-	++	++	++	-	+/-	+++	+/-	++	-	++	+++	-	+	+++
IFIT1	-	++	++	+++	-	+	+++	-	+++	+/-	+++	+++	+/-	+++	+++
RI58	-	++	++	++	-	+	+++	-	++	-	++	+++	+	++	++
PANK	++	++	++	++	++	++	++	++	++	++	++	++	++	++	++
MPHOSPH1	+++	+++	+++	+++	+++	+++	+++	+++	++++	+++	+++	+++	+++	+++	+++

Four flanking genes, which were selected on the basis of frequent deletion and low expression (*FLJ11218*) or function, the phospholipid phosphatase *MINPP*, the sulfatase *PAPSS2* and the tumour necrosis factor receptor *TNFRSF6*, were searched for point mutations. In *MINPP1* and *TNFRSF6*, polymorphisms, but no somatic mutations were detected. In *PAPSS2* a polymorphism, which is presumed to decrease its function (Met295 in PC133; see Xu et al. 2002b), one frame-shift (deletion C in codon 355 in LNCaP) and a missense mutation [GTG to ATG (V45M) in PC346] were found. In *FLJ11218* we frequently observed the polymorphism GAG/GAC (E37D) in exon 1, and polymorphisms in introns 4 and 6. In addition, we detected in PC374 the missense mutation ATT to AGT (N232S) in exon 5.

Summarizing, inactivation of *PTEN* always paralleled low expression or inactivation of 1 or 2 copies of the telomeric flanking gene *FLJ11218*. Less frequent alterations in *PAPSS2* copy number, structure or expression were detected. In approximately half of the DNAs one copy of *MINPP1* was lost.

DISCUSSION

In this study, we describe the genetic characterization of chromosome 10 in xenografts and cell lines derived from human prostate cancer. Although it can be argued that xenografts and cell lines acquire extra genetic alterations during *in vivo* and *in vitro* culturing, it is our experience (Vlietstra et al. 1998; Trapman, unpublished data), that as far as we were able to check for xenografts, most genetic alterations are already present in the tumour tissue from which they are derived. We choose for xenogafts and cell lines because they lack normal human cells, which enabled accurate study of homozygous deletions, regions of allelic loss and gene expression.

First, overall chromosome 10 losses and gains were investigated by CGH. Our findings indicated a pattern of loss of distal 10p, gain of proximal 10p and 10q and loss of distal 10q. Previous CGH studies of DNA from prostate cancer patients showed a similar pattern, although at a lower frequency and less clear (Cher et al. 1996; Nupponen et al. 1998b; Alers et al. 2000). Like in this study, in previous studies loss of distal 10q was the most frequent chromosome 10 alteration. Absence of chromosome 10 alterations in LNCaP and TSU, and the loss-gain-loss pattern in PC3 were in accordance with previous CGHs (Nupponen et al. 1998a). In PC3, chromosome 10

76 | Chapter 3

seems to be involved in several translocations (Pan et al. 1999). A frequently detected breakpoint in PC3 and other prostate cancer cell lines maps at band 10q22 (Pan et al. 2001). A similar breakpoint at 10q22, combined with amplification of centomeric 10q and loss of telomeric 10q might be present in xenograft PC135. A recent CGH study on a different panel of xenografts provided an overall picture as shown here, loss of distal 10p and distal 10q and gain of the middle part of chromosome 10 (Laitinen et al. 2002).

Allelotype analysis confirmed loss of distal 10p in PC310, PC324 and PC3. We did not find alterations in *KLF6* at 10p15, as previously reported (Narla et al. 2001; Chen et al. 2003). Although it can be argued that the number of samples studied is small, we favour the hypothesis that the most frequently affected tumour suppressor gene on 10p remains to be detected.

CGH data did not match allelotype data for 10p and 10q in TSU. No chromosome copy change was seen in CGH, but allelotype analysis showed one allelic form for all markers. This is strong evidence for chromosome 10 isodisomy in this cell line. A similar observation has been made for chromosomes 6 and 8 (Verhagen et al. 2002; Van Alewijk, unpublished), suggesting defective chromosome segregation in TSU.

CGH data for 10q are less informative than allelotype studies. Although in some DNAs hampered by microsatellite instability, allelotype analysis was more informative, due to the occurrence of many small regions of chromosomal loss at the *PTEN* locus. Six small deletions were not detected by CGH. It can be assumed that CGH of prostate cancer tissues also frequently misses loss of *PTEN* alleles, underestimating complete inactivation or haplo-insufficiency of *PTEN* in patients.

Allelotype analysis of 10q23.3 in tumours from patients frequently showed a higher percentage of allelic loss at the *PTEN* locus than *PTEN* alterations (Feilotter et al. 1998; Pesche et al. 1998). There might be several explanations for this discrepancy. First of all, complete inactivation of *PTEN* by point mutation or homozygous deletion might have been missed, because many studies were incomplete. Secondly, it has long been thought that in these samples not *PTEN*, but a more distal gene was inactivated. Although this cannot be excluded completely, our present study supports the assumption that this is not the case. We could not detect alterations in *MXI1*, a gene at 10q25.2, previously implicated at low frequency in prostate cancer (Eagle et al. 1995; Prochownik et al. 1998). Neither did we detect at

high frequency deletions in the unstable *DMBT1* gene, which has been described as a candidate tumour suppressor gene in several other tumour types (Mollenhauer et al. 1999). We also could not find complete inactivation of one of the *PTEN* flanking genes in the absence of *PTEN* alteration. Therefore, an attractive alternative is *PTEN* haplo-insufficiency as the underlying genetic defect on 10q in early stage prostate cancer. *PTEN* haplo-insufficiency seems to be present in only 1 xenograft (PC135) and 1 cell line (DU145), which represent late stage prostate cancer. Careful examination of DNA from micro-dissected prostate cancer samples with a variety of experimental approaches should be carried out to address this important question. Complementary to mutation analysis, array CGH on a distal 10q tilted BAC contig should provide this important information. In favour of an important role of *PTEN* haplo-insufficiency in human prostate cancer are mouse model studies. In 4 prostate cancer models based on prostate specific SV40-Tag expression (TRAMP model), inactivation of *Ink4a/Arf*, *Cdkn1b* or *Nkx3.1 Pten* haplo-insufficiency contributed to tumorigenesis (Di Cristofano et al. 2001; Kim et al. 2002b; You et al. 2002).

Although we could not find evidence for a PTEN independent role of flanking genes in tumourigenesis, a complementary role cannot be excluded. Two observations are important in this regard: the high frequency of *PTEN* inactivation by homozygous deletion, and the remarkable large number of small mono-allelic regions that accompany PTEN inactivation. The latter observation limits the candidate genes to those directly flanking PTEN. We did not study in detail FLJ14600, because it is clearly expressed in all xenografts and cell lines except PCEW and PC133. MINPP1 would have been an interesting candidate, because like PTEN it is able to affect phospholipid metabolism. However, the specificity of MINPP1 seems different from PTEN, and *MINPP1* is, with the exception of PCEW only affected by loss of one gene copy in part of the samples. The sulfatase gene PAPSS2 is more frequently affected by loss of one copy of the gene. In addition, PAPPS2 is not expressed or expressed at a low level in several xenografts and cell lines. Interestingly, also 2 presumed inactivating mutations were detected, although both are in samples (PC133 and LNCaP), which already show a low level of PAPSS2 expression. Arguing against a role of PAPSS2 deficiency in prostate cancer is the expression of the related gene PAPSS1 in all cell lines and xenografts (data not shown). The strongest candidate to complement *PTEN* in tumourigenesis is the candidate gene *FLJ11218*. It is inactivated

by homozygous deletion in 4 DNAs, and loss of 1 copy or low expression is found in many other samples (Figures 5 and 6). Unfortunately, so far the function of FLJ11218 is unknown. Elucidation of this function, combined with study of deletion and expression of the gene in prostate cancer tissues should be carried out to substantiate the hypothesis.

ACKNOWLEDGEMENTS

We thank Leen Blok, and Angelique Ziel-van der Made for RNA samples, and Janneke Alers for CGH advice. The study was supported by a grant from the Dutch Cancer Society KWF.

CHAPTER 4

IDENTIFICATION OF A HOMOZYGOUS DELETION AT 8P12-21 IN A HUMAN PROSTATE CANCER XENOGRAFT

Dirk C. Van Alewijk, Marcel M. Van Der Weiden, Bert J. Eussen, Lydia D. Van Den Andel-Thijssen, Conny C. Ehren-Van Eekelen, Josée J. König, Gert J. Van Steenbrugge, Winand N. Dinjens, and Jan Trapman

Genes Chromosomes Cancer 1999; 24: 119-26

ABSTRACT

One of the most frequent genetic abnormalities in prostate cancer is loss of the complete, or part of the short arm of chromosome 8, indicating the localization of one or more tumor suppressor genes on this chromosomal arm. Using allelotyping, a frequently deleted region in prostate cancer in a genetic interval of approximately 17cM between sequence tagged sites *D8S87* and *D8S133* at chromosome 8p12-21 was previously detected. A detailed physical map of this region is now available. Using known and novel polymorphic and nonpolymorphic sequence tagged sites in this interval, a search for homozygous deletions in DNAs from fourteen prostate cancer derived cell lines and xenografts was carried out. In DNA from xenograft PC133, the presence of a small homozygously deleted region of 730-1320 Kb was unambiguously established. At one site, the deletion disrupts the Werner syndrome gene. Data from allelotyping were confirmed and extended by fluorescence *in situ* hybridization analysis of PC133 chromosome spreads using centromere, YAC, and PAC chromosome 8 probes.

INTRODUCTION

Prostate cancer is the most frequently diagnosed cancer in men in Western countries, and the second leading cause of male cancer death (Parker et al. 1997). Frequent chromosomal abnormalities in sporadic prostate cancers are deletions of chromosome arms 8p, 10q, and 16q (Carter et al. 1990; Bergerheim et al. 1991; Kunimi et al. 1991), which indicate the localization of tumor suppressor genes (TSGs) at these regions. So far, only the TSG at 10q (*PTEN / MMAC1*) has been identified (Cairns et al. 1997; Vlietstra et al. 1998).

Fluorescence in situ hybridization (FISH) analysis with chromosome specific probes and comparative genomic hybridization (CGH) studies have confirmed and extended the previous allelotyping data showing frequent deletion of 8p (Isaacs 1995; Cher et al. 1996; Kallioniemi and Visakorpi 1996). Several groups have focused on more precise mapping of a deleted region at 8p, as a first step to identify candidate TSGs (Isaacs 1995; Kallioniemi and Visakorpi 1996). In many tumor DNAs, the complete or almost complete short arm of one chromosome 8 was deleted. Furthermore, comparison of the different data showed a complex pattern of overlapping and non-overlapping regions of allelic imbalance, suggesting the presence of at least two separate regions involved in prostate cancer, 8p22 and 8p12-21, respectively. Chromosome arm 8p deletions have also been found in cancers derived from colon, lung, liver, and breast. Functional evidence for a TSG on 8p has been provided by transfer of the complete or part of chromosome arm 8p to colorectal or prostate cancer cell lines, resulting in a less tumorigenic phenotype (Ichikawa et al. 1994; Gustafson et al. 1996; Nihei et al. 1996; Tanaka et al. 1996).

To fine-map the position of a TSG, the search for a homozygous deletion (HD) is an important first step. This procedure was pivotal in the identification of the *BRCA2* gene in hereditary breast cancer (Wooster et al. 1995), the *DPC4* gene in pancreatic cancer (Hahn et al. 1996), and the *PTEN/MMAC1* gene in neuroblastoma (Li et al. 1997; Steck et al. 1997). In prostate cancer, an HD of approximately 1 Mb at 8p22 has been described by Bova et al. (1993, 1996) (Bova et al. 1993; Bova et al. 1996). Additionally, Kagan et al. (1995) reported less defined large HDs in three different prostate tumors (Kagan et al. 1995). In the present study, the identification of a small HD at 8p12-21 in a human prostate cancer xenograft is presented.

MATERIALS AND METHODS

Prostate Cancer Cell Lines and Xenografts

The in vitro growing cell lines DU-145, PC-3, and LNCaP were cultured under standard conditions. PC82, PC-EW, PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346, and PC374 were grown by serial transplantation as xenografts on male nude mice (van Weerden et al. 1996).

Genomic DNA

Standard protocols were used to isolate genomic DNA from DU-145, PC-3, and LNCaP cell pellets (Sambrook et al. 1989). Genomic DNA from xenografts was isolated from five consecutive 5 μ m cryostat tissue sections according to standard procedures (overnight proteinase K incubation at 55°C, phenol extraction, and ethanol precipitation). DNA was dissolved in TE (10 mM Tris.HCl, pH 7.8; 1 mM EDTA).

Chromosome Markers

D8S278, D8S283, D8S540, D8S1733, D8S1769, AFMa224wh5 (Dib et al., 1996), D21S275 (Chumakov et al. 1992), D8S87 (Weber et al. 1991), D8S133 (Wood and Schertzer 1992), D8S136 (NIH/CEPH Collaborative Mapping Group 1992), and D8S2162 (Yu et al. 1996b) are polymorphic STSs. Other STSs are from YAC end fragments: 721D7-R (forward: aggtttgtatcccagcttttccag; reverse: caggetetatactaagetettete), 896F4-L (forward: tccatcaccctatttggcca; reverse: ttctacaggcaagaagcagg), 721D7-L (forward: ctaactcaaggcacaggcggca; reverse: gtgttggcatctgtggaaaggag), and 761A2-L (forward: gcatcgatttgtaagtcaacatgac; reverse: tgagtttcaagtcctgggtatca). MSR exon 4 (forward: gatgtgacagtggaagctatgg; reverse: ggaaaaatgtggtatatctgaagctc), POLB exon 12 (forward: ttaagccttaagtttagaacatc; reverse: gagggagaaaacgagacaag), WRN exon 1 (forward: tgctgatttggtgtctagcct; reverse: cgagaagatccagtccaacag), exon 26 (forward: cttgtgagaggcctataaactgg; reverse: ggtaaacagtgtaggagtctgc) (Yu et al. 1996a), and exon 35 (forward: tcttctgggagcctacgtgag; reverse: tgcggtttcattttcactgccctg) are gene specific markers. Detailed mapping of novel chromosome 8 STSs will be published elsewhere.

PCR Analyses

PCR amplifications, using *Taq* polymerase (Promega, Madison, WI), included 30 cycles of 1 min at 95°C, 2 min at 55°C, and 2 min at 72°C in a 50 μ I reaction volume. In radioactive multiplex PCRs, 1 μ Ci [α^{32} P]dATP (Amersham, Buckinghamshire, UK) was added to the 15 μ I reaction mix. Radio-labeled PCR products were separated on a sequence gel. Gels were dried and overnight exposed. Unlabeled PCR fragments were separated on a 2% agarose gel.

YACs and PACs

YACs y721D7, y953H12, and y888D12 were identified and selected by PCR screening and Southern blotting of the CEPH-YAC libraries (Albertsen et al. 1990; Chumakov et al. 1992). YAC DNA was isolated as described by Green and Olson (1990) (Green and Olson 1990). PACs were obtained from Genome Systems (St. Louis, MO) after screening gridded filters with specific hybridization probes derived from the markers *MSR* (p23G19), AFMa224wh5 (p243O21), *D8S1769* (p81H5), 761A2-L (p252H7), and *POLB* (p65P22), respectively. PAC DNA was isolated as described by the protocol of the manufacturers.

Fluorescence In Situ Hybridization

Chromosome preparations were made from phytohemagglutinin stimulated lymphocytes and from xenograft PC133 according to standard protocols. Slides containing chromosome spreads were washed in PBS for 5 min at room temperature. Next, they were incubated in 0.01 N HCl, containing 5 mg pepsin (Serva, Heidelberg, Germany) per 100 ml, for 10 min at 37°C, subsequently rinsed in 2xSSC, directly denatured in 70% formamide (Merck, Darmstadt, Germany)/2xSSC for 5 min at 75°C, and serially dehydrated. YAC and PAC DNAs were either biotin-14-dATP or digoxigenin-11-dUTP labeled by nick translation (nick translation kit, Gibco-BRL, Gaithersburg, MD). The biotin labeled chromosome 8 specific probe was obtained from Cambio LTD (Cambridge, UK). The chromosome-specific α -satellite probe D8Z2 (Donlon et al. 1986) was directly labeled with tetra methyl rhodamine-6-dUTP by nick translation (nick translation kit, Boehringer Mannheim, Germany). Hybridizations were done after preannealing with 1-3 µg human Cot-1 human DNA (Boehringer Mannheim, Germany) for 1 h at 37°C. All hybridizations were performed in 50%

formamide, 2xSSC, and 10% dextran sulfate in a moist chamber for 16 h at 37°C. Slides were washed twice in 50% formamide / 2xSSC for 5 min at 45°C, twice for 5 min in 2xSSC at 45°C, and once in 0.1xSSC for 5 min at 45°C. The hybridization signal was visualized by incubation with FITC conjugated avidin (Vector, Burlingame, CA), or anti-dig-rhodamine (Boehringer Mannheim, Germany). The FITC signal was amplified with two additional layers of biotinylated goat anti-avidin and FITC conjugated avidin. The slides were mounted in Vectra Shield (Vector, Burlingame, CA) containing 0.1 ng/µl DAPI (Sigma, St. Louis, MO). Analyses were performed with a Leica DM-RXA microscope equipped with a PowerGene image analyses system (PSI, Chester, UK).

RESULTS

Screening of Human Prostate Cancer Cell Line and Xenograft DNAs for Homozygous Deletions

Previously, the 8p12-21 region between markers *D8S87* and *D8S133* was found to be frequently deleted in prostate cancer (Trapman et al. 1994). Ten sequence tagged sites (STSs) that mapped between *D8S87* and *D8S133* were used to screen for HDs in DNAs from fourteen prostate cancer cell lines and xenografts (Fig. 1A, and J. Trapman, unpublished observations). For nine STSs, an amplified fragment could be detected in all DNA samples in multiplex PCRs, using chromosome 21 STS *D21S275* as an internal control (see as an example the polymorphic marker *D8S1733*; Fig. 1B). In contrast, in PC133 DNA a specific fragment was absent in the *D8S1769* amplification reaction, suggesting that part of 8p12-21, encompassing the *D8S1769* fragment, was lacking in this DNA. In some DNAs, two bands of different length were detected, showing that two allelic forms are retained in the tumor DNA. One band indicates the presence of one allele or two alleles of identical lengths. In PC133 DNA, each polymorphic marker gave rise to one fragment, with the exception of the negative *D8S1768* (Fig. 1B), and *D8S87*, which showed two bands (data not shown).



Figure 1.(A) Physical map of the region between markers *D8S133* and *D8S87* at 8p12-21, which is frequently deleted in prostate cancer. **(B)** Radioactive multiplex PCR analysis of *D8S1733* (telomeric of HD region) and *D8S1769* (in HD region) in fourteen prostate cancer cell lines. *D21S275* is used as the internal control.

Fine-mapping of the HD Region in PC133

To obtain more detailed information about the HD in PC133 DNA, additional amplification reactions were carried out with STSs flanking *D8S1769*, and between *D8S540* and *D8S278*, the two markers which were found to be present in PC133 DNA (Fig. 2). AFMa224wh5 and 761A2-L, located telomeric and centromeric to *D8S1769*, respectively (Fig. 2A), gave a PCR product in PC133. In contrast, PC133 genomic DNA was negative for STSs 721D7-R, *D8S2162*, 896F4-L, and 721D7-L, which are more closely linked to *D8S1769* (Fig. 2B). From these data it was concluded that the boundaries of the deleted region in PC133 DNA were given by AFMa224wh5 and 761A2-L. YAC 721D7 has an estimated length of 730 Kb (CEPH-Genethon database). Both end-fragments of this YAC could be mapped at 8p, showing that it was not chimeric. Both end-fragments were absent in PC133 DNA, which indicates that the HD has a length of at least 730 Kb (Fig. 2A).



Figure 2.(A) Schematic representation of the markers used in the fine mapping of the HD region in PC133. **(B)** Fine-mapping of the HD region in PC133. **(C)** Mapping of *WRN* in the HD region by amplification of specific exons.

Genes in the HD Region

The marker *D8S2162* has been mapped close to the Werner syndrome gene (*WRN*) (Yu et al. 1996a; Yu et al. 1996b). Using *WRN* exon 1, 26, and 35 specific primers, the position of *WRN* to the HD region was determined. No PCR products could be detected for *WRN* exon 26 or 35, whereas *WRN* exon 1 was found to be present (Fig. 2C). This finding shows that the HD in PC133 interrupts the *WRN* gene.

Characterization of Chromosome 8 in PC133

To extend our knowledge of the chromosomal composition of PC133, especially focussing on chromosome 8, FISH analyses of chromosome spreads with a variety of hybridization probes was done. First, the number of chromosome 8 copies in PC133 was determined by chromosome 8 painting and hybridization with the centromere probe D8Z2. This resulted in the detection of two apparently intact chromosome 8, one 8p-, one copy of a dicentric chromosome 8, one solitary chromosome 8 centromere fragment, and a chromosome 8 fragment in a marker chromosome (Fig. 3A).

In a second experiment, YAC 721D7, which is in the HD region, and YAC 888D12, which is centromeric to the HD (J. Trapman, unpublished observations) were used as hybridization probes (Fig. 2A). YACs 721D7 and 888D12 are linked by YAC 953H12, which has a length of 1320 Kb (Fig. 2A and CEPH-Genethon database). Although 721D7 provided a clear signal at normal control lymphocyte chromosomes (Fig. 3B), as expected, no signal could be detected in PC133 chromosome spreads (Fig. 3C). Hybridization with the control YAC 888D12 gave rise to a signal at the two apparently intact chromosomes 8 and two signals at the dicentric chromosome 8. Other chromosome 8 fragments were negative for 888D12 (Fig. 3C).

Subsequently, PACs 243O21 and 252H7, which are located directly telomeric and centromeric to the HD region, respectively (Fig. 2A), were used as hybridization probes. In a double FISH experiment on PC133 chromosome spreads, probes derived from both PACs gave signals on the two apparently intact chromosome 8 copies (Fig. 3D). However, other chromosome 8 fragments, including the dicentric chromosome 8, which was positive for YAC 888D12, were negative. PAC 81H5, which is completely located within the HD region (Fig. 2A) did not hybridize with any of the chromosome 8 copies (data not shown). A double FISH experiment with PAC 23G19, which contains the more telomeric *MSR* gene, and with the more centromeric *POLB* positive PAC

65P22 (Fig. 2A) provided additional information about the different chromosome 8 fragments, including the dicentric chromosome (Fig. 3E). PAC 23G19 (*MSR*; green) gave only a signal on the two apparently intact chromosomes 8, whereas 65P22 (*POLB*; red) hybridized to all chromosomes 8, including the small centromeric fragment, but excluding the chromosome 8 fragment in the marker chromosome.



Figure 3. Characterization of chromosome 8 fragments in PC133 by FISH analysis. (A) A paint of chromosome 8 (green) in combination with the D8Z2 (red) chromosome 8 centromere probe. (B) Double FISH of YAC 721D7 (green) and YAC 888D12 (red) on lymphocyte and (C) on PC133 chromosome spreads. (D) Double FISH of PAC 243O21 (green) and PAC 252H7 (red) on a PC133 metaphase. (E) Double FISH of PAC 23G19 (green) and PAC 65P22 (red) on PC133 chromosomes.

A summary of the FISH data is schematically depicted in Figure 4. Six chromosome 8 fragments were detected (see also Fig. 3A). Two copies seemed completely identical, and contained the small 8p12-21 deletion (A and B in Fig. 4). One chromosome 8 had a deletion of the major part of 8p (C in Fig. 4). The dicentric chromosome contained a duplicated 8p fragment, and most likely an intact q arm (E in Fig. 4). The 8p deletion in E (Fig. 4) was much smaller than in C (Fig. 4), but larger than in A and B (Fig. 4). Furthermore, two less well defined small fragments (D and F in Fig. 4) were found to be present.





DISCUSSION

In this study, we describe a homozygous deleted region at chromosome arm 8p in the prostate cancer xenograft PC133, derived from a skeletal metastasis. Original patient material was not available for further study. However, the HD was already found in the first passage of PC133 on nude mice, showing that it was not the result of selection during subsequent transplantation steps. The HD was detected by PCR on genomic DNA, using five different markers, and confirmed by FISH analyses. The HD is located at 8p12-21 between *D8S133* and *D8S87*, a region frequently showing allelic

loss in prostate cancer (Trapman et al. 1994). The length of the HD is less than 1320 Kb and more than 730 Kb, as judged from the lengths of YACs 953H12 and 721D7.

FISH analyses showed that the HD is present in a duplicated, but otherwise intact chromosome 8. Further, PC133 shows a complex chromosome 8 composition: 8p-, a dicentric chromosome 8, one solitary chromosome 8 centromere, and a small fragment in a marker chromosome. All these chromosome 8 copies lack parts of the p arm, which are larger than the minimal HD, as judged from the absence of p23G19 (*MSR*) and p252H7 (Fig. 4). The centromeric breakpoints in the dicentric chromosome must be close to the centromeric side of the HD, because FISH results with YAC 888D12 are positive. It is not possible to deduce with certainty from the complex chromosome 8 profile the order of genetic changes. Obviously, the small HD has arisen prior to duplication of this particular chromosome 8 copy, suggesting that the deletion is a relatively early event (Fig. 4 A and B). Furthermore, because two allelic forms of the polymorphic marker *D8S87* could be detected, some of, or all chromosomes C-F in Figure 4 must be derived from the other chromosome 8 homologe.

So far, we have not been able to identify an HD, overlapping the PC133 HD, in a different xenograft or cell line. This suggests that HDs at 8p12-21 are relatively rare events in prostate cancer and that inactivation of the putative TSG located in this region by complete or partial deletion is infrequent. Small or overlapping larger HDs have been pivotal in the identification of TSGs like *BRCA2* (Schutte et al. 1995; Wooster et al. 1995), *DPC4* (Hahn et al. 1996), and, more recently, *PTEN/MMAC1* (Li et al. 1997; Steck et al. 1997). Four other HD regions at 8p have been described in prostate cancer. A small (0.7 to 1 Mb) HD region has been identified at the *MSR* locus (8p22). Several genes have been mapped within this HD region (Macgrogan et al. 1996). However, there is, as yet, no evidence that one of these genes is a TSG. Additional HDs have been described at 8p21 and 8p22 by Kagan et al. (1995) (Kagan et al. 1995). However, these HDs are very large, and the experiments lack proper controls.

The HD region in PC133 is expected to contain approximately 30 genes. One of the 8p prostate cancer TSGs is expected to be one of these genes. So far, the only gene found to be located within the HD is the *WRN* gene. One copy is completely absent, whereas the second copy misses at least the last 10 exons. Werner syndrome is an autosomal recessive disease characterized by accelerated aging and by predisposition to certain cancers. WRN shows significant similarity to DNA helicases (Yu et al. 1996a; Yu et al. 1997). Because of its function and the localization in the HD region, *WRN* might be a candidate TSG involved in prostate cancer. The structure of *WRN* in other xenografts and tumor DNA samples is under investigation.

ACKNOWLEDGMENTS

We are indebted to Mr. Frank van de Panne (Department of Pathology) and the photolaboratory of the Department of Cell Biology for prints and photographs. Furthermore, we would like to thank Dr. Jan Van Hemel (Dept. of Clinical Genetics) for support.

CHAPTER 5

GENETIC ANALYSIS OF CHROMOSOME INTERVAL 8P12-P21 IN PROSTATE CANCER XENOGRAFTS AND CELL LINES: CHARACTERIZATION OF A HOMOZYGOUS DELETION IN XENOGRAFT PC133 DISRUPTING THE WERNER SYNDROME GENE

Dirk C. Van Alewijk, Lydia D. Van Den Andel-Thijssen, Karin G. Hermans, Conny M. Ehren-Van Eekelen, Phebe S. Wulf, Janneke C. Alers, Gert-J. Van Steenbrugge, Winand N. Dinjens and Jan Trapman

Submitted for publication

ABSTRACT

Chromosome 8p deletions have frequently been described in prostate cancer. Previously, we identified a homozygous deletion at 8p12 in prostate cancer xenograft PC133, which disrupts the Werner Syndrome gene (WRN). In the present study we extend our search for the role of chromosome 8 alterations in prostate cancer, mainly focused on 8p12-p21. Eleven xenografts and four cell lines derived from human prostate cancers were searched for chromosome 8 alterations by Comparative Genomic Hybridisation (CGH) and allelotyping. CGH showed 8p loss in ten DNAs. In eight of the samples loss of 8p included the 8p12-p21 interval. In most DNAs allelotyping matched CGH data. The homozygous deletion in PC133 was found to be 890 Kbp. Thirty-seven STSs mapping in the homozygous deletion were selected for screening of the xenografts and cell lines for overlapping homozygous deletions. However, further homozygous deletions were not detected. To identify candidate tumour suppressor genes in the homozygous deletion, two approaches were used. Search for genomic sequences encoding novel genes identified processed pseudogenes of IDH3A, SMT3H2 and FLJ20038, but no functional genes. RT-PCR showed that EST sts-N22494 was part of the 3'-UTR of a long WRN transcript. Also, exon trapping with appropriate restriction fragments from six PACs overlapping the homozygous deletion failed to detect novel genes. Finally, to test WRN as a candidate tumour suppressor gene in prostate cancer, mutation analysis of this gene in the xenografts and cell lines was performed. Except for PC133 no alteration in WRN were found. So, although complete absence of functional WRN might contribute to PC133 tumour growth, in none of the other xenografts and cell lines evidence was found for a role of complete inactivation of WRN in tumour development.

INTRODUCTION

Prostate cancer is the most frequently diagnosed cancer and the second leading cause of male cancer death in countries with a Western lifestyle (Greenlee et al. 2001). At present, an adequate therapy of metastatic prostate cancer is not available. In order

to develop novel, targeted therapeutic regimens, knowledge of the major molecular defects in prostate cancer is urgently needed.

Tumour cells are characterized by specific chromosomal alterations, mutations in tumour suppressor genes and oncogenes, and modulated gene expression patterns by chromatin modification, promoter hypermethylation and epigenetic events. Comparative Genomic Hybridization (CGH) has been utilized for genome-wide screening of chromosomal amplifications and deletions in sporadic prostate tumours (Visakorpi et al. 1995; Cher et al. 1996; Nupponen et al. 1998b; Alers et al. 2000; El Gedaily et al. 2001). Most frequent deletions have been found for chromosome arms 6q, 8p, 13q and 16q, indicating the localization of important tumour-associated genes on these chromosome arms. Other common chromosomal deletions in prostate cancer were losses of 5q, 10q, 17p and 18q. The most frequently amplified chromosome arm was 8q, followed by 1q, 3q, 7p, 7q, 17q and 20q. Frequent deletion of 8p is not only found in prostate cancer, but also in many other tumours, including colon, lung, liver and breast cancer (Emi et al. 1992; Emi et al. 1993; Fujiwara et al. 1993; Becker et al. 1996; Yaremko et al. 1996; Chughtai et al. 1999; Pineau et al. 1999; Wistuba et al. 1999).

Although allelotyping has been used in genome-wide screening studies of prostate cancer (Kunimi et al. 1991; Cunningham et al. 1996; Saric et al. 1999), it is generally applied in the accurate mapping of a deleted or amplified chromosomal region. At least two separate regions of allelic loss on chromosome 8p have been found, which is indicative for the localization of two or more distinct tumour suppressor genes on this chromosome arm (Bova et al. 1993; MacGrogan et al. 1994; Trapman et al. 1994; Macoska et al. 1995; Suzuki et al. 1995; Vocke et al. 1996; Prasad et al. 1998). Deletion of 8p22-p23 has been implicated in tumour initiation, whereas 8p12-p21 deletion may play a more significant role in tumour progression (Oba et al. 2001).

An important first step in the identification of tumour suppressor genes can be the detection of a homozygous deletion. Well known examples of tumour suppressor genes in sporadic tumours isolated by this approach are *DPC4* at 18q and *PTEN* at 10q (Hahn et al. 1996; Li et al. 1997; Steck et al. 1997). In prostate cancer tissues, homozygous deletions of chromosome arm 8p have been identified at 8p22 (Bova et al. 1993; Levy et al. 1999), 8p21 (Kagan et al. 1995), and at 8p12 (Prasad et al. 1998). Although the gene *N33* mapping in the homozygous deletion at 8p22 has been suggested as a candidate tumour suppressor gene, conclusive evidence for such a role is lacking (Macgrogan et al. 1996; Arbieva et al. 2000).

Xenografts and cell lines are powerful tools in the analysis of human cancer, because they are available in unlimited quantities, and they can be manipulated by modifications of growth conditions. Additionally, xenografts and cell lines facilitate reliable mRNA and protein expression profiling. Importantly, they lack contaminating normal cells of human origin, which simplifies the detection of homozygous deletions and small mutations in their genomic DNAs. Previously, we identified a small homozygous deletion at 8p12-p21 in the prostate cancer xenograft PC133 (Van Alewijk et al. 1999). This deletion disrupted the Werner syndrome gene (*WRN*).

In this study we extend our search for the role of chromosome region 8p12-p21 in prostate cancer, by detailed comparison of CGH and allelotyping in human prostate cancer derived xenografts, which were generated in our laboratory (Hoehn et al. 1980; Hoehn et al. 1984; Noordzij et al. 1996; Van Weerden et al. 1996), and in prostate cancer cell lines. In addition, we present the characterization of the homozygous deletion in PC133, including the gene content search, and the expression and structural analysis of *WRN* in prostate cancer xenografts and cell lines.

MATERIALS AND METHODS

Prostate Cancer derived Xenografts and Cell lines

The *in vitro* growing cell lines LNCaP, PC-3, DU-145 and TSU were grown under standard cell culture conditions. The *in vivo* xenografts PC82, PCEW, PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346 and PC374 were propagated on male nude mice as described (see Van Weerden et al. 1996).

DNA and RNA preparation

Genomic DNA from xenografts and cell lines utilized in PCR was isolated according to standard procedures, including overnight proteinase K incubation at 55°C, phenol extraction and ethanol precipitation. DNA pellets were dissolved in TE buffer (10 mM Tris.HCl, pH7.8; 1 mM EDTA). Genomic DNA for CGH was further purified by RNAse treatment. Normal mouse DNA was used as a control in allelotyping studies.

RNA was isolated by standard guanidinium isothiocyanate (cell lines) or LiCl (xenografts) protocols (Sambrook et al. 2001). Total RNAs from fetal brain and normal prostate were purchased from Clontech (Palo Alto, CA).

Comparative Genomic Hybridisation

CGH was performed essentially as described (Kallioniemi et al. 1992). In brief, tumour DNA and normal male reference DNA samples were labeled by nick translation (nick translation system, Life Technologies, Rockville, MD) with biotinylated dUTP (Roche Diagnostics, Almere, The Netherlands) and digoxigenin (Roche Diagnostics), respectively. Labeled DNA samples (200 ng each) and 15 mg Cot-1 DNA (Roche Diagnostics) were ethanol-precipitated and dissolved in 10 ml hybridization mix (50% formamide, 0.1% Tween-20, and 10% dextran sulphate in 2x SSC, pH7.0). The probe mixture was denatured and hybridized to normal male chromosome spreads (Vysis Inc., Downers Grove, IL) for 72 h at 37°C. After the slides were washed, fluorescent detection of the biotin- and digoxigenin-labelled DNA probes was accomplished by fluorescein isothiocyanate (FITC)-conjugated avidin (Vector Labs, Burlingame, CA) and anti-digoxigenin-rhodamine (Roche Diagnostics), respectively. Samples were counterstained with DAPI (4'6'-diamidino-2-phenylindole) (Sigma, St. Louis, MO) in Vectashield anti-fade solution (Vector Labs).

Analyses were performed with a Leica epifluorescent microscope, equipped with a cooled CCD camera (Photometrics Inc., Tuscon, AZ), triple-band pass beam splitter emission filters (P-1 filter set, Chroma Technology, Brattleboro, VT), and a Quips XL image analysis system (version 3.0.1; Vysis Inc.). Chromosomal regions were scored as deleted if the mean green to red ratio was below 0.85, and gained if the ratio was above 1.15.

PCR amplification, allelotyping and SSCP

PCR amplifications, using *Taq* polymerase (Promega, Madison, WI) included 30 cycles of denaturation for 1 min at 95°C, 1 min annealing at 50°C or 55°C, and extension for 1 min at 72°C. For allelotyping and PCR-SSCP, 1 μ Ci [alpha-³²P]dATP (Amersham, Buckinghamshire, UK) was added to the 15 μ I reaction mixture. For allelotyping with polymorphic microsatellite markers, radiolabeled PCR products were separated on a polyacrylamide sequence gel. For allelotyping with single nucleotide polymorphisms (SNPs), and for SSCP, radiolabeled PCR products were separated on a 6% non-

denaturing polyacrylamide gel containing 10% glycerol at low power, overnight at room temperature. Gels were dried and exposed to X-ray film for an appropriate period. Non-radioactive PCR amplifications were performed in a 50 µl reaction volume. Amplified fragments were separated on a 2% agarose gel.

Markers for allelotyping of chromosome 8

Genomic DNA was allelotyped by twenty-seven polymorphic microsatellite markers at 8p: *D8S264*, *D8S1742*, *D8S1819*, *D8S1706*, *D8S503*, *D8S549*, *D8S1731*, *D8S261*, *D8S288*, *D8S282*, *D8S133*, *D8S1733*, *D8S1752*, *D8S1739*, *D8S1771*, *D8S131*, *D8S1820*, *D8S1809*, *D8S339*, *AFMa224wh5*, *D8S1770*, *D8S1769*, *D8S1711*, *D8S1810*, *D8S283*, *D8S1722* and *D8S532*, and seven markers at 8q: *D8S260*, *D8S1797*, *D8S1705*, *D8S273*, *D8S267*, *D8S1774* and *D8S272*. Primer sequences and fragment lengths are available in the Genome Database (http://www.gdb.org). SNP markers utilized were Wiaf-3724, Wiaf-1897, Wiaf-3834, Wiaf-3345, Wiaf-283, Wiaf-72, Wiaf-3420, Wiaf-4228, and Wiaf-3809. For details see http://www.ncbi.nlm.nih.gov/SNP.

Screening for homozygous deletions

Screening for homozygous deletions directly downstream of *WRN* in genomic DNAs were done by standard PCR with thirty-seven STSs. Amplified fragments were separated on a 2% agarose gel. Details of the primer sets are available upon request.

Gene prediction of genomic DNA sequences

Human genome sequences in GenBank were searched for candidate genes by the on-line gene prediction programs GRAIL, Pipeline Analysis (Oak Ridge National Laboratory; http://compbio.ornl.gov), and Genefinder (Sanger Center; http://genomic.sanger.ac.uk/gf/gf.html). In addition, the UCSC database (http://www.cse.ucsc.edu) was searched for annotated candidate genes.

Generation of a PAC/cosmid contig of the region deleted in PC133

Gridded PAC filters (Genome Systems, St. Louis, MO), were hybridized with ³²Plabeled probes derived from the markers *AFMa224wh5*, 721D7R, WRN exon 26, 721C, WRN exon 35, 107C2-T7, 721A, *D8S1770*, *D8S1769*, *D8S124*, *D8S1711*, 978B9R, 872E4R, 896F4L, 192D20-T7, 13P8-Sp6, 721D7L, 693D8L, *D8S1810* and 761A2L. Sequences of primers utilized to generate the fragments are available upon request. DNA of individual PACs was isolated according to the manufacturers' protocol.

Exon trapping and structural analysis of novel fragments

Candidate exons were isolated from genomic DNA fragments, essentially according to the protocol of the manufacturer of the utilized exon trap kit (Life Technologies, Rockville, MD). In short, PACs 281D2, 107C2, 33H11, 22O23, 57E9 and 13P8 were digested with BamHI/BgIII and PstI. Fragments were ligated into the exon trapping vector pSPL3. The libraries were propagated in JM109. COS cells were transfected with pSPL3 libraries by electroporation. After short-time culturing, total RNA was isolated from transfected cells and trapped exons were amplified by RT-PCR. PCR fragments of appropriate lengths were cloned in pGEM-T Easy (Promega, Madison, WI) and sequenced by the dideoxychain termination method. GenBank was searched for homologous sequences by BLAST programs (http://www.ncbi.nlm.nih.gov/blast).

For candidate exons, which contained an open reading frame, and which lacked repetitive sequences, a forward primer was developed and used in combination with the site-primer (described below) in a semi-nested PCR on cDNA from fetal brain and normal prostate RNA to try to generate a downstream flanking fragment.

mRNA expression

mRNA expression was assayed by RT-PCR. First-strand cDNA synthesis was on 1 µg RNA for 1 h at 37°C using 200 U M-MLV-RT (Life Technologies) and a T₁₂-site primer (5-GCATGCGAATTCGGATCCT₁₂-3) in first-strand buffer (Life Technologies), containing 10 mM DTT, 1 mM dNTPs, and 40 U RNAsin (Promega). The first PCR reaction was performed on 1/20th of the cDNA reaction mixture, utilizing a specific forward (F) primer in combination with site-primer 5-GCATGCGAATTCGGATCC-3. Next, semi-nested PCR fragments were obtained with gene specific forward and reverse primers (F/R) on 1/500th of the first PCR product. Gene specific primer combinations for RT-PCR of *WRN* were: WRN-F 5-TCTTCTGGGAGCCTACGTGAG-3 and WRN-R 5-TGCGGTTTCATTTTCACTGCCCTG-3, respectively. Primer combinations for RT-PCR of EST sts-N22494 was; sts-N22494-F 5-ACTGAGAACAGCAGCATTGT-3 and sts-N22494-R 5-GGAAACTATTACTGGTCCAC-3. Control RT-PCRs were carried out under the same conditions, using RNA polymerase II specific primers: Pol-F (5-GCTGAGAGAGCCAAGGATAT-3) and Pol-R (5-CACCACCTCTTCCTCCTCTT-3).

Structural analysis of ESTs

Sequences overlapping with sts-N22494 were found in three other cDNA clones derived from different libraries (Al248120, fetal liver spleen; BG029683 and Bl850480, mammary carcinoma; AW779829, kidney).

EST sts-N22494 RT-PCR fragments were obtained according to the RT-PCR protocol as described above. Specific primer combinations for the first PCR amplification were: 34F 5-GGTTCTGAAGAGAGATCTGTTC-3 and 35R 5-CCCTGGTCAACTAATACCA-3 or WRN/N22494 5-CCAACTCTGTGTCACAGTATAG-3. The primer set for the nested amplification was: 34F2 5-GCAAGGAAGAAGTAGGCATC-3 and 35R2 5-CCAGAAGACCCAGAAACTAC-3.

Structural analysis of WRN

For PCR-SSCP analysis of *WRN*, overlapping 700-800 bp cDNA fragments, encompassing the complete open reading frame, were amplified. Next, smaller overlapping fragments were obtained by (semi)nested PCR and analyzed for mutations and polymorphisms by SSCP and sequencing. Primer sequences are available upon request. For sequencing, amplified fragments were purified over QIAquick spin columns (Qiagen, Hilden, Germany) and ligated into pGEM-T Easy.

RESULTS

Identification of chromosome 8 alterations in prostate cancer derived xenografts and cell lines

Chromosome 8 alterations are the most frequent genetic defects in prostate cancer. Because xenografts and cell lines are important starting materials for detailed molecular genetic studies, chromosome 8 was investigated by CGH and allelotyping in these specimens. We utilized genomic DNA from the xenografts PCEW, PC82, PC133, PC135, PC295, PC324, PC329, PC310, PC329, PC339, and PC374, and the cell lines LNCaP, PC3, DU145 and TSU.

Comparative Genomic Hybridisation

In all samples, except for xenografts PCEW and PC82, and the LNCaP cell line, chromosome 8 alterations were detected by CGH (Fig. 1A). In PC295, PC324, PC329, and PC374 loss of 8p paralleled amplification of 8q. In PC133 and PC3 apparent chromosome break points could be found in the short arm or in the long arm, respectively (Fig. 1B; see also Van Alewijk et al. 1999). In PC310, only the distal parts of 8p and 8q were deleted and amplified, respectively (Fig. 1B). Loss of 8p without 8q amplification was detected in TSU; 8q amplification without 8p deletion was present in PC346. The largest deviations from the general pattern of 8p loss and 8q gain were found for PC135 and PC339. CGH indicated in both xenografts additionally to gain of 8q, amplification of parts of 8p (Fig. 1C).



Figure 1. (A) Chromosome 8 alterations identified by comparative genomic hybridisation in fifteen prostate cancer xenografts and cell lines (CGH: mean 0.85-1.15; n=12-17). A bar to the right of an ideogram indicates gain; a bar to the left indicates loss. (B) Samples showing gain of 8q and proximal 8p, and loss of distal 8p (PC133), gain of 8q and loss of 8p (PC310), and gain of distal 8q and loss of proximal 8q and 8p (PC3). (C) Samples showing gain of whole chromosome 8 (PC339), and gain of distal 8p and distal 8q (PC135).

Allelotyping

To obtain independent complementary, more detailed information on chromosome 8 alterations, the same genomic DNAs were allelotyped by thirty-five highly polymorphic microsatellite markers. The highest marker density was chosen for 8p12-p21, where in previous studies loss of heterozygosity in DNA from prostate tumour samples, and a homozygous deletion in xenograft PC133 have been described by us (Trapman et al. 1994; Van Alewijk et al. 1999). Allelotyping results are summarized in Fig. 2A; some informative examples are depicted in Fig. 2B. In PC346, PC374, LNCaP and DU145 30-65% instability of the microsatellite markers was detected, which limited the accuracy of the analyses in these samples [see *D8S1731* in PC346 and LNCaP as an example (Fig. 2B)].

Almost all xenografts and cell lines showed two allelic forms of at least one polymorphic marker on 8q. For many 8q and some 8p markers a clear imbalance of the two alleles was found, which is indicative of gain [see *D8S1731* (PC135), *D8S1722* (PC133), *D8S273* (PC135) and *D8S272* (PC339) as examples in Fig. 2B]. In PC3 and TSU all seven 8q and twenty-eight 8p markers demonstrated one allelic form, indicating the loss of one copy of the whole chromosome. In PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346 and P374 large series of consecutive markers on 8p showing apparent homozygosity were detected, strongly suggesting loss of one copy of the corresponding chromosomal regions. The maximum number of consecutive markers showing one band in PCEW was five: from *D8S258* till *D8S1752*, in a 5 Mbp interval. This might represent a small region of chromosomal loss. In the region between *AFMa224wh5* and *D8S283*, which is completely deleted in PC133, no clear evidence for allelic loss was found in PCEW, PC82, PC310, PC346, LNCaP and DU145.

p

q



Chromosome 8

В



Figure 2. (A) Chromosome 8 alterations as identified by allelotyping of fifteen prostate cancer xenografts and cell lines. Two allelic bands are given as 2; 1 indicates one band, which might represent one allele or two alleles of identical length. Occasionally, one of two allelic bands was stronger, indicative of gain (data not shown). The homozygous deletion in PC133 (REF) is represented by "0". MSI: microsatellite instability. The previous described LOH region (Trapman et al. 1994) between *D8S133* and *D8S87* at 8p12-p21 is

represented by a vertical bar. (B) Selection of samples demonstrating one and two allelic forms, amplification (2+), and MSI.

To obtain more information on loss of chromosome 8 in the four samples with frequent MSI (PC346, PC374, LNCaP and DU145), they were typed by nine single

nucleotide polymorphisms (SNPs) scattered along 8p22-pter (see Materials and Methods). In both LNCaP and DU145, several SNP markers revealed two alleles across 8p22-pter, which is indicative of absence of 8p loss in these cell lines. PC346 showed two allelic forms of a clusterin (*CLU*) SNP, which mapped between *D8S1771* and *D8S131*, but one allelic form for all more distal SNPs, indicating loss of the distal part of 8p (data not shown). In PC374 none of the 8p SNPs showed two alleles.

Comparison of Comparative Genomic Hybridization and allelotyping of prostate cancer xenografts

In general, the CGH and allelotyping data matched well, although there were some discrepancies. Most remarkable are TSU, for which allelotyping of the whole chromosome showed the presence of single alleles, without indication of loss of 8q, PC3 with 8q amplification by CGH analysis and loss by allelotyping, and PC339 with 8p amplification by CGH and 8p deletion by allelotyping.

The homozygous-deleted region in xenograft PC133

Length of the homozygous-deleted region in PC133, and search for novel, overlapping homozygous deletions

As shown previously (Van Alewijk et al. 1999) and in this study (Fig. 2A and 3), xenograft PC133 contained a homozygous deletion at 8p12. The deletion disrupted at its telomeric border *WRN*. As deduced from YAC lengths, the length of the homozygous-deleted region was previously estimated to be between 0.7 and 1.3 Mbp (Fig. 3C; Van Alewijk et al. 1999).

The borders of the deleted region were more accurately determined with STS markers. The telomeric border mapped in *WRN* intron 8 (Fig. 3B). Centromeric, *D8S1810* was found to be located in the deleted region, whereas 13P8-T7 was just proximal to the deletion (Fig. 3B). Comparison with the human genome sequence showed that the homozygous deletion spanned approximately 890 Kbp (Fig. 3E). Within this region, thirty-seven STS markers, together with markers that mapped at the borders of the homozygous-deleted region, were applied in screening for overlapping homozygous deletions in all available prostate cancer xenografts and cell lines. However, in none of the DNAs a novel homozygous deletion could be detected.




Gene search in the homozygous deletion by sequence analysis

Annotation of the human genome sequence by different gene prediction programs (UCSC, Ensembl, NCBI) showed *WRN* as the only known gene in the homozygous deletion. Gene finding programs like GRAIL, Genefinder, and Pipeline Analysis also failed to detect *bona vide* genes, except *WRN*. Sequence homology searches identified processed pseudogenes of *IDH3A*, *SMT3H2* and *FLJ20038* in the homozygous deletion (Fig. 3F).

Many ESTs mapped in the homozygous deletion. However, critical study of EST sequences and their flanking genomic sequences, absence of a polyadenylation signal, presence of a flanking genomic A-track, presence of a repeat sequence, absence of a long open reading frame, combined with low expression made it very unlikely that they were parts of genes. The exception was sts-N22494, which is discussed below.

Gene search in the homozygous-deleted region by exon trapping

We isolated the 890 Kbp genomic segment in a PAC/Cosmid contig. Appropriate genomic DNA fragments from six overlapping PACs (see Fig. 3D) that covered the homozygous deletion were screened for novel genes by exon trapping. Approximately one hundred different genomic fragments were identified. From PAC P281D2 *WRN* exons 14 to 34 were isolated. None of the other trapped fragments was identical to a known gene or EST sequence. Twenty fragments had an open reading frame and did not contain repetitive sequences. To study whether these fragments were present in mRNA, semi-nested RT-PCRs were done on RNA from normal prostate and from fetal brain. However, none of the PCRs was positive, accordingly none of the fragments were part of a *bona fide* mRNA.

Characterization of sts-N22494

Sts-N22494 mapped just downstream of the last *WRN* exon (Fig. 3F). Overlapping sequences were found in 4 other ESTs (see Materials and Methods). All contained a polyadenylation signal, but lacked a long open reading frame (data not shown). The genomic sequence was identical to the cDNA sequence, but lacked a polyA-stretch downstream of the polyadenylation signal. Sts-N22494 expression was analyzed by RT-PCR (Fig. 4A). Sts-N22494 was easily detected in all RNA samples from prostate cancer xenografts and cell lines, with the exception of PC133. The genomic sequence revealed that sts-N22494 was located approximately 2.2 Kbp downstream of the *WRN* gene, in the same orientation. Because no convincing open reading frame could be found in the genomic sequence between *WRN* and sts-N22494,

the possibility that sts-N22494 was part of a longer *WRN* transcript was considered (Fig. 4B and 4C). First, a RT-PCR reaction from *WRN* exon 34 across the initially described *WRN* exon 35 was done (34F-WRN/N22494R, Fig. 4B). Subsequently, the product was visualized by a nested PCR from *WRN* exon 34 to exon 35 (34F2-35R2 in Fig. 4B; 335 bp fragment in 4C); the 34F-35R RT-PCR followed by 34F2-35R2 nested PCR was carried out as a control (*WRN* in Fig. 4C). The positive result provided strong evidence that indeed sts-N22494 was derived from a longer *WRN* transcript.

Expression and structural analysis of WRN

Because WRN turned out to be the only gene present in the homozygous deletion in PC133, WRN expression and structure were investigated in all other prostate cancer xenografts and cell lines. In all samples, except PC133, WRN mRNA was detectable by RT-PCR (Fig. 4). The entire open reading frame of WRN mRNA was analyzed for mutations by RT-PCR-SSCP. All fragments showing an altered PCR-SSCP pattern were sequenced. The altered sequences turned out to be previously described polymorphisms: TGT(Cvs)171TGC(Cvs); ATG(Met)387ATA(Ile); CTT (Leu)787CT(Leu); TTT(Phe)1074TTG(Leu); AGC(Ser)1361AGT(Ser) and TGT(Cys)1367CGT(Arg) (Bernardino et al. 1997; Meisslitzer et al. 1997; Ye et al. 1997). So, no evidence could be found for substantial downregulation of WRN expression or inactivating mutations in the open reading frame.



Figure 4. (A) RT-PCR analysis of STS-N229494 and *WRN* mRNA expression in prostate cancer cell lines and xenografts and cell lines. RNA pol II was utilized as a control. RT: reverse transcriptase. (B) Schematic representation of RT-PCR and nested PCR to detect *WRN* and STS-N22494 transcripts. (C) Amplified fragments obtained by nested PCR (34F2/35R2) of *WRN* and STS-N22494 transcripts, showing that sts-N22494 represents a longer *WRN* transcript. FB: fetal brain; NP: normal prostate; RT: reverse transcriptase.

DISCUSSION

In this study, we describe the genetic characterization of chromosome 8 in fifteen human prostate cancer derived xenografts and cell lines, mainly focused on the homozygous deletion at 8p12 in xenograft PC133 (Van Alewijk et al. 1999). Xenografts and cell lines are unique starting materials for the identification of genes associated with prostate cancer and the functional characterization of such genes.

Loss of 8p and gain of 8q are among the most frequently observed chromosomal alterations in clinical prostate cancer samples (Bova et al. 1993; Trapman et al. 1994; Visakorpi et al. 1995; Cher et al. 1996; Vocke et al. 1996; Nupponen et al. 1998b; Prasad et al. 1998; Alers et al. 2000). Loss of 8p indicates the localization of one or more tumour suppressor genes and amplification of 8q suggests the presence of oncogenes. By CGH, ten out of fifteen xenografts and cell lines showed loss of 8p, encompassing the region 8p12-pter in eight samples. Furthermore, in eleven samples gain of 8q was found. So, the xenografts and cell lines reflect the chromosome 8 alterations found in patient materials.

Concomitant 8p loss and 8q gain has been attributed to the presence of isochromosome 8q (Macoska et al. 2000). In at least four xenografts this mechanism might have occurred (PC295, PC324, PC329, and PC374). However, in four xenografts and cell lines, different chromosome alterations must have taken place. CGH data indicate loss of 8p without gain of 8q in TSU, amplification of 8q13-qter in PC3, loss of distal 8p and gain of distal 8q in PC310, and a dicentric chromosome 8 in PC133 (Van Alewijk et al. 1999).

In general, CGH data matched those obtained by allelotyping. Major discrepancies were found for PC339, PC3, TSU and DU145. In PC339, CGH indicated gain of the entire chromosome 8. However, all polymorphic markers on 8p showed one allelic form, which is strong evidence for 8p loss. Probably, one copy of chromosome 8 lost its p-arm, followed by amplification of the second, intact, copy. Similarly, CGH indicated gain of 8q in PC3, but all seven 8q markers showed one allelic band. Possibly, one copy of chromosome 8 was lost and the q-arm of the second copy was amplified. In DU145, CGH showed 8p loss and by allelotyping two alleles for the most distal marker *D8S264* and four 8p SNPs were found. The most likely explanation for the discrepancy

112 | Chapter 5

is loss of 8p in the partially triploid cell line DU145. In TSU, CGH indicated loss of 8p, but allelotyping suggested loss of one entire copy of chromosome 8. Interestingly, a similar observation was made by allelotyping of chromosomes 6 and 10 (Verhagen et al. 2002; Hermans, unpublished), suggesting a defect in chromosome segregation in this cell line. Our CGH data of the cell lines PC3, LNCaP, TSU and DU-145 were in accordance with previous CGH and SKY studies (Bernardino et al. 1997; Nupponen et al. 1998a; Pan et al. 1999). The complementary allelotyping showed that genomic alterations in these cell lines might be more complex than indicated by CGH. Small regions of amplification or loss might be most accurately identified by array CGH (Pinkel et al. 1998). This might lead to detection of high-level amplifications on 8q and small regions of loss in those xenografts and cell lines that seem to contain a normal 8p content. Array CGH might also be applied to confirm the proposed small region of loss in PCEW (Fig. 2) and to identify novel small homozygous deletions on 8p.

In PC346, PC374, LNCaP and DU145 we observed a high frequency of MSI, implicating defects of the mismatch repair machinery. In DU145 a splice acceptor site mutation has been found in *MLH1* (Boyer et al. 1995; Chen et al. 2001). LNCaP contained a homozygous deletion of *MSH2* (Leach et al. 2000; Chen et al. 2001); Trapman, unpublished). The defects in mismatch repair genes in PC346 and PC374 remain to be established. MSI was also detected in the clinical specimen from which xenograft PC346 was derived (data not shown). The frequency of MSI or aberrant expression of mismatch repair genes in clinical prostate cancer specimens, as reported so far, is very variable, ranging from 0% to 65% (Egawa et al. 1995; Uchida et al. 1995; Cunningham et al. 1996; Rohrbach et al. 1999). Our observations warrant renewed search for the role of mismatch repair genes in prostate cancer.

Although we carried out an extensive search, we were unable to find in any of the xenografts and cell lines a homozygous deletion overlapping the one previously detected in PC133 (Van Alewijk et al. 1999). The deletion in PC133 has a length of approximately 890 Kbp and disrupts *WRN*. The complete sequence of this genomic region is known. Except for *WRN*, no genes mapped in this region, utilizing a variety of experimental and gene prediction approaches. Because in PC133 the homozygous deletion was selected during tumour growth, it is very likely that *WRN* plays a role in the development of this tumour. *WRN* encodes a RecQ helicase that also contains DNA exonuclease activity. It is believed to function in maintaining the integrity of the genome (Moser et al. 2000a) reviewed by (Hickson 2003). Werner Syndrome patients are characterized by premature

aging and by predisposition for tumour development. In mouse models, *Wm* deficiency accelerates tumour growth and broadens the tumour profile of *p53* null mice (Lombard et al. 2000; Lebel et al. 2001). Although prostate cancer has never been associated with Werner Syndrome these properties make it a candidate tumour suppressor gene in PC133. *WRN* seems not to be involved in the other xenografts and cell lines, because no deletions or point mutations have been found. However, it cannot be excluded that diminished *WRN* expression, due to haplo-insufficiency contributes to tumour formation. Targeted inactivation of *Wm* in appropriate mouse cancer models will be instrumental to investigate this hypothesis.

ACKNOWLEDGEMENTS

We thank Leen Blok and Angelique Ziel-van der Made for RNA preparations. This work was supported by a grant from the Dutch Cancer Society, Koningin Wilhelmina Fonds (KWF).

CHAPTER 6

PRELIMINARY DATA ON SCREENING FOR HOMOZYGOUS DELETIONS, AND EXPRESSION AND MUTATION ANALYSIS OF CANDIDATE TUMOUR SUPPRESSOR GENES ON CHROMOSOME 8P12-P21 IN PROSTATE CANCER

Dirk C. Van Alewijk, Lydia D. Van Den Andel-Thijssen, Conny C. Ehren-Van Eekelen, Phebe S. Wulf, Wytske van Weerden, Winand N. Dinjens and Jan Trapman

INTRODUCTION

One of the most frequent genomic alterations in prostate cancer is partial or complete loss of chromosome arm 8p, which indicates the localization of one or more tumour suppressor genes on this chromosomal arm (Kunimi et al. 1991; Visakorpi et al. 1995; Cher et al. 1996; Cunningham et al. 1996; Nupponen et al. 1998b; Saric et al. 1999; Alers et al. 2000). An important first step in the identification of tumour suppressor genes is the identification of a homozygous deletion. In prostate cancer, several homozygous deletions have been identified at chromosome 8p (Bova et al. 1993; Kagan et al. 1995; Prasad et al. 1998; Levy et al. 1999; Van Alewijk et al. 1999). However, this has so far not resulted in the isolation of a *bona fide* tumour suppressor aene. The homozygous deletion that we identified disrupted the Werner syndrome gene (WRN) in xenograft PC133 (Van Alewijk et al. 1999). However, structural alterations or absence of WRN expression could not be found in any of 11 prostate cancer xenografts, generated in Rotterdam (Noordzij et al. 1996; Van Weerden et al. 1996: and references therein), or the cell lines PC-3, LNCaP, TSU and DU-145. It was concluded that complete inactivation of WRN had no significant role in prostate cancer (Chapter 5; Van Alewijk et al. submitted).

We continued to search for a tumour suppressor gene on 8p12-p21 by two parallel experimental approaches. First, screening for novel homozygous deletions, and secondly, structural and expression analysis of candidate genes. As a start we investigated *LZTS1/FEZ*, *NKX3-1*, *EPHX2* and *CLU* on 8p12-p21, *POLB* on 8p11 and *PRLTS* on 8p22. Except for *EPHX2* and *CLU*, all genes were previously implicated in prostate cancer (Dobashi et al. 1994; Komiya et al. 1997; Ishii et al. 1999; Kim et al. 2002a). *EPHX2* and *CLU* were selected because of their physiological function.

In both the screening for homozygous deletions, and the expression and mutation analysis we used the xenografts and cell lines as described above. Xenografts and cell lines are powerful tools for the genetic analysis of cancer because they lack contaminating normal cells of human origin. Like in prostate cancer tissues, deletion of 8p in xenografts and cell lines is a frequent chromosomal defect (Chapter 5; Van Alewijk et al., submitted). If the expression and mutation analysis of the xenografts and cell lines showed a possible tumour related alteration, we continued by screening for mutations in DNAs from 43 locally progressive prostate tumour samples.

MATERIALS AND METHODS

Prostate Cancer derived Xenografts and Cell lines

The *in vitro* growing cell lines LNCaP, PC-3, DU-145 and TSU were grown under standard cell culture conditions. The *in vivo* xenografts PC82, PCEW, PC133, PC135, PC295, PC310, PC324, PC329, PC339, PC346 and PC374 were propagated on male nude mice as described (Van Weerden et al. 1996). Prostate tumour tissues were from patients with a confirmed clinical history of locally progressive prostate cancer. Tissue samples were obtained by transurethral resection (TUR).

DNA and RNA preparation

Genomic DNA from xenografts and cell lines was isolated according to standard procedures, including overnight proteinase K incubation at 55°C, phenol extraction and ethanol precipitation. DNA pellets were dissolved in TE buffer (10 mM Tris.HCl, pH7.8; 1 mM EDTA). Tumour DNA was isolated from five consecutive 5 µm cryostate tissue sections according to the standard procedure. In the vast majority of cases, control DNA was from white blood cells. For a minority of tumours, control DNA was from non tumour regions of prostate tissue.

RNA was isolated by standard guanidinium isothiocyanate (cell lines) or LiCl (xenografts) protocols (Sambrook et al. 2001). Total RNA from foetal brain and normal prostate were purchased from Clontech (Palo Alto, CA).

PCR amplification and SSCP

PCR amplifications, using *Taq* polymerase (Promega, Madison, WI) included 30 cycles of denaturation for 1 min at 95°C, 1 min annealing at 50°C or 55°C, and extension for 1 min at 72°C. For PCR-SSCP, 1 μ Ci [alpha-³²P]dATP (Amersham, Buckinghamshire, UK) was added to the 15 μ l reaction mixture. For SSCP, radio-labelled PCR products were separated on a 6% non-denaturing polyacrylamide gel containing 10% glycerol at low power, overnight at room temperature. Gels were dried and exposed to X-ray film for an appropriate period. Standard PCR amplifications were performed in a 50 μ l reaction volume. Amplified fragments were separated on a 2% agarose gel.

Screening for homozygous deletions

The 15 xenograft and cell line DNAs were analysed by PCR for homozygous deletions in a 17 Mbp region on 8p12-p21, utilising approximately 300 specific markers (see Appendix A6 - supplementary Table I). Amplified fragments were separated on a 2% agarose gel. Primer sets are available upon request.

mRNA expression

mRNA expression was assayed by RT-PCR. First-strand cDNA synthesis was on 1 µg RNA for 1 hr at 37°C using 200 U M-MLV-RT (Life Technologies, Rockville, MD) and a T₁₂-site primer (5-GCATGCGAATTCGGATCCT₁₂-3) in first-strand buffer (Life Technologies), containing 10 mM DTT, 1 mM dNTPs, and 40 U RNAsin (Promega). The first PCR reaction was performed on 5% of the cDNA reaction mixture, utilizing a specific forward (F) primer in combination with site-primer 5-GCATGCGAATTCGGATCC-3. Next, semi-nested PCR fragments were obtained with gene specific forward and reverse primers (F/R) on 1/500 of the first PCR product. Gene specific primer combinations for RT-PCR of *PDGFRL*, *LZTS1*, *NKX3-1*, *EPHX2*, *CLU*, *POLB*, and *RNA pol-II* are available upon request.

Structural analysis

Amplified fragments were purified over QIAquick spin columns (Qiagen, Hilden, Germany), ligated into pGEM-T Easy, and sequenced by the dideoxy chain termination method.

RESULTS AND DISCUSSION

Here, the preliminary results of the search for homozygous deletions at chromosome 8p12-p21, and the expression and mutation analysis of the genes *PDGFRL*, *LZTS1*, *NKX3-1*, *EPHX2*, *CLU*, and *POLB* in prostate cancer are described.

High-density screening for homozygous deletions at 8p12-p21

To screen the xenografts and cell lines for a homozygous deletion in a 17 Mbp region on 8p12-p21, approximately 300 markers mapping between LPL and BC028701 were selected (see Appendix A6 - supplementary Table I). Except for the previously described homozygous deletion in PC133 (Van Alewijk et al. 1999), no homozygous deletion could be identified, despite the high marker density.

Expression and mutation analysis of candidate tumour suppressor genes

In previous studies, 4 genes have been claimed to be involved in prostate cancer, PDGFRL (8p22), LZTS1/FEZ (8p21), NKX3-1 (8p21) and POLB (8p11). PDGFRL was suggested as a candidate tumour suppressor gene because of 1 missense mutation in a series of 69 prostate tumour samples (Fujiwara et al. 1995). In addition, missense and frameshift mutations were found in 2 out of 48 hepatocellular cancer and 1 out of 28 colorectal cancers (Fujiwara et al. 1995). LZTS1 was suggested to be a tumour suppressor gene, because expression was undetectable in more than 60% of epithelial tumours, whereas it is ubiquitously expressed in normal tissues (Ishii et al. 1999). In addition, in a series of 194 epithelial tumours, Ishii et al (1999) found 2 missense mutations in primary oesophageal cancers and 1 nonsense mutation in the prostate cancer cell line PC-3. Further, LZTS1 expression levels were significantly reduced in gastric carcinomas (Vecchione et al. 2001). Germ line sequence variants of the LZTS1 gene were associated with an increased prostate cancer risk (Hawkins et al. 2002). Because of its function and prostate specific expression the homeobox gene NKX3-1 was associated with prostate cancer (He et al. 1997; Sciavolino et al. 1997; Bhatia-Gaur et al. 1999). POLB showed 1 frameshift and 1 missense mutation in a series of 12 prostate cancer samples (Dobashi et al. 1994). In addition, 1 missense and 1 frameshift mutation were found in 2 out of 6 colorectal cancers (Wang et al. 1992).

In addition to candidate tumour suppressor genes from literature searches, *EPHX2* (8p21) was studied because of its function as a cytosolic member of the family of the epoxide hydrolases (Larsson et al. 1995; Sandberg and Meijer 1996). In various organisms epoxide hydrolases have an important protective function because they are able to convert potentially harmful epoxide-containing compounds into diols, which are less reactive and easier to secrete (Meijer and DePierre 1988). Interestingly, a significant association was found between polymorphisms and activity of microsomal epoxide hydrolase 1 (*EPHX1*), and cancers of the respiratory tract (Benhamou et al. 1998; Jourenkova-Mironova et al. 2000). The authors postulated that *EPHX1* was an important genetic determinant for smoking-induced lung cancer.

CLU (8p21) has also been investigated as a candidate tumour suppressor gene because of its function. Clusterin has been implicated in several diverse physiological

processes, one of these is apoptosis of normal and prostate cancer cells (Trougakos and Gonos 2002; Miyake et al. 2003).

For neither *PDGFRL*, nor *CLU* or *POLB* any alteration in expression, as analysed by RT-PCR, or structure, as determined by PCR-SSCP and sequencing, was detected in the 15 xenografts and cell lines, indicating an insignificant role in prostate cancer (data not shown).

As shown in Table I, expression of *LZTS1* mRNA was detectable in all RNA samples, except for PC133. *NKX3-1* mRNA was detectable in all RNA samples from prostate. As expected *NKX3-1* mRNA expression was undetectable in foetal brain (He et al. 1997). Except for TSU, the expression of *EPHX2* mRNA was detectable in all RNA samples. In summary, no significantly altered expression pattern in prostate cancer was found for these 3 genes.



	PC82	PCEW	PC133	PC135	PC295	PC310	PC324	PC329	PC339	PC346	PC374	PC3	LNCaP	TSU	Du145	Normal prostate	Fetal Brain
LZTS1	++	++	-	++	++	++	++	++	++	++	++	++	++	++	++	++	++
NKX 3.1	++	++	+	+	++	++	+	++	++	++	++	+	++	++	++	++	-
EPHX2	+++	+++	+	+++	+++	+++	+++	+++	+++	+++	+++	+++	+++	-	+++	+++	+++
RNA pol II	++	++	++	++	++	++	++	++	++	++	++	++	++	++	++	++	++

LZTS1 and *EPHX2* were initially analysed for mutations by PCR-SSCP on genomic DNA of the xenografts and cell lines. Subsequently, samples showing aberrant SSCP bands were sequenced. Because paired control DNA was not available, 50 independent normal DNAs were used as controls. *NKX3-1* sequence was analysed by PCR-SSCP of overlapping amplified cDNA fragments, encompassing the complete open reading frame. *NKX3.1* single nucleotide polymorphisms present in databases were used as controls. The experimental results are listed in Table II. Frequently found polymorphisms are not included.

LZTS1

As shown in Table II, we identified in *LZTS1* 1 silent polymorphism and mutations or rare polymorphisms in DU-145 (AAC>AGC, codon 223) and in LNCaP (CAG>CGG, codon 556), respectively. In both DU-145 and LNCaP the wild type allele was retained. We could not find the previously by Ishii et al. (1999) reported nonsense mutation in *LZTS1* at codon 501 in PC3. Possibly, the PC3 cell line analysed by Ishii and co-workers has gained the mutation during culturing. Although our results implicated a minor role for *LZTS1* in PrCa, support for *LZTS1* as a tumour suppressor gene was published by Cabeza-Arvelaiz et al. (2001). These authors demonstrated that transfer of YAC and BAC clones containing the *LZTS1* gene into rat prostate AT6.2 cells reduced their colony-forming efficiency. Subsequent experiments showed that over-expression of *LZTS1* cDNA inhibited colony-formation in soft agar of AT6.2, HEK-293 and LNCaP cells (Cabeza-Arvelaiz et al. 2001).

NKX3-1

Two different *NKX3-1* sequences were detected. Previously, these were identified as single nucleotide polymorphisms (Voeller et al. 1997). Although no structural indications were found, evidence is growing for a role of *NKX3-1* in prostate cancer. The mechanism through which NKX3-1 functions in prostate cancer appears not to be by bi-allelic inactivation but by haplo-insufficiency (Bhatia-Gaur et al. 1999; Abdulkadir et al. 2002; Kim et al. 2002a).

EPHX2

Although minor differences were found in *EPHX2* mRNA expression, structural analysis revealed interesting data. As shown in Table II, we detected in the xenografts PC133 and PC324 and in the cell line PC3 unique sequences, which were not present in the 50 control DNAs. The wild type allele was lost in all 3 DNAs. For PC324, the corresponding tumour tissue (T1.7) and paired control DNA were available. These DNAs also contained the unique TCT codon 344.

Subsequent study of *EPHX2* in DNAs from 43 prostate tumour tissues showed 4 rare polymorphisms, leading to amino acid substitutions. Remarkably, in all cases the wild type allele was lost (Table II). Some of these amino acid residues were in conserved regions, suggesting that they could affect EPHX2 function. In a Swedish (n=25) and a Japanese (n=48) study (Sandberg et al. 2000; Saito et al. 2001) the

polymorphisms that we described were not found in normal control DNAs. Although absence of these rare polymorphisms might be explained by a difference in ethnic background and by the low number of samples, they support a possible role of *EPHX2* as a susceptibility gene in prostate cancer. This hypothesis however needs to be substantiated by screening of a significant larger cohort of preferably Dutch individuals, and by functional studies of mutated EPHX2.

ACKNOWLEDGEMENTS

We thank Leen Blok and Angelique Ziel-van der Made for RNA preparations.

| Chapter 6

CHAPTER 7

GENERAL DISCUSSION









The work described in this thesis covers the search for tumour suppressor genes located in chromosomal regions at 8p and 10q that are frequently deleted in prostate cancer (PrCa). The study was originally guided by the paradigm that cancer is a genetic disease involving dominant mono-allelic activation of proto-oncogenes and recessive bi-allelic inactivation of tumour suppressor genes. Many tumour suppressor genes and oncogenes have indeed been identified that follow this classical principle. The number of tumour suppressor genes as well as the number of tumour types in which they are shown to be involved is however still relatively limited. Evidence is growing that the general genetic and molecular mechanisms underlying tumourigenesis is more complex and that the classical principles require adjustment. This seems also true for PrCa.

PTEN: a classical PrCa tumour suppressor gene at chromosome 10q

Deletions of chromosome 10q are among the most frequent chromosomal alterations observed in advanced PrCa (50-70%). In 1997, Li and co-authors identified bi-allelic inactivating mutations in the *PTEN* gene in multiple tumour types, including PrCa. *PTEN* is located at chromosome 10q and is now known as the classical PrCa tumour suppressor gene. We and others found frequent bi-allelic *PTEN* inactivation in PrCa. We identified *PTEN* deletions or mutations in 7 out of 11 human PrCa xenografts (Vlietstra et al. 1998). Overall, *PTEN* mutations are observed in a small proportion of primary PrCa (5-15%) and in a majority of metastatic PrCa (30-60%). The frequencies of bi-allelic inactivations of the *PTEN* gene in PrCa are however consistently lower than the rates of 10q loss of heterozygosity in these tumours.

Is there a classical PrCa tumour suppressor gene at chromosome 8p?

Loss of chromosome 8p is the most common chromosomal alteration in PrCa, occurring already in PIN (20-60%) and increasing during PrCa tumour progression (45-85%), indicative for the presence of more than one tumour suppressor gene on this chromosomal arm. The location of a tumour suppressor gene involved in PrCa at 8p was substantiated by chromosomal transfer experiments. Many independent studies, including several from our laboratory, have revealed a complicated picture of the allelic losses at 8p. There are multiple regions of frequent loss of heterozygosity at this chromosomal arm, including 8p22-p23, 8p12-p21, and 8p11-p12. The losses of 8p22-

p23 and 8p21.3 correlate with tumour grade and the losses of 8p21.1-p21.2 correlate with tumour progression, suggesting that chromosome 8p may contain multiple tumour suppressor genes involved in PrCa. Obviously, the presence of multiple tumour suppressor genes at 8p will complicate the identification of any of these genes. Although we and others have taken this possibility into account and have adjusted our gene searches accordingly, not a single classical tumour suppressor genes at 8p has been found as yet. This raises the question if there exists such genes at 8p. If not, why then would 8p losses correlate with tumourigenesis?

Haplo-insufficiency as an additional mechanism in tumourigenesis

Like for chromosome 8p, searches for PrCa tumour suppressor genes were also unsuccessful for other frequenty deleted chromosome arms such as 5q, 13q and 16q. Similar difficulties were encountered for the characteristic losses in other tumour types. In 2000, Cook and McCaw proposed an important extension of the paradigm in cancer genetics by suggesting that mono-allelic inactivation of a tumour suppressor gene may eventuate in a selective growth advantage of the mutant cell. Such haploinsufficiency would result in a lower dosage of the encoded protein, presumably enough to effectuate clonal outgrowth. Ever since, a growing number of reports has provided evidence for a role of haplo-insufficiency in tumourigenesis (reviewed by Quon and Berns 2001; Fodde and Smits 2002). Importantly, haplo-insufficiency would very well explain the inability to find a classical PrCa tumour suppressor gene at 8p, as mono-allelic inactivation through loss of heterozygosity would suffice (Kim et al. 2002a).

Is haplo-insufficiency a common mechanism in PrCa?

Although it has been shown that *PTEN* is a classical tumour suppressor gene in PrCa and in many other human malignancies, the rate of bi-allelic inactivation seems significantly lower than the rate of loss of heterozygosity at the *PTEN* locus (Cairns et al. 1997; Teng et al. 1997; Feilotter et al. 1998; Whang et al. 1998). Apart from a second tumour suppressor gene located nearby *PTEN*, this could point to a role of *PTEN* haplo-insufficiency in tumour growth. In mice, concomitant inactivation of a single *Pten* allele with one or both of Cdkn1b, Cdkn2a or Nkx3-1 alleles was found to accelerate spontaneous neoplastic transformation (Di Cristofano et al. 2001; Kim et al. 2002b; You et al. 2002). Crosses of TRAMP mice, having SV40 early genes driven by

the prostate-specific probasin promoter, with heterozygous *Pten* (+/-) mice also revealed that mono-allelic inactivation of the *Pten* gene promoted the progression of PrCa (Kwabi-Addo et al. 2001). Together these findings provide evidence that *Pten* haplo-insufficiency indeed is involved in PrCa in mice. Similarly, loss of heterozygosity at the *PTEN* locus in human PrCa may promote tumourigenesis by haplo-insufficiency of the gene, thus explaining the discordance between the *PTEN* mutation rate and the rate of loss of heterozygosity at 10q.

Whereas haplo-insufficiency may perhaps be of minor importance for the PTEN gene, it could be a particularly important mechanism for the multiple PrCa tumour suppressor genes thought to be located at 8p. It may be that the additive effect of monoallelic inactivation of several tumour suppressor genes at the same time by loss of a large part of the chromosomal arm has a stronger effect on tumour growth than the biallelic inactivation of just a single tumour suppressor gene at this arm. In my opinion, at least four candidate tumour suppressor genes at 8p deserve attention, in particular because of their potential to promote PrCa by haplo-insufficiency. In mouse models, unambiguous evidence has been provided that the NKX3-1 gene at 8p21.2 promotes PrCa tumour growth by haplo-insufficincy. Mono-allelic inactivation of Nkx3-1 in adult mice results in the development of hyperplasia and apparent PIN lesions in the prostate (Abdulkadir et al. 2002; Bhatia-Gaur et al. 1999; Kim et al. 2002a). Several alterations have been observed for the LZTS1 gene at 8p21.3, including significantly reduced mRNA expression levels, sparce missense mutations and germline sequence variants. In addition, restoration of LZTS1 function by gene transfer experiments implicated LZTS1 as a tumour suppressor gene (Hawkins et al. 2002; Ishii et al. 1999; Vecchione et al. 2001). We found the WRN gene at 8p12 to be disrupted by a homozygous deletion in one of our PrCa xenografts. The WRN protein belongs to the highly conserved family of RecQ helicases. WRN and other family members like BLM and RECQ4 are associated with rare chromosomal instability disorders that also include a predisposition to develop cancer (Hickson 2003). Heterozygous human WRN (+/-) cells expressed reduced levels of WRN protein and had reduced helicase activity, which was associated with an increased sensitivity to DNA damaging agents and genetic instability (Moser et al. 2000b). Haplo-insufficiency may also be of importance for other RecQ helicases, because Blm (+/-) heterozygous mice showed an enhanced tumour formation (Goss et al. 2002). Lastly, I believe the EPHX2 gene at 8p21.1 to be a promising candidate tumour suppressor gene in PrCa. We identified germline sequence

variants of the *EPHX2* gene in 4 out of 43 PrCa samples, whereas no variations were identified in 50 control samples. Importantly, all of 4 PrCa samples with a variant *EPHX2* allele had lost the wild-type *EPHX2* allele, suggesting bi-allelic inactivation of the gene. Although the numbers are small, it is tempting to speculate that *EPHX2* is a low-penetrant susceptible gene in PrCa.

How to hunt for genes inactivated by haplo-insufficiency across the genome?

The majority of tumour suppressor genes identified to date follow the classical paradigm of bi-allelic inactivation. Bi-allelic inactivation of a tumour suppressor gene is however presumed to have a stronger effect on clonal outgrowth than mono-allelic inactivation of the gene (Fero et al. 1998, Smits et al. 2000), which inevitably will have biased the identification of the classical tumour suppressor genes. Indeed, the tumour types in which they have been identified generally are rather aggressive in their growth pattern. In this respect, clinical diagnosed PrCa generally has a late age of onset and is a slow growing tumour type. It may be that the tumour suppressor genes involved in PrCa commonly exert their tumour promoting effect by haplo-insufficiency. If so, the identification of these genes may require a different approach. The more subtle the tumour promoting effect of a mutation in a tumour suppressor gene, the larger the amount of tumour samples necessary to prove such an effect. In other words, searches for PrCa genes should exploit high-throughput screening procedures.

Recent technological developments, in particular microarray analysis, now allow the performance of many tests in parallel. On a microarray, DNA or protein fragments are organized in a gridded way at a flat surface, such as a glass slide or a porous substrate. Microarray analysis can be applied to identify both larger chromosomal alterations and smaller genomic changes at the nucleotide level (e.g. CGH and SNP microarrays, respectively). Large-scale expression analysis can be performed on RNA and protein level using cDNA and oligo microarrays and protein microarrays, respectively. The integration of such high-throughput genomics and proteomics analysis with functional studies should eventually result in the identification of the PrCa tumour suppressor genes on chromosomes 5q, 8p, 13q and 16q.



Table I. HPC loci and candidate genes	e Family origin References Gene analyzed Gene alterations References (age of onset) in HPC	North Amerika (Chang et al. 2002) <i>HSD3B1</i> Combination of (Chang et al. 2002) (±61) HSD3B2 polymorphisms	North America and (Gibbs et al. 1999 ; Xu et al. 2001b) P73 Silent (Peters et al. 2001a) Canada (<66) splice sites) splice sites)	South and West (Berthon et al. 1998; Berry et al. 2000a; - Europe (<60-66) Cancel-Tassin et al. 2001)	North America, (Smith et al. 1996; Cooney et al. 1997; <i>RNASEL</i> Nonsense (Carpten et al. 2002) Sweden, Finland Hsieh et al. 1997; Gronberg et al. 1999; (<55-69) 2000a; Goode et al. 2000; Xu 2000)	Ashkenazi Jews (Xu et al. 2001c) PG1 Homozygous (Xu et al. 2001c)	Sweden (<65) (Wiklund et al. 2003) LZTS1 Missense (Hawkins et al. 2002)	MSR1 Missense (Xu et al. 2002a ; Xu et al. 2003)	North America (Gibbs et al. 2000)	North America (Paris et al. 2000; Suarez et al. 2000) -	North America (Tavitgian et al. 2001) <i>ELAC2</i> Frame-shift (Rebbeck et al. 2000; Rokman et al. (<>) Missense 2001; Tavitgian et al. 2001; Vesprini Combination of et al. 2001; Vesprini Ser217Leu and Ala541Thr	North America (Berry et al. 2000b; Zheng et al. 2001)	(>es) North America, (Xu et al. 1998; Schleutker et al. 2000; - Sweden, Finland Peters et al. 2001b) (>60)
	Family origin F (age of onset)	North Amerika (((±61)	North America and ((Canada (<66)	South and West (F Europe (<60-66) C	North America, ((Sweden, Finland H (<55-69) N	Ashkenazi Jews ()	Sweden (<65) (V		North America (((>65)	North America (F	North America (<>)	North America (E	 >(>65) North America, () Sweden, Finland P (>60)
	Susceptible locus	1p13	1p36	1q42.2-q43	1q24-q25	8p22-p23			÷	16q23	17p11	20q13	Xq27-q28
	Symbol		CAPB (BC-PC)	PCAP	HPC1						HPC2	HPC20	НРСХ

Appendix A1 | 133

stails on chromosom

currence	pp'98 (n=37)	% (1p31 + 1p36-pter) *		。(2p24-pter)	% (2œn-q22)		o (3qcen-q23)	o (4cen-pter)	o (4q34-qter)		% (5q15-q23)*	o (6pter)	% (6q16 + q24-qter)*	, (7p14-pter)		% (8p12-p22 + p23)*		o (9pter)	o (9cen-qter)	% (10p11)*	% (10cen-q21 + q26)*	o (11p15-pter)	% (11q24-qter)	o (12p12-pter)	% (12q24-qter)	% (13q12 + q21)*	o (14q31)	% (15cen-q21 + q25-qter)*	% (16p12-pter)	% (16q24) *	% (17p) *		o (18cen-pter)	% (18q22-qter)*	% (pter-q13)*		% (20cen-pter)	% (20cen-q22)*	。(21cen-qter)	% (22q13)*	% (Xpter)	。(Xq27-qter)			
Recurrence Re	Vis'95 (n=9) Nu	22% (1p21-p31) 54°	pu	22% (2p15-p21) 5%	22% (2q11-q31) 11 ⁴	11% (3p22-pter) nd	nd 5%	22% (4p15-pter) 3%	nd 5%	nd nd	44% (5q14-q23)* 16°	22% (6p23-pter) 8%	44% (6q13-q21)* 27°	nd 3%	11% (7q22-qter) nd	78% (8p21-pter)* 73°	pu pu	22% (9p21-pter) 3%	nd 3%	nd 22°	11% (10cen-q23) 46'	nd 8%	nd 14 ⁶	22% (12cen-pter) 5%	11% (12q24) 14 ^c	56% (13cen-q21)* 51°	28 pu	nd 35'	nd 16 ^c	56% (16q22-qter)* 46'	22% (17p12-p13) 41°	11% (17q21-q24) nd	nd 5%	22% (18q12-qter) 19 ⁶	nd 43	22% (19cen-qter)	nd 14 ⁶	nd 22°	3% g	11% (22cen-qter) 46°	nd 11 ⁶	nd 5%	pu pu		
Distant metastases	Alers'00 (n=5)	20%	20%	pu	33%	pu	pu	pu	33%		53%	pu	73%	pu	pu	53%	pu	20%	pu	pu	33%	pu	pu	27%	pu	80%	pu	pu	pu	40%	pu	pu	pu	27%	pu	pu	pu	pu	pu	pu	pu	pu	27%		
LN-metastasis	Alers'00 (n=18)	8%	6%	pu	11%	pu	pu	pu	11%	pu	11%	pu	22%	pu	pu	39%	pu	11%	pu	pu	22%	pu	pu	6%	pu	39%	pu	pu	pu	11%	pu	pu	pu	22%	pu	pu	pu	pu	pu	pu	pu	pu	6%		
Metastasis	Cher'96 (n=20)	25%	30%	15%	42% * (50%)	25%	15%	30%	20%	20%	39% * (45%)	20%	39% * (45%)	15%	20%	80% * (80%)	5%	5%	10%	25%	50% * (50%)	30%	15%	25%	15%	75% * (75%)	20%	39% * (45%)	15%	55% * (55%)	50% * (50%)	20%	15%	15%	5%	15%	20%	20%	5%	10%	10%	15%	20%	eu	
Primary PrCa	Alers'00 (n=23)	6%	4%	pu	13%	nd	pu	pu	13%	pu	35%	pu	39%	pu	pu	52%	pu	9%	pu	pu	4%	pu	pu	6%	pu	65%	pu	pu	pu	26%	pu	pu	pu	13%	pu	pu	pu	pu	pu	pu	pu	pu	17%	orpi, Cher, and Nuppon	
Primary PrCa	Vis'95 (n=31)	6% (1p32-pter)	3% (1q25-qter)	6% (2p16)	6% (2q23-q33)	3% (3cen-p21)	pu	pu	6% (4q28-q34)	pu	6% (5q14)	pu	22% (6cen-q21)*	pu	3% (7q21-q35)	32% (8p12-pter)*	pu	16% (9p23-pter)*	6% (9q34)	pu	10% (10q22-q23)	pu	3% (11q14-q24)	pu	3% (12q24-qter)	32% (13q21-q31)*	pu	3% (15q22-qter)	6% (16cen-pter)	19% (16cen-q23)*	6% (17cen-pter)	6% (17cen-qter)	pu	19% (18q22-qter)*	6% (19cen-pter)	6% (19cen-qter)	pu	3% (20cen-qter)	pu	13% (22qcen-qter)	pu	3% (Xq21-qter)	pu	s determined by Visakc	
NId	Loss Zitz'01 (n=12)	1p nd	1q nd	2p 8%	2q nd	3p 8%	3q 8%	4p nd	4q 17%	5p nd	5q 8%	6p nd	6q 8%	7p nd	7q hd	8p 8%	8q nd	9p nd	9q nd	10p nd	10q 8%	11p nd	11q nd	12p nd	12q 8%	13q 25%	14q nd	15q nd	16p nd	16q nd	17p nd	17q nd	18p nd	18q 8%	19p nd	19q nd	20p nd	20q nd	21q nd	22q nd	Xp nd	bn pX	Y 8%	* percentages and region nd = not detected	נוח = ווחו חפופרופח

134 | Appendix A2

	NIA	Primary CaP	Primary Cap	Metastasis	LN-metastasis	Distant metastases	Recurrence	Recurrence
Gain	Zitz'01 (n=12)	Vis'95 (n=31)	Alers'00 (n=23)	Cher'96 (n=20)	Alers'00 (n=18)	Alers'00 (n=5)	Vis'95 (n=9)	Nupp'98 (n=37)
¢	8%	pu	pu	20%	pu	pu	pu	pu
þ	pu	pu	13%	52% * (55%)	11%	27%	pu	16% (1q25-q32)
d d	pu	pu	pu	45% * (50%)	pu	pu	pu	8% (2p23-pter)
8	pu.	pu	pu	25%	pu.	nd	pu	27% (2q33)*
ਲੇ	pu	pu	11%	40%	pu	27%	pu	nd
Ř	nd	pu	22%	52% * (45%)	6%	27%	nd	24% (3q25-q26)*
4	8%	pu	pd	25%	pu	pu	pu	pu
4	pu	pu	4%	40%	17%	7%	pu	19% (4q13-q23)*
Бр	pu	3% (5cen-pter)	pq	20%	nd	nd	11% (5cen-pter)	pu
2	pu	pu	pu	25%	pu	pu	pu	14% (5q14-q31)*
dg (pu.	pu.	8%	25%	17%	13%	pu	pu.
60	pu	pu	pu	15%	pu	na	nd	nd
d2	25%	6% (7cen-pter)	13%	35%	pu	27%	56% (7p13)*	32% (7p15-p21)*
Ъ2	25%	6% (7q32-qter)	13%	45%	11%	47%	pu	43% (7q21 + q31)*
8 8	pu	pu	pu	15%	nd	nd	nd	pu
8	42%	6% (8q)*	35%	85% * (85%)	17%	90%	89% (8q)*	73% (8q21 + q23-qter)*
96	pu	pu	pu	40%	pu	pu	pu	19% (9p21-pter)
Ъ6	pu	pu	17%	45%	11%	27%	pu	pu
10p	pu	pu	pu	25%	5%	20%	pu	nd
10q	pu	pu	pu	20%	pu	20%	pu	11% (10q21)
11p	pu	3% (11cen-p14)	pu	52% * (55%)	pu	pu	11% (11cen-pter)	11% (11p14)
11q	8%	3% (11q14-q22)	pu	30%	pu	pu	11% (11cen-q22)	22% (11q22)*
12p	pu	pu	pu	10%	pu	pu	pu	3% (12p12-pter)
12q	17%	pu	pu	30%	pu	nd	pu	22% (12q21)*
13q	pu	pu	pu	20%	pu	pu	pu	19% (13q31)*
14q	pu	pu	pd	20%	pu	nd	pu	11% (14q21)
15q	17%	pu	nd	25%	nd	nd	nd	hd
16p	25%	pu	nd	40%	nd	nd	pu	8% (16cen-pter)
16q	pu	pu	nd	10%	nd	nd	nd	pd
17p	25%	pu	nd	5%	nd	20%	nd	hd
17q	25%	pu	8%	35%	33%	33%	nd	8% (17q22-qter)
18p	pu	pu	nd	35%	nd	nd	nd	19% (18pter)
18q	pu	pu	pu	35%	nd	pu	pu	30% (18q12)*
19p	33%	13% (19cen-pter)	bd	20%	nd	nd	pu	pd
19q	33%	pu	nd	25%	pu	nd	pu	pu
20p	25%	pu	nd	35%	nd	nd	pu	3% (20cen-pter)
20q	25%	pu	pu	10%	nd	pu	nd	3% (20q12-qter)
21q	pu	pu	pu	20%	pu	nd	pu	5% (21q22-qter)
22q	17%	pu	pu	20%	pu	pu	pu	nd
ď	pu	pu	17%	20%	11%	20%	56% (Xp11-q13)*	24% (Xp21-pter)*
ъХ	17%	pu	22%	25%	11%	40%		35% (Xcen-q13)*
≻	pu	pu	pu	10%	pu	pu	pu	pu
* perce	antages and regions o	letermined by Visakorp	oi, Cher, and Nupponen					
10 1	or detected							

Appendix A3 | 135

Gene	Summary	Tumour rel	ated alterations in PrCa	References	Tumour r tumours	elated alterations in other	References
N33	Locus: 8p22 (chr8:15.2-15.4) Function: unknown Similarity: belongs to the ost3 family.	HD Nonsense Frameshift Missense Low expr.	1 / 52 metastases	(Bova et al. 1993)	HD Mutation Low expr.	 1/3 pancreatic tumor cell line 1/15 colorectal cancer cell lines 1/8 lung cancer cell lines 1/4 liver cancer cell lines 	(Levy et al. 1999) (Macgrogan et al. 1996)
PDGFRL	Locus: 8p22 (chr8:17.5-17.6) Function: unknown Similarity: 27% and 25% identity with the axtracellular comain of platelet-derived growth factor receptor-beta (PDGF-B) and FMS- like tyrosine kinase.	HD Nonsense Frameshift Missense Low expr.	- - 168 primary PrCa 1/68 primary PrCa 0/15 xenografts and cell lines 0/15 xenografts and cell lines	(Komiya et al. 1997) (van Alewijk et al., unpublished results)	HD Mutation Low expr.	- 2 / 48 hepatocellular carcinomas 1 / 28 colorectal carcinomas -	(Fujiwara et al. 1995)
LZTS1	Locus: 8p21.3 (chr6:20.1) Function: unknown Brillairty: DNA-binding domain of the cAMP-responsive activating transcription factor-5	HD Nonsense Frameshift Missense Low expr	nt 1 / 3 cell lines - 1 / 15 xenografts and cell lines 1 / 15 xenografts and cell lines	(Ishii et al. 1999) (van Alewijk et al., unpublished results)	HD Mutation Low expr.	 - 53 esophageal squamous cell carcinomas - 1 / 39 gastric carcinomas 60% epithelial tumors 8 8 gastric carcinomas 98 gastric carcinomas 98 gastric carcinomas 	(Ishii et al. 1999) (Vecchione et al. 2001) (Ishii et al. 1999) (Ishii et al. 1999)
NKX3.1	Locus: 8p21.2 (chr6:23.9) Eunction: transcription factor, which binds preferentially the consensus sequence 5'-taag(fag)-3' and can behave as a transcriptional repressor. Could play an important repressor. Could play and an important repressor. Could play an im	Mutation Low expr	0 / 50 prim PrCa and metastases metastases 0 / 15 prim PrCa 0 / 15 senografs and cell lines 11 / 52 prim PrCa 78 % metastases 4 / 15 xenografts and cell lines	(Voeiler et al. 1997) (Ornstein et al. 2001) (Van Alewijk et al. submitted) (Xu et al. 2000) (Xu et al. 2000) (van Alewijk et al., unpublished results)	E		

EPHX2	Locus: 8p21.1 (chr8:27.9) harthu epoxide-convers potentially harthu epoxide-contraining compounds into diols, which are less reactive and easier to excrete (Meijer and DePierre 1988). Similarity: belongs to the epoxide hydrolase family.	Mutation Low expr.	0 / 15 xenografts and cell lines 2 / 15 xenografts and cell lines	(van Alewijk et al., unpublished results)	ŧ		
CLU	Locus: 8p21.1 (chr8:28.0) Eunction: not yet clear, it is known to be expressed in a variety of fissues and it seems to be able to bind to cells, membranes, and hydrophobic programmed cell death (apoptosis). Similarity: belongs to the clusterin family.	Mutation Low expr.	0 / 15 xenografts and cell lines 0 / 15 xenografts and cell lines	(van Alewijk et al., unpublished results)	ŧ		
WRN	Locus: 8p12 (chr8:31.1-31.5) Function: seasential for the formation of DNA replication focal centers: stably associates with foci elements generating inding state for tp-a. Exhibits a magnesium-dependent ATP-dependent DNA helicase activity. May be involved in the control of genomic stability. Similarity: belongs to the helicase family (RECQ subfamily).	HD Mutation Low expr.	1 / 15 xenografts and cell lines 0 / 15 xenografts and cell lines 0 / 15 xenografts and cell lines	(Van Alewijk et al. 1999) (Van Alewijk et al., submitted) (Van Alewijk et al., submitted)	ŧ		
BIOA	Locus: 8p11.21 (chr8:42.2-42.3) Euction: repart polymerase, which conducts "gap-filling" DNA synthesis in a stepwise distributive frashion arther than in a processive frashion as for other DNA polymerases. Similarity: belongs to DNA polymerase type-x family.	Mutation Frameshift Missens Low expr.	0 / 15 xenografts and cell lines 1 / 12 prim PrCa 1 / 12 prim PrCa 0 / 15 xenografts and cell lines	(van Alewijk et al., unpublished res ults) (Dobashi et al. 1994) (van Alewijk et al., unpublished res ults)	HD Mutation Low expr.	0. / 14 bladder cancer cell lines 0./ 90 bladder carcinomas 2./ 6 colorectal carcinomas	- 1997) (Wang et al. 1992)

Locus is obtained from UCSC April 2003 freeze database Function and similarity are obtained from Weizman GeneCards (red NCBI LocusLink)

	References			(Gimm et al. 2001)	
Ø	elated alterations in other			- 1 / 4 (15) follicular thyroid carc. -	
ed in PrC	Tumour r tumours	HD Mutation Low expr.	HD Mutation Low expr.	HD Mutation Low expr.	HD Mutation Low expr.
mosome 10 involv	References	(Naria et al. 2001)	(Trapman et al., unpublished results) (Srivastava et al. 2001)	(Hermans et al., submitted)	(Hermans et al., submitted)
Table V ssor genes on chro	elated alterations in	- 1 / 33 primary PrCa - 1 / 33 primary PrCa 	0 / 15 xeno. and cell lines 1 / 7 (20) prim. PrCa - -	1 / 15 xeno. and cell lines - -	1 / 15 xeno. and cell lines - 5 / 15 xeno. and cell lines
ur suppre	Tumour re PrCa	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.
Candidate tumo	Summary	Locus: 10p15.2 (chr10: 3.9) Function: transcriptional activator (by similarity), which binds a GC box motif. Could play a role in b-cell growth and development. Similarity: belongs to the krueppel family of C2H2-type zinc- finger proteins	Locus: 10q22.2 (chr10: 75.0-75.1) Function: calcium-dependent membrane- binding protein, fuses membranes and acts as a voltage-dependent calcium channel schainerity: belongs to the annexin familiy similarity: belongs to the annexin familiy binding proteins	Locus: 10q23.31 (chr10 : 89.4) Function: dephosphorylates Ins(1,3,4.5)P4. hydrolyzes inositol pentakisphosphate and inositol hexakisphosphate Similarity: is a distinct group within the histidine phosphatese family.	Locus: 10q23.31(chr10: 89.6) Function: Bifunctional polypeptide; has ATP sufurylase and adenoratine 5:- ATP sufurylase and adenorations: phosphoadenorations 6: photophosultate for the synthesis of suffornate donor 3- tor the synthesis of suffornate donor 3- binitiarity: the N-terminal section belongs to the APS kinase family. The C-terminal section belongs to the suffate adenylytrasferase family.
	Gene	COPEB	ANXA7	MNP1	PAPSS2

(Cantley and Neel 1999)		- (Lee et al. 2000) (Santouriidis et al. 2001)	(Wang et al. 2000)	(Mollenhauer et al. 1997) (Somerville et al. 1998) (Takeshita et al. 1999) (Wu et al. 1999) (Wu et al. 1999) (Mueller et al. 2002) (Mori et al. 1999) (Mori et al. 1999) (Mori et al. 1999) (Wu et al. 1999) (Wu et al. 1999)
Glioma, prostate, kidney, endometrium and breast carcinoma cell lines or tumour specimens		- 3 / 21 burn scar-related squamous cell carcinomas 3 / 15 bladder carcinomas	2 / 16 bladder carc. cell lines	 2 / 20 medulloblastomas 9 / 39 glioblastomas 8 / 21 glioblastomas 2 / 23 lung carc. cell lines 2 / 24 lung carc. cell lines 2 / 24 brant unnor cell lines 5 / 79 gliomas 4 / 5 brant unnor cell lines 5 / 79 gliomas 4 / 24 concretal carc. 12 / 15 oesophaa carc. cell lines 5 / 4 ogastric carc. 2 / 2 gliomas 2 / 2 lines 5 / 9 gliomas 6 / 14 NSCLC cell lines
HD Mutation Low expr.	HD Mutation Low expr.	HD Mutation Low expr.	HD Mutation Low expr.	HD Mutation Low expr.
(Li et al. 1997) (Caims et al. 1997) (Whang et al. 1998) (Caims et al. 1997) (Steck et al. 1997) (Steck et al. 1997) (Caims et al. 1998) (Feilotter et al. 1998)	(Hermans et al., submitted)	(Takayama et al. 2001) (Takayama et al. 2001) (Hermans et al., submitted) (Santourlidis et al. 2001)	(Prochownik et al. 1998) (Eagle et al. 1995) (Prochownik et al. 1998) (Ergele et al. 1998) (Prochownik et al. 1998)	(Hermans et al., submitted)
2 / 4 cell lines 6 / 23 (80) prim. PrCa & LNM 1 / 10 cell lines 1 / 23 (80) prim. PrCa & LNM 1 / 3 cell lines 3 / 23 (80) prim. PrCa & LNM 1 / 25 (51) prim. PrCa	 4 / 15 xeno. and cell lines - 1 / 15 xeno. and cell lines 4 / 15 xeno. and cell lines 	- 1/11 high grade PIN - 1/11 high grade PIN 1/15 xeno. and cell lines 4/15 prim. PrCa	- 1 21 (40) prim. PrCa 2 / 10 (10) prim. PrCa 1 / 21 (40) prim. PrCa 2 / 10 (10) prim. PrCa 2 / 10 (10) prim. PrCa	1 / 15 xeno. and cell lines - - 2 / 15 xeno. and cell lines
HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Frameshift Missense Low expr.	HD Nonsense Framsense Missense Low expr.
Locus: 10q23.31 (chr10: 89,8-89.9) Function: dual specific phosphatase: anagonizes signal transduction of P1-3 kinase by dephosphorylating Ins(3,4,5)P3; also active as a phosphatase on tyrosine, serine and threonine residues. Similarity: belongs to the non-receptor class of the protein- tyrosine phosphatase family.	Locus: 10q23.31 (chr10: 90.2-90.5) Function: unknown Similarity: none, hypothetical protein	Locus: 10q23.31 (chr.10: 90.9) Function: signals apoptotic cell death in sensitive cells similarity: member of the tumour Similarity: member of the tumour family	Locus: 10q25.2 (112.1-112.2) Function: transcriptional repressor, which heterodimerizes with MAX to antagonize MYC activity MYC activity MYC activity Similarity: belongs to the basic helix-loop- helix (bHLH) family of transcription factors.	Locus: 10q26.13 (gap) Function: Little is known about its function but a transcript variant has been shown to bind surfactant protein D independent of carbohydrate recognition. This indicates that DMBT i may not be a classical tumour suppressor gene, but rather play a role in the interaction of tumour cells and the immus system. Similarity: Member of the scavenger receptor cysteine-rich (SRCR) superfamily
PTEN	FLJ11218	TNFRSF6	MXI1	TaMd

A HIGH-RESOLUTION PHYSICAL AND TRANSCRIPT MAP OF CHROMOSOME 8P12-P21

To facilitate search for homozygous deletions and mapping of candidate tumour suppressor genes we isolated the 8p12-p21 region in a large YAC contig. The contig was finished in 2000. YACs were identified and selected by standard PCR screening and Southern blotting of the CEPH YAC library (Albertsen et al. 1990) and the CEPH mega YAC library (Chumakov et al. 1992). Single YACs were tested for the presence of specific markers by PCR. The contig is bordered by the telomeric marker LPL at 8p21.3 and the centromeric marker BC028701 at 8p12. It is composed of 172 overlapping YACs and 303 markers, including polymorphic microsatellite markers, sequence-tagged sites (STSs), ESTs, genes and pseudogenes, were mapped in the contig (Table I). From selected YACs the end fragments were isolated and mapped. Chimeric YACs were included in the contig, because the 8p part of these YACs turned out to be of value for the construction of the contig and for accurate marker mapping. The final map was constructed such that a minimal number of YACs contained an internal deletion.

In 2001, the first draft sequence of the human genome was published (Lander et al. 2001; Venter et al. 2001). Recently, the completion of the human genome was announced by The International Human Genome Sequencing Consortium (Nature April 27, 2003). We compared the several consecutive releases of the gene maps of the University of California Santa Cruz (UCSC) database (http://genome.ucsc.edu) with the map deduced from the YAC contig. A minority of the genes and markers from the YAC contig were present in the 2001 UCSC map. Although the UCSC November 2002 map was detailed, some regions were differently ordered as compared to the YAC map. The UCSC April 2003 release and the YAC map show a good fit (Table I). As calculated from the human genome sequence, the YAC contig spans approximately 17 Mbp. However, as shown in Table I by grey-boxed areas, there are some small discrepancies between both maps. The most obvious discrepancy is the region around the NEF3 and NEFL genes. Some of differences might be due to previously unrecognized internal deletions in the YACs. However, although the human genome sequence from which the UCSC map is derived, is considered as over 99.9% accurate, it might still contain some alignment errors. The UCSC genome sequence contains one small gap, between Mbp 29.28 and 29.78.

Table II summarizes the ordered genes in the UCSC map and the YAC map. The UCSC map contains 95 genes, 72 of these were previously mapped in the YAC contig. For 9 genes the positions in the YAC map and UCSC map slightly differ. MGC29816 could not be located on the UCSC map. Most likely, MAPKK maps in the gap present in the UCSC map. 8p12-p21.3 contains 3 gene poor regions, between Mbp 19.9 –21.3, 30.7-32.4 and 33.3-35.5, respectively. The homozygous deletion in PC133 maps in the Mbp 30.7-32.4 segment.

LEGENDS TO THE TABLES

Supplementary Table I. YAC-contig of chromosome 8p12-p21 spanning the region between the markers LPL and BC028701. Gene names are listed in the first column. Alternative names, ESTs, ESTs by which genes were mapped, and STSs without a locus number are listed in the second column. In case a definite order could not be established, markers are boxed by a dark line. In the third column, the marker position in the UCSC April 2003 database are listed (Mbp from the top of 8p). In the fourth column, marker types are listed: (p) polymorphic marker; (s) STS; (e) EST; (g) gene; (pg) pseudogene.

Supplementary Table II. Alignment of known genes from the YAC-contig and the UCSC April 2003 database. Known genes that were mapped in the YAC contig are listed in the first column, the UCSC genes are listed in the second column. In the third column, the marker positions in the UCSC April 2003 database are listed. In the fourth column the chromosomal band is indicated.

		T 00 0001		-04/																																																		
- Di	VIGO		рд																																																			
	01010	19.61	_	904D/																																																		
	99612-1M	19.14 6	_	904U/																																																		
	stSG30884	19.76 e	_	904D7																																																		
ATP6V1B2	WI-19010	19.89 g	_	904D7																																																		
LZTS1	stSG10118	19.91 g	_	904D7 913F	10																																																	
D8S258	AFM107xb6	20.18 p		904D7 913F	-10																																																	
D8S280	AFM205wc5	20.24 p		904D7 913F	10																																																	
D8S1983	WI-10327	21.04 s		904D7 913F	F10 713G1	220G1	690F5	746H6	36	23G7																																												
D8S1509	WI-6088	21.16 s			713G1	220G1	690F5	746H6	66C7 92	23G7																																												
D8S282	AFM234vf4	21.22 p			713G1	220G1	690F5	746H6	66C7 92	23G7 900G	12																																											
D8S1116	CHLC.ATA18B1	21.24 p			713G1	220G1	690F5	746H6	66C7 92	23G7 900G	12																																											
D8S196	cC18-281	<u> </u>			713G1	220G1	690F5	746H6	66C7 92	23G7																																												
GFRA2	SHGC-31819	21.36 g			713G1	220G1	690F5	746H6	66C7 92	23G7																																												
D8S560	AFMa127ye5	21.41 p			713G1	220G1	690F5	746H6	66C7 92	23G7																																												
D8S232	cCl8-448	<u> </u>			713G1	220G1	690F5	746H6	66C7 92	23G7																																												
	57D1R	. 0				220G1	690F5	746H6	56C7 92	33G7 57D	-																																											
	E-1077-2								56C7 95	23G7 57D																																												
	AEM004:440								5	-0-2 5.064																																												
007200		1 00.12	_						10	-023 2062																																												
	stsG36099	6 AC'17							76	1/6 /562	-																																											
D8S133		21.79 p	_						36	23G7 57D	1 675A7																																											
D8S1937	WI-8527	21.62 e								57D	1 675A7																																											
	stSG16422	21.77 e								57D	1 675A7																																											
RAI16	R30665	21.78 g								57D	1 675A7																																											
RAI16	A005V15	21.78 0								57D	1 675A7																																											
НВ	WI-6304	21.79 0	_							27D	1 675A7																																											
н		2179 0								57D	1 675A7																																											
	46769.0		_							57D.	-	467E9																																										
	40/EZ L	0								5																																												
EPB49	WI-20118	21./6 9	_							0/9	1 6/5A/	46/E2																																										
HH	stSG4685	21.79 9	_							5/D	1 6/5A/	46/E2																																										
XP07	SGC33989	21.68 g	_							57D	-	467E2																																										
XP07	R20754	21.68 g	_							57D	-	467E2																																										
LGI3	WI-15206	21.82 g	_							57D	-	467E2																																										
	E-1025-5	S								57D	-	467E2	238C6																																									
SFTPC	stSG9073	21.84 g	_							57D	-	467E2	238C6																																									
BMP1		21.84 g	_							57D	-	467E2	238C6																																									
FLJ22246	stSG47781	21.82 g								57D	-		238C6																																									
РНҮНІР	WI-6273	21.90 g								57D	-		238C6	777G4																																								
	57D1L	S		730H9						57D	-		238C6	777G4																																								
	WI-7676	21.93 g		730H9 672h	110 670GE								238C6	777G4																																								
	stSG52127	21.93 e		730H9 672H	110 670GE								238C6	777G4																																								
PIWIL2	stSG53946	22.03 g		730H9 672H	110 670GE								238C6	777G4																																								
	A009B38	22.10 e		730H9 672F	110 670GE		~	310A1					238C6	777G4																																								
D31887	WI-6536	22.10 e		730H9 672F	110 670G5			310A1					238C6	777G4																																								
PPP3CC	stSG3913	22.22 g		672H	110 670G5	5 287C5		310A1					238C6	777G4																																								
SCAM-1	SHGC-30970	22.25.0		6725	110 670G5	2870.5	-	310A1					23806																																									
PDI IM2	ctSG8068	22.27	_	6721	110 67065	28705	-	31041																																														
FL 194715	et SGROOPS	0 20 28 0	_	672	110 67065	20100	-	10012					23806																																									
		2	_	;		20.00							20004		_																																							
																																															81G5					81G5	81G5	
----------	---------	---	--------	--------	--------	--------	--------	---------	--------	-----------------	--------	----------	----------	--------	----------	--------	--------	--------	---------	----------	--------	---------	-------	--------	---------	--------	---------	--------	---------	---------	--------	--------	--------	--------	--------	--------	---------	--------	---------	--------	--------	---------	-------	--------	---------	---------	---------	--------	--------	--------	--------	----------	----------	--
																																															308F9 9				,	308F9 9	308F9 9	
																																														767G4	767G4 8	'67G4	767G4	767G4	767G4	67G4 8	67G4 8	
																																													984A3	384A3	384A3	384A3	384A3	984A3	984A3	384A3	984A3	
																																													926H8	926H8	926H8	926H8	926H8	926H8	926H8	926H8	926H8	
																																													949B8	949B8	949B8	949B8	949B8	949B8	949B8	949B8	949B8	
																																820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6	820H6							965E1	
																									1220	8/5E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4	875E4										
238C6	238C6		238C6	238C6	238C6	238C6	238C6													871E3	871E3	871E3	871E3	871E3		8/1E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3	871E3								46E9	46E9	46E9	
																											730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8	730F8										799G9			
																													734F10	734F10	734F10	734F10	734F10	734F10	734F10	734F10	734F10	734F10	734F10	734F10									155D10	155D10	155D10	155D10	155D10	
																												721D10	721D10	721D10	721D10	721D10	721D10	721D10	721D10	721D10	721D10	721D10	721D10										686G9			686G9	686G9	
																786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7	78657	1 100 /	/ 801/	786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7	786F7										784E10	784E10	784E10	784E10	784E10	784E10	
															410A4	410A4	410A4	410A4	,	410A4	410A4	410A4																										769E10	769E10	769E10	769E10	769E10	769E10	
810A1	810A1		810A1	810A1	810A1	810A1	810A1	810A1	810A1	81041	810A1		810A1	810A1	810A1	810A1	810A1	810A1	810A1	810A1	810A1	810A1	810A1	81041																							733C10	733C10	733C10	733C10	,	733C10	733C10	
					98A1	98A1	98A1	98A1	98A1	0801	98A1																																					758H6	758H6	758H6	758H6	758H6	758H6	
287C5					287C5																																										240G10	240G10	240G10	240G10	240G10	240G10	240G10	
670G5	670G5		670G5	670G5	670G5	670G5	670G5	670G5	670G5	670GE	670G5	2000																																	803C1		803C1	803C1	803C1	803C1	803C1	803C1	803C1	
672H10	672H10		672H10	672H10	672H10	672H10	672H10	672H10	672H10	670H10	672H10	0111210	672H10																																		943C4	943C4	943C4	943C4	943C4	943C4	943C4	
																																															774C1	774C1	774C1	774C1	774C1	774C1	774C1	
28 e	29 e	מ	29 g	30 g	25 p	30 g	s	34 p		28.0	37 0	n	14 e	13 e	49 p	59 g	70 g	70 g	94 g	30 g	30 e	81 a	37 P	0 1 0		6 -	13 e	16 e	98 g	17 e	17 e	22 e	23 e	23 e	24 g	25 g	25 g	29 S	25 e	26 g	26 g	52 g	2 e	30 e	03 g	s	s	30 g	58 s	56 e	54 e	33 g / p	75 p	
22.2	22.2		22.2	22.5	22.2	22.0		22.3		6	3 8	4		22.4	22.7	22.6	22.7	22.7	22.9	22.8	22.8	22.8	3 66	18	N S	23	23.	23.1	22.9	23.1	23.1	23.2	23.2	23.2	23.2	23.2	23.2	23.2	23.2	23.2	23.2	23.5		23.8	24.0			24.6	24.5	24.5	24.5	24.6	24.7	
6								Г				٦.	<u>م</u>	_	~				_	r		4			Т					-		I										_		1										
-AA03125	-H96791		G62344	-15087		G28683	3C6R	M197vc1	8-319	G 26022	15547	- 1004/	GC-3622	G31244	Mb055za9	-16696	-14836	-15306	-W49732	-11701	G34952	GC-1599	3300	100000	0002100	032556	-H92890	-16980	-U89942	GC-5249	-18184	G8140	G63330	G8498	G8797	-14743	-H40450	-961	-R98014	G52445		C35548	7343	G21122	G42867	7G4L	1G5L	-7612	-5833	G47039	G28797		Ma346ze£	
sts.	sts	1	stS	Ň		stS	235	AFI	Ö	0 ⁺⁰		A O	HS 9	sts	AF.	Ň	Ň	N.	sts.	<u>W</u>	stS	Т О	ğ		Į	5	sts	1N	sts	Ϋ́	Ň	stS	stS	stS	stS	Ň	sts.	-M	sts	stS		9S G	T9.	stS	stS	767	.86	Ň	Ń	stS	stS		AF	
								~							~	32	32	10B	116	-10C		10D																~							~	c			~				m	
_			DBC-1	DBC-1	D8S136	BIN3		D8S173	D8S220		EG B3	2			D8S175	RHOBT	RHOBT	TNFRSF	MGC296	TNFRSF		TNFRSF				LYSALI			LOXL2						MSCP	MSCP	MSCP	D8S107		NKX3-1	NKX3-1	STC1			AD AM26	D8S2210		NEF3	D8S167			NEFL	D8S173	

	R4 R A 71	0	77401	04304	803C1	240610 7	758H6 733C	010				,		757B1 0	QREF1 0	0 AGRA 0	DEHR 0	8443 76	7G4 AC	AFG QR	1GF 84	847
D8C1705	A EMAQOR-HE	2 00 10	17404	LOOP O	BURC 1	240610	758H6	2						75781 0				97 2AA9	20 B			207
		04 00 PC	17404			010010	2 00					700020		76701				97 CVV0				
NEFL	WI-16152	24.63 g	774C1	943C4	803C1							-		2				-	5.0	8F9 98	1G5 84	8A7
	SGC34359	25.09 g	774C1	943C4	803C1			257Dt	9 65D3	308A5	486F12	,		0,	965E1			- 76	57G4 80	8F9 98	1G5 84	8A7
GNRH1		25.10 g	774C1	943C4	803C1		209(310 257D:	9 65D3	308A5	486F12	799G9		0,	965E1		ð	84A3 76	57G4 8C	8F9 98	1G5 84	8A7
D8C1771	WI-15182	25,25 e	774C1	943C4	803C1	90F11 3	94D12 209(G10 257D	9 65D3	308A5	486F12 486F12	799G9	617F5 617E5	620A7 9	965E1		6	84A3 76	57G4 80	8F9 98	1G5 84	8A7 8A7
1 1 1 000	Arimuszovas SHGC-36250	25,18 e	774C1	943C4	803C1		- 2090	310 257D(9 65D3	308A5	486F12	299G9	617F5	620A7					58	8F9 98	1 <u>6</u> 5 12	ŝ,
FLJ25804	WI-11501	25,19 g	774C1	943C4	803C1	e	194D12 2090	310 257D:	9 65D3	308A5	486F12	799G9	617F5	620A7 9	965E1				80	8F9 98	1G5 84	8A7
	stSG46607	25,37 e	774C1	943C4	803C1	ę	194D12 2090	310 257D	9 65D3	308A5		799G9		0,	965E1				80	8F9 98	1G5 84	8A7
D8S5	TL11	٩	774C1															42	1C4 80	8F9 98	1G5 84	8A7
	848A7R	S	774C1																	- 98	1G5 84	8A7
	sts-T69732	25,55 e	774C1															42	104 80	8F9 98	1G5	
	A009Z01	25,94 e	774C1																	- 86	1G5	
	SHGC-36329	26,02 e	774C1																	- 98	1G5 2-	
PPP2R2A	WI-18898	26,05 g	774C1				854B8												8	8F9 981	<u>65</u>	
PPP2R2A	stSG63671	26,05 g	774C1	741B6	741B7	819B4	854B8 935(31												86	165	
	stSG15253	26,08 e	774C1	741B6	741B7	819B4	854B8 935(311												86	1G5	
BNIP3L	WI-8264	26,09 g	774C1	741B6	741B7	819B4	854B8 935(115												86	165	
BNIP3L	sts-Z40756	26,09 g	774C1	741B6	741B7	819B4	854B8 935(511												88	1G5	
	sts-W93086	26,14 e	774C1	741B6	741B7	819B4	854B8 935(115												86	1G5	
PNMA2	SHGC-36722	26,18 g		741B6	741B7	819B4	854B8 935(511														
DPYSL2	WI-6563	26,34 g		741B6	741B7	819B4	854B8 935(115														
ADRA1A	WI-17142	26,43 g				819B4	854B8 935(511		810D6												
ADRA1A	SGC35558	26,43 g				819B4	854B8 935(511		810D6												
	stSG41355	26,76 g					935(311		,												
	93A5R	S					935(311 93A£		•												
STMN4	TIGR-A002J23	26,91 g					935(311 93A5		•												
D8S2299	791B9L	s	791B9				935(311 93A£		•												
TRIM35	A009Z35	26,96 g	791B9					93A5		,				771G9								
TRIM35	stSG26015	26,96 g	791B9					93A£		•				771G9								
T RIM35	SGC33557	26,97 g	791B9					93A5		•				771G9								
	stSG62910	26,97 e	791B9					93AE		•				771G9								
PTK2		27,14 g	791B9					93A5		810D6	326D7			771G9								
D8S239A	cCl8-535	S	791B9					93AE		810D6	326D7			771G9								
D8S1839	AFMa083xb5	27,20 p	791B9					93A5	100B4	810D6	326D7		444E7	771G9								
	sts-AA011480	27,21 e	791B9					93AE	100B4	810D6	326D7		444E7	771G9								
D8S242A	cCl8-554	d	791B9					93AE	100B4	810D6	326D7		444E7	771G9								
EPHX2	SGC35790	27,22 g	791B9					93AE	100B4	810D6	326D7	189C9	444E7	771G9								
GULOP		Бd	791B9					93A£	100B4	810D6	326D7	189C9	444E7	771G9								
CLU		27,28 g	791B9						100B4	810D6	326D7	189C9	444E7	771G9								
ADRA1A	stSG21579	26,43 g	791B9							810D6	326D7	189C9	444E7	771G9 7	755B3							
D8S131		27,16 p	791B9							810D6	326D7		444E7	771G9								
FLJ10853	WI-12131	27,41 g	791B9										444E7	771G9								
	stSG30266	2/,44 e	19169										444E/	//169								
100107	147/40018	27,40 G	70100	406110									444E/	1/169	-							
161000	ete-1167000	27.48 p	70180										1444	1 001/1	COCC/							
	stSG54076	27.48 e	791B9										444E7	771G9								
торк	stSG48493	27,49 g	791B9										44E7	771G9								
D8S1161	WI-1172	27,53 s	791B9											771G9 7	755B3							

_	E-705-1	s	791B	- 6			514E7							771G9 75	55B3		
	stSG47267	27,64 e	791B	- 60			514E7							771G9	ي ،	72A3	
D8S1820	AFMa051zb1	27,82 p	791B	39 406H12	2 195D7	305C12	514E7 8	97C5						771G9 75	55B3 5.	72A3	
ELP3	stSG9277	27,84 g	791B	39 406H12	2 195D7	305C12	514E7 8	97C5						771G9	ίΩ.	72A3	
ELP3	A002C05	27,87 g	791B	39 406H12	2 195D7	305C12	514E7 8	97C5 7	98E4					771G10	20	72A3	
ELP3	WI-19683	27,87 g	791B	39 406H12	2 195D7	305C12	514E7 8	97C5 7	98E4					771G9	Ω.	72A3	
D8S2211	791B9R	s	791B	39 406H12	2 195D7	305C12	514E7 8	97C5 7	98E4 80	5B8							
	A002009	28,02 g		406H12	2 195D7	305C12	514E7 8	97C5 7	98E4 80	5B8 7501	6						
	Cda1ca05	28,03 e		406H12	2 195D7	305C12	514E7 8	97C5 7	98E4 80	5B8 750	6						
LOC55893	stSG34833	28,03 g		406H12	2 195D7	305C12	514E7 8	97C5 7	98E4 80	5B8 7501	6						
D8S1809	AFM333yg9	28,01 p		406H12	2 195D7		80	97C5 7	98E4 80	5B8 7501	6						
PNOC	WI-15701	28,02 g		406H12	2 195D7		æ	97C5 7	98E4 80	5B8 7501	6						
D8S2029	SHGC-10443	27,98 e		406H12	2 195D7		œ	97C5 7	98E4 80	5B8 7501	61						
	N33122	28,13 e					œ	97C5 7	98E4 80	5B8 7501	61						
FZD3	sts-AA024552	28,18 g					80	97C5 7	98E4 80	5B8 7501	61						
FZD3	SGC31127	28,24 g					æ	97C5 7	98E4 80	5B8 7501	6						
	stSG2251	28,25 e	281F	12			80	97C5 7	98E4 80	5B8 7501	61						
	A007B14	28,25 e	281F	12			80	97C5 7	98E4 80	5B8 7501	60						
EXTL3	Cda13d01	28,43 g	281F	12 645B9			æ	97C5 7	98E4 80	5B8 7501	61						
FLJ10871	WI-6274	28,45 g	281F	12 645B9	699D4	778B12	æ	97C5 7	98E4 80	5B8 7501	E9 887C	~					
FLJ10871	SHGC-31253	28,53 g	281F	12 645B9	699D4	778B12	æ	97C5 7	98E4 80	5B8 7501	E9 887C	~					
	stSG22385	28,70 e				778B12	œ	97C5 7	98E4 80	5B8 7501	E9 887C	7 936F7	768D6				
	A006K04	28,74 e				778B12		7	98E4	7501	E9 887C	7 936F7	768D6				
FLJ21616	SHGC-15326	28,69 g						7	98E4	7501	E9 887C	7 936F7	768D6				
FLJ21616	WI-20805	28,73 g						7	98E4	750	E9 887C	7 936F7	768D6				
FLJ21616	stSG62535	28,73 g						7	98E4	7501	E9 887C	7 936F7	768D6				
KIF13B	WI-14186	28,75 g						2	98E4	7501	E9 887C	7 936F7	768D6				
	stSG39502	28,76 e						7	98E4	750	E9 887C	7 936F7	768D6				
KIF13B	SHGC-52529	28,87 g								7501	E9 887C	7 936F7	768D6				
KIF13B	WI-16549	28,87 g								750	E9 887C	7 936F7	768D6				
	stSG15815	28,89 e								750	E9 887C	7 936F7	768D6				
	SHGC-36290	28,91 e								7501	E9 887C	7 936F7	768D6				
	A009U09	28,91 e								7501	E9 887C	7 936F7	768D6				
	750A	S		814E11	_					7501	E9 887C	7 936F7	768D6				
DUSP4	SGC35388	29,01 g		814E11	_					750	E9 887C	7 936F7	768D6				
	WI-21790	28,94 e		•						750	E9 887C	7 936F7	768D6	816C10			
	A008Q36	28,94 e		•						7501	E9 887C	7 936F7	768D6	816C10			
DUSP4	WI-22140	29,02 g		•						750	E9 887C	7 936F7	768D6	816C10			844E2
DUSP4	A004W39	29,01 g		•						750	E9 887C	7 936F7	768D6	816C10			844E2
	750C	s		•						7501	E9 887C	7 936F7	768D6	816C10			844E2
	SGC33982	29,24 e		'						7501	61	936F7	768D6	816C10			844E2
	stSG30527	29,24 e		•						7501	6	936F7	768D6	816C10			844E2
	A004P36	29,28 e		•						750	6	936F7	768D6	816C10			844E2
MAPKK		gap g	936G	4 814E11	_					750	6	936F7		816C10 82	24C8	914B12	844E2
	E-1051-1	S	936G	i4 814E11	_					750	60	936F7		816C10 82	24C8	914B12	844E2
MGC8721	WI-20115	29,78 g	936G	i4 814E11	807F11					7501	61	936F7		816C10 82	24C8	914B12	844E2
MGC8721	GAAT-P33725	29,80 g	936G	i4 814E11	807F11					7501	6	936F7		816C10 82	24C8 7	10C3 914B12 654A12	844E2
LEPROTL1	W11-6E	29,82 g	936G	i4 814E11	807F11					750	61	936F7		816C10 82	24C8 7	10C3 914B12 654A12	844E2
LEPROTL1	SGC32717	29,82 g	936G	i4 814E11	807F11					1501	61	936F7		816C10 82	24C8 7	10C3 914B12 654A12	844E2
	W13-3D-I	Ð	9360	i4 814E11	807F11					150	60	936F7		816C10 82	24C8 7	10C3 914B12 654A12	844E2
DCTN6	WI-18347	29,90 g	9366	14 814E11	807F11					120	ព	936F7		816C10 82	54C8	10C3 914B12 654A12	844E2
	AA112203	29,91 e	9366	i4 814E11	807F11					1092	6	9361-/		816C10 82	24C8 /	10C3 914B12 654A12	844E2

	750B	s	936G4 8:	14E11 807F11	_				750E9		336F7	816C10	·	14B12 654A12	844E2			
	stSG31257	29,34 e	936G4 81	14E11 807F11					750E9		336F7	816C10			844E2	729A6		
1D12A	SHGC-11047	30,13 g	- 8	14E11 -					750E9		336F7				844E2	729A6		
RBPMS		30,10 g	- -	14E11 -				762	F7 750E9	•••	936F7				844E2	729A6		
	E-1046-2	S	8	14E11 -				762	F7 750E9		936F7				844E2			
TUBB	SHGC-10364	30,07 pg	936G4 8:	14E11 807F11	_			762	F7 750E9	•	336F7				844E2			
GSR	W3-5	30,43 g	936G4 8:	14E11 807F11	_			762	F7 750E9						844E2	729A6		
	WP2252B42	30,16 e	- 8:	14E11 807F11	896F4			762	F7 750E9						844E2	729A6		
	stSG16429	30,19 e	9	14E11 807F11	896F4			762	F7 750E9						844E2	729A6		
D8S339	WT251	30,21 p	9	14E11 807F11	896F4			762	F7 750E9						844E2	729A6		
RBPMS	WT10905	30,22 g	- 8	14E11 807F11	896F4	693D8		762	F7						844E2	729A6		
	SGC34085	30,13 e	- 8	14E11 807F11	896F4	693D8									844E2	729A6		
	SGC30104	30,23 e	έō '	14E11 807F11	896F4	693D8									844E2	729A6		
	E-1055-1	s	- 8	14E11 807F11	896F4	693D8										729A6		
	W1-6	30,27 e	9	14E11 807F11	896F4	693D8							324C8			729A6		
RBPMS	W1-25	30,28 g	- 8	14E11 807F11	896F4	693D8										729A6		
RBPMS	W11-4D	30,28 g	8	14E11 807F11	896F4	693D8										729A6		
	stSG60069	30,28 e	9	14E11 807F11	896F4	693D8												
	stSG48118	30,30 e	9	14E11 807F11	896F4	693D8												
GTF2E2		30,29 g	- -	14E11 807F11	896F4	693D8							953H12			729A6		
GTF2E2	W13-5A	30,31 g	936G4 8:	14E11 807F11	896F4	693D8							953H12					
	stSG62551	30,36 e	936G4 81	14E11 807F11	896F4	693D8		780	E6 672C9				953H12					
	763A7L	s	936G4 8:	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6 672C9				953H12					
	stSG28623	30,36 e	936G4 8:	14E11 807F11	896F4	693D8	- 76	3A7 780	E6 672C9	~	304A1		953H12					
D8S540	AFM281yb9	30,37 p	936G4 81	14E11 807F11	896F4	693D8	- 76	3A7 780	E6 672C9	686G9			953H12					
GSR		30,39 a	936G4 81	14E11 807F11	896F4	693D8	- 76	3A7 780	E6 672C9	686G9			953H12					
GSR	WI-13416	30,39 a	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6				953H12					
GSR	WI-21746	30,39 g	936G4 81	14E11 807F11	. 896F4	693D8 7	86D10 76	3A7 780	E6		,		953H12					
	A006C09	30.41 e	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12					
	765D9L	S	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
D8S1630	WI-3666	30,46 s	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
D8S2298E	D8S2298E	30,48 g	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
PPP2CB		30,50 g	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
	W13-3D-II	Ð	936G4 81	14E11 807F11	896F4	693D8 7	86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
	stSG22284	30,51 e	936G4 8:	14E11 807F11	896F4	693D8 7	'86D10 76	3A7 780	E6	~	304A1		953H12	765D9				
	AFMa224wh5	30,64 p	ē	14E11 807F11	896F4	693D8 7	'86D10 76	3A7	791A1	234H4 8	304A1 9	01E10	953H12	765D9				
	SHGC-30538	30,72 e		807F11	B96F4	693D8 7	86D10 76	3A7	791A1	234H4 8	304A1 9	01E10	953H12	765D9				
	815E8L	S			896F4	693D8	'86D10 76	3A7	791A1	234H4	304A1 9	01E10	953H12	765D9		815E8		
WRN	WI-17945	30,78 g			896F4	693D8 1	86D10 76	3A7	791A1	234H4	304A1 9	01E10	953H12	765D9		815E8 7	98B11	
	721D7R	s			896F4	693D8	'86D10 76	3A7	791A1	234H4	304A1 9	01E10	953H12	765D9	721D7	815E8 7	98B11	
WRN	WRN exon 35	ŋ			896F4	693D8 1	86D10 76	3A7 613	E1 791A1	234H4	304A1 9	01E10 3	36D12 953H12	765D9	721D7	815E8 7	98B11	
	sts-N22494	30,89 e			896F4	693D8 7	86D10 76	3A7 613	E1 791A1	234H4 8	304A1 9	01E10 3	36D12 953H12	765D9	721D7	815E8 7	98B11	
	721C	s			896F4	693D8 7	'86D10 76	3A7 613	E1 791A1	234H4 8	304A1 9	01E10 3	36D12 953H12	765D9	721D7	815E8 7	98B11	
	721A	S		82D5	896F4	693D8 7	'86D10 76	3A7 613	E1 791A1	234H4 8	304A1 9	01E10 554G3 3	36D12 953H12 1	I7C10 765D9	721D7	815E8 7	98B11	
	721B	S	8	48D12 82D5	896F4	693D8	76	3A7 613	E1 791A1	234H4	304A1 9	01E10 554G3 3	36D12 953H12 14	17C10 765D9	721D7	815E8 7	38B11	
D8S1770	AFM238vd10	31,00 p	600D8 84	48D12 82D5	896F4	693D8	76	3A7 613	E1 791A1	234H4	304A1 9	01E10 554G3 3	36D12 953H12 1	ITC10 765D9	721D7	815E8 7	38B11	
D8S1769	AFM220yf8	31,00 p	600D8 84	48D12 82D5	896F4	693D8	76	3A7 613	E1 791A1	234H4	304A1 9	01E10 554G3 3	36D12 953H12 1	I7C10 765D9	721D7	815E8 7	98B11	
D8S124	pBS8	31,06 s	600D8 84	48D12 82D5	896F4	693D8	76	3A7 613		234H4	304A1 9	01E10 554G3 3	36D12 953H12 1	I7C10 765D9	721D7	815E8 7	38B11 76	38H6
D8S1711	AFM158xg7	31,18 p			896F4	693D8	76	3A7	•		თ	01E10 554G3 3	36D12 953H12 1	I7C10 765D9	721D7	815E8 7	38B11 78	38H6
	978B9R	s		978B9	896F4	693D8	76	3A7	•			554G3 3	36D12 953H12 1	17C10 765D9	721D7	815E8 7	38B11 78	38H6
	872E4R	s	872E4 9	978B9	896F4	693D8			•				953H12 1	17C10 765D9	721D7	815E8 7	38B11 78	38H6
D8S2217	896F4L	s	872E4 5	378B9	896F4	693D8			•				953H12 1	17C10 765D9	721D7	815E8 7	38B11 78	38H6

_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_	_		_	_	_			
788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6	788H6																																	-
798B11	798B11	798B11	798B11	798B11	798B11	798B11	798B11	798B11																																						
815E8	815E8	815E8	815E8	815E8	815E8	815E8	815E8	815E8	815E8	815E8																845G8	845G8	845G8	845G8	845G8	845G8	845G8	845G8													
721D7	721D7	721D7												268E1	268E1	268E1	268E1							302F7	302F7	302F7	302F7	302F7	302F7	302F7	302F7	302F7	302F7	302F7												
765D9	765D9	765D9	765D9	765D9	765D9		765D9							10C12	10C12	10C12	10C12	308C7	308C7	308C7				308C7	308C7	308C7	308C7	308C7	308C7	308C7	308C7	308C7														50H12
1-	1-						-							2	7	2	49C2 7	149C2 8	149C2 8	349C2 8	149C2	149C2		149C2 8	849C2 8	149C2 8	149C2 8	149C2 8	149C2 8	149C2 8	149C2 8	149C2 8	149C2	149C2									31562			315B2 / 315B2 7
53H12	53H12	53H12	53H12	53H12	53H12	53H12											ω	ω	ω	15D1 8	15D1 8	15D1 8	15D1	15D1 8	15D1 8	15D1 8	15D1 8		15D1 8	ω	w	ω	ω	ω							98E11		98E11 8	98E11	2001-	
6	90	ð	6	6	ð	б														47B2 2	47B2 2	47B2 2	0	CI	N	CI	0		0										8000	9898	98G8 7		9868 /3	98G8 /	20000	
																				e	55C4 3	55C4 3															1000	580	1000	1280	08C1		180			
																	6H11	6H11	6H11	6H11	6H11 9	6H11 9														ļ	47F9	5 0 5 1/t	- 0 d	6 - 1 - 1	47F9 9		47F9 9	47F9 47E0	0	
															0C10	0C10	0C10 30	0C10 30	0C10 30	0C10 30	0C10 30	0C10 30															56E1 8	56E1 8	00E 1 20E 1 8		56E1 8		56E1 8	56E1 &		
													51E7	51E7	51E7 42	51E7 42	51E7 42	51E7 42	51E7												1G11	1G11	2191	2 F	: F	2	1G11 75	< i	× 1	2 1131						
				91A1	91A1	91A1	91A1						22	9F6 75	9F6 75	9F6 7!	9F6 75	17	12	12	12	12	12												8G11 95	8G11 95	- 95	8611 95		8611	8G11 95	1.00	8G11	S0	3	
				32	52	52	52							7E8 7	7E8 7	7E8 7	7E8 7																	9A11	9A11 89	- 89	-	111 896 FLA	08	5	68		0F3 891	OF3		0F3
														tD11 28	tD11 28	28	28																	ő	ö6	G12		-			9G12	202		12 AG	202	D R
													3B9	3B9 754	75/																			9A2	9A2	936					7D5 936	מים הבי	705	7D5 936		70%
3D8	3D8	3D8	3D8	3D8		2F2	2F2	2F2	2F2	2F2	2F2	2F2	2F2 44	2F2 44	2F2	2F2	2F2											3G9	3G9				9 G 9	3G9 45	46						2D5 90		205 90	2D5 90		2D5 90
69	69	69	69	69	D12	D12 74	D12 74	D12 74	D12 74	- 74	D12 74	D12 74	D12 74	D12 74	D12 74	D12 74	74											716	IE1 716	Ē1	Ē	Ē	716	710							5E5 95.		DE5 952			5E5 95.
					IA2 888	A2 888	A2 888	IA2 888	IA2 888	A2	A2 888	IA2 888	IA2 888	A2 888	IA2 888	A2 888	IA2	IA2	A2	A2	A2	A2						G12	G12 71 ⁻	G12 71 ⁻	G12 71 ⁻	G12 71 ⁻							ŝ	PH I	6 10 100		6H3 90	NA SU		5H3 906 5H3 906
8B9	B9	B9	889	88	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761	B9 761						D5 726	D5 726	726	726	D5 726							ORF	DCA	B4 956		184 956	B4 950		1B4 956
E4 978	E4 976	E4 978	E4 978	E4 976	E4 978	E4 978	E4 976	E4 976	E4 976	E4 976	E4 976	326	376	976	376	326	976	976	376	326	976	976					11	C11 627	C11 627		Ē	C11 627				-	512		1 0 0	212	312 960	106 210	312 960	312 90U		312 9ou 960
872	872	872	872	872	872	872	872	872	872	872	872																7830	7830	7830	7830	7830	7830					847	84/1 12/12	1710		8471		847	1/4/1	1150	84.1
31,42 e	31,42 e	S	S	31,55 p	s	S	31,76 e	31,84 p	31,85 e	S	s	32.06 e	32.22 p	32,40 p	32,48 g	S	S	32,76 e	32,76 e	32,81 s	32,91 p	32,92 p	33,07 e	33.31 g	33,31 g	o o	33,33 p	33,55 p	33,21 g	33,48 p	33,48 p	33,53 p	chr17 g	s	34,31 p	34,74 e	34,93 s	35,00 e	20,14 8	35,10 p	35,25 p		35,66 p	35,50 p		36,25 p
~	60			5				yg9	56			00	79G02	e1				7	33		zf1	b2	8099		33		e5	69	177	xg5	h12	/a9			Ŋ	22		224	T	Ţ	xd5	yua 	AZZAU	Ę,]	
stSG2166	stSG3060	721D7L	393D8L	AFM345t	761A2L	742F2R	WI-12975	AFMa143	stSG468t	315E8R	372E4L	sts-Z4090	CHLC.AT	4FM200V		388D12L	742F2L	SGC3387	sts-N333	оL8	AFMb292	AFM107y	sts-AA02	stSG4710	stSG1007	E-1030-2	4FM127x	AFM265z	SHGC-32	AFMa165	AFM238y	AFM211 ⁴	WI-6601	849C2L	AFM198t	stSG476	WI-5543	SHGC-32	VVI-400U		AFMa192		CHLC.AI	AFMD280	10+0/12	4FM312A MI-11525
	55		<u> </u>	`			_	~	3)			0					.~		9)		. ~	4					4		.,		~	<u>`</u>			-		<u> </u>	<u> </u>				<u>, </u>	~ 1	<u>~ </u>	~	
				10	26								25	8			21				58	6		136	136		45	2	549	60	8	e	8	20	5	;	10	36	00		19	00	21		- [;]	203 201
				D8S18	D8S22								D8S11	D8S27(NRG1		D8S22			D8S71	D8S17	D8S25(MGC1	MGC1		D8S18-	D8 S53	LOC84	D8S17	D8S28.	D8S51.	PIP5K	D8S22	D8S50.		D8515	10 01 11		D8S8/	D8S17		D8S11.			D8S18 RC028

Supplementary Table II Alignment of known genes from the YAC-contig and the UCSC April 2003 database

	Mapping		
JNI - 2000	UCSC Apr 03	Mbp	Band
LPL	LPL	19.60	8p21.3
	SLC18A1	19.81	
ATP6V1B2	ATP6V1B2	19.88	
LZTS1	LZTS1	19.91	
GFBA2	GFBA2	21.35	
DOK2	DOK2	21.58	
	XPO7	21.68	
	FGF17	21 72	
	FPB49	21 76	
BAI16	BAI16	21 78	
10.010	FL 122494	21 78	
HB	HB	21.70	
	FL 122246	21.73	
EDB/0	1 2022240	21.01	
XP07			
LGI3	LGI3	21.82	
SFTPC	SFTPC	21.84	
BMP1	BMP1	21.84	
FLJ22246		-	
PHYHIP	PHYHIP	21.89	
	POLR3D	21.92	
PIWII 2	PIWII 2	22.03	
PPP3CC	PPP3CC	22.21	
SCAM-1	SCAM-1	22 25	
	PDLIM2	22 27	
FL 134715	FL 134715	22.27	
DBC-1	DBC-1	22.20	
BING	BING	22.28	
DING	EL 11/107	22.29	
ECD2		22.32	
Lano	MGC22776	22.30	9n01 0
		22.39	opz 1.2
		22.00	
MCCO0916	INFROFIUD	22.69	
		00.70	
		22.79	
INFRSFIUD	TNERGEIOD	22.81	
	INFROFIUA	22.87	
	MGC29816	22.93	
		00.07	
LUXLZ		22.97	
	LIGALI	23.10	
		23.24	
MSCD	PRO1496	23.21	
		00.05	
STC1	STC1	23.35	
		23.52	
ADAIVIZO		24.03	
		24.06	
NEF3	NEF3	24.59	
INEFL		24.63	
	FLJ21034	25.07	
GNRH1	GNRH1	25.10	
	FLJ20038	25.11	

	Mapping		
JNI - 2000	UCSC Apr 03	Mbp	Band
FLJ25804	FLJ25804	25.18	
	EBF2	25.52	
PPP2R2A	PPP2R2A	26.04	
BNIP3L	BNIP3L	26.08	
PNMA2	PNMA2	26.18	
DPYSL2	DPYSL2	26.33	
ADRA1A	ADRA1A	26.43	
STMN4	STMN4	26.91	
TRIM35	TRIM35	26.96	
PTK2B	PTK2B	27.13	
	CHRNA2	27.14	8p21.1
FPHX2	EPHX2	27.22	- 1-
CLU	CLU	27.27	
ADRA1A	010	26.43	
	SCARA3	27.31	
EL 110853	EL 110853	27.01	
		27.41	
IOFK	MGC45780	27.40	
		27.50	
LLFJ	DNOC	27.03	
		28.02	
10055000	DKFZP434K1210	28.02	
PNOC	LOC55893	28.02	
	FBX16	28.11	
FZD3	FZD3	28.18	
EXTL3	EXTL3	28.43	
FLJ10871	FLJ10871	28.44	
FLJ21616	FLJ21616	28.68	
KIF13B	KIF13B	28.74	8p12
DUSP4	DUSP4	29.01	
MAPKK		gap	
MGC8721	MGC8721	29.77	
LEPROTL1	LEPROTL1	29.81	
DCTN6	DCTN6	29.89	
	RBPMS	30.09	
1D12A	1D12A	30.13	
BBPMS	10121	00.10	
GTE2E2	GTE2E2	30.31	
GSB	GSB	30.39	
D8S2298F	D8S2298E	30.48	
PPP2CB	PPP2CB	30.49	
111200	TEX15	20.55	
		20.33	
		30.74	
		30.78	
		32.47	
		33.09	
	LUC64549	33.21	
	FLJ23203	33.21	
MOOIIO	FLJ 12526	33.26	
	NIGC 1 136	33.30	
LUC84549			
UNC5D		35.50	
BC028701	BC028701	36.64	

REFERENCES

Abate-Shen C, Shen MM (2000) Molecular genetics of prostate cancer. Genes Dev 14: 2410-34.

- Abdulkadir SA, Magee JA, Peters TJ, Kaleem Z, Naughton CK, Humphrey PA, Milbrandt J (2002) Conditional loss of Nkx3.1 in adult mice induces prostatic intraepithelial neoplasia. Mol Cell Biol 22: 1495-503.
- Ahmed MN, Kim K, Haddad B, Berchuck A, Qumsiyeh MB (2000) Comparative genomic hybridization studies in hydatidiform moles and choriocarcinoma: amplification of 7q21-q31 and loss of 8p12-p21 in choriocarcinoma. Cancer Genet Cytogenet 116: 10-5
- Albertsen HM, Abderrahim H, Cann HM, Dausset J, Le Paslier D, Cohen D (1990) Construction and characterization of a yeast artificial chromosome library containing seven haploid human genome equivalents. Proc Natl Acad Sci U S A 87: 4256-60
- Alers JC, Rochat J, Krijtenburg PJ, Hop WC, Kranse R, Rosenberg C, Tanke HJ, Schroder FH, van Dekken H (2000) Identification of genetic markers for prostatic cancer progression. Lab Invest 80: 931-42
- Anbazhagan R, Fujii H, Gabrielson E (1998) Allelic loss of chromosomal arm 8p in breast cancer progression. Am J Pathol 152: 815-9
- Arbieva ZH, Banerjee K, Kim SY, Edassery SL, Maniatis VS, Horrigan SK, Westbrook CA (2000) Highresolution physical map and transcript identification of a prostate cancer deletion interval on 8p22. Genome Res 10: 244-57
- Armitage P, Doll R (1954) The age distribution of cancer and a multi-stage theory of carcinogenesis. Br J Cancer 8: 1-12.
- Baffa R, Santoro R, Bullrich F, Mandes B, Ishii H, Croce CM (2000) Definition and refinement of chromosome 8p regions of loss of heterozygosity in gastric cancer. Clin Cancer Res 6: 1372-7
- Becker SA, Zhou YZ, Slagle BL (1996) Frequent loss of chromosome 8p in hepatitis B virus-positive hepatocellular carcinomas from China. Cancer Res 56: 5092-7.
- Benhamou S, Reinikainen M, Bouchardy C, Dayer P, Hirvonen A (1998) Association between lung cancer and microsomal epoxide hydrolase genotypes. Cancer Res 58: 5291-3.
- Bergerheim US, Kunimi K, Collins VP, Ekman P (1991) Deletion mapping of chromosomes 8, 10, and 16 in human prostatic carcinoma. Genes Chromosom Cancer 3: 215-20.
- Bernardino J, Bourgeois CA, Muleris M, Dutrillaux AM, Malfoy B, Dutrillaux B (1997) Characterization of chromosome changes in two human prostatic carcinoma cell lines (PC-3 and DU145) using chromosome painting and comparative genomic hybridization. Cancer Genet Cytogenet 96: 123-8.
- Berry R, Schaid DJ, Smith JR, French AJ, Schroeder JJ, McDonnell SK, Peterson BJ, Wang ZY, Carpten JD, Roberts SG, Tester DJ, Blute ML, Trent JM, Thibodeau SN (2000a) Linkage analyses at the chromosome 1 loci 1q24-25 (HPC1), 1q42.2-43 (PCAP), and 1p36 (CAPB) in families with hereditary prostate cancer. Am J Hum Genet 66: 539-46
- Berry R, Schroeder JJ, French ÅJ, McDonnell SK, Peterson BJ, Cunningham JM, Thibodeau SN, Schaid DJ (2000b) Evidence for a prostate cancer-susceptibility locus on chromosome 20. Am J Hum Genet 67: 82-91.
- Berthon P, Valeri A, Cohen-Akenine A, Drelon E, Paiss T, Wohr G, Latil A, Millasseau P, Mellah I, Cohen N, Blanche H, Bellane-Chantelot C, Demenais F, Teillac P, Le Duc A, de Petriconi R, Hautmann R, Chumakov I, Bachner L, Maitland NJ, Lidereau R, Vogel W, Fournier G, Mangin P, Cussenot O, et al. (1998) Predisposing gene for early-onset prostate cancer, localized on chromosome 1q42.2-43. Am J Hum Genet 62: 1416-24
- Bhatia-Gaur R, Donjacour AA, Sciavolino PJ, Kim M, Desai N, Young P, Norton CR, Gridley T, Cardiff RD, Cunha GR, Abate-Shen C, Shen MM (1999) Roles for Nkx3.1 in prostate development and cancer. Genes Dev 13: 966-77
- Bishop DT, Kiemeney LA (1997) Family studies and the evidence for genetic susceptibility to prostate cancer. Semin Cancer Biol 8: 45-51
- Bishop JM (1987) The molecular genetics of cancer. Science 235: 305-11.
- Bostwick DG, Pacelli A, Lopez-Beltran A (1997) Ultrastructure of prostatic intraepithelial neoplasia. Prostate 33: 32-7.
- Bova GS, Carter BS, Bussemakers MJ, Emi M, Fujiwara Y, Kyprianou N, Jacobs SC, Robinson JC, Epstein JI, Walsh PC, et a (1993) Homozygous deletion and frequent allelic loss of chromosome 8p22 loci in human prostate cancer. Cancer Res 53: 3869-73.

- Bova GS, Macgrogan D, Levy A, Pin SS, Bookstein R, Isaacs WB (1996) Physical mapping of chromosome 8p22 markers and their homozygous deletion in a metastatic prostate cancer. Genomics 35: 46-54.
- Bowen C, Bubendorf L, Voeller HJ, Slack R, Willi N, Sauter G, Gasser TC, Koivisto P, Lack EE, Kononen J, Kallioniemi OP, Gelmann EP (2000) Loss of NKX3.1 expression in human prostate cancers correlates with tumor progression. Cancer Res 60: 6111-5.
- Boyer JC, Umar A, Risinger JI, Lipford JR, Kane M, Yin S, Barrett JC, Kolodner RD, Kunkel TA (1995) Microsatellite instability, mismatch repair deficiency, and genetic defects in human cancer cell lines. Cancer Res 55: 6063-70.
- Bratt O, Kristoffersson U, Lundgren R, Olsson H (1999) Familial and hereditary prostate cancer in southern Sweden. A population-based case-control study. Eur J Cancer 35: 272-7.
- Bratt O (2000a) Hereditary prostate cancer. BJU Int 85: 588-98.
- Bratt O (2000b) Hereditary prostate cancer. BJU Int 85: 611-3.
- Brothman AR, Watson MJ, Zhu XL, Williams BJ, Rohr LR (1994) Evaluation of 20 archival prostate tumor specimens by fluorescence in situ hybridization (FISH). Cancer Genet Cytogenet 75: 40-4
- Brothman AR, Maxwell TM, Cui J, Deubler DA, Zhu XL (1999) Chromosomal clues to the development of prostate tumors. Prostate 38: 303-12.
- Byun DS, Cho K, Ryu BK, Lee MG, Park JI, Chae KS, Kim HJ, Chi SG (2003) Frequent monoallelic deletion of PTEN and its reciprocal associatioin with PIK3CA amplification in gastric carcinoma. Int J Cancer 104: 318-27.
- Cabeza-Arvelaiz Y, Sepulveda JL, Lebovitz RM, Thompson TC, Chinault AC (2001) Functional identification of LZTS1 as a candidate prostate tumor suppressor gene on human chromosome 8p22. Oncogene 20: 4169-79.
- Cairns P, Okami K, Halachmi S, Halachmi N, Esteller M, Herman JG, Jen J, Isaacs WB, Bova GS, Sidransky D (1997) Frequent inactivation of PTEN/MMAC1 in primary prostate cancer [In Process Citation]. Cancer Res 57: 4997-5000.
- Cancel-Tassin G, Latil A, Valeri A, Mangin P, Fournier G, Berthon P, Cussenot O (2001) PCAP is the major known prostate cancer predisposing locus in families from south and west Europe. Eur J Hum Genet 9: 135-42.
- Cannizzaro LA, Bollum FJ, Huebner K, Croce CM, Cheung LC, Xu X, Hecht BK, Hecht F, Chang LM (1988) Chromosome sublocalization of a cDNA for human DNA polymerase-beta to 8p11----p12. Cytogenet Cell Genet 47: 121-4
- Cantley LC, Neel BG (1999) New insights into tumor suppression: PTEN suppresses tumor formation by restraining the phosphoinositide 3-kinase/AKT pathway. Proc Natl Acad Sci U S A 96: 4240-5
- Cappellen D, Gil Diez de Medina S, Chopin D, Thiery JP, Radvanyi F (1997) Frequent loss of heterozygosity on chromosome 10q in muscle-invasive transitional cell carcinomas of the bladder. Oncogene 14: 3059-66.
- Carpten J, Nupponen N, Isaacs S, Sood R, Robbins C, Xu J, Faruque M, Moses T, Ewing C, Gillanders E, Hu P, Bujnovszky P, Makalowska I, Baffoe-Bonnie A, Faith D, Smith J, Stephan D, Wiley K, Brownstein M, Gildea D, Kelly B, Jenkins R, Hostetter G, Matikainen M, Schleutker J, Klinger K, Connors T, Xiang Y, Wang Z, De Marzo A, Papadopoulos N, Kallioniemi OP, Burk R, Meyers D, Gronberg H, Meltzer P, Silverman R, Bailey-Wilson J, Walsh P, Isaacs W, Trent J (2002) Germline mutations in the ribonuclease L gene in families showing linkage with HPC1. Nat Genet 30: 181-4.
- Carter BS, Ewing CM, Ward WS, Treiger BF, Aalders TW, Schalken JA, Epstein JI, Isaacs WB (1990) Allelic loss of chromosomes 16q and 10q in human prostate cancer. Proc Natl Acad Sci U S A 87: 8751-5
- Carter BS, Bova GS, Beaty TH, Steinberg GD, Childs B, Isaacs WB, Walsh PC (1993) Hereditary prostate cancer: epidemiologic and clinical features. [Review]. J Urol 150: 797-802
- Carter HB, Coffey DS (1990) The prostate: an increasing medical problem. Prostate 16: 39-48.
- Catalona WJ, Avioli LV (1987) Diagnosis, staging, and surgical treatment of prostatic carcinoma. Arch Intern Med 147: 361-3.
- Chang BL, Zheng SL, Hawkins GA, Isaacs SD, Wiley KE, Turner A, Carpten JD, Bleecker ER, Walsh PC, Trent JM, Meyers DA, Isaacs WB, Xu J (2002) Joint effect of HSD3B1 and HSD3B2 genes is associated with hereditary and sporadic prostate cancer susceptibility. Cancer Res 62: 1784-9.
- Chang M, Tsuchiya K, Batchelor RH, Rabinovitch PS, Kulander BG, Haggitt RC, Burmer GC (1994) Deletion mapping of chromosome 8p in colorectal carcinoma and dysplasia arising in ulcerative colitis, prostatic carcinoma, and malignant fibrous histiocytomas. Am J Pathol 144: 1-6

Chen C, Hyytinen ER, Sun X, Helin HJ, Koivisto PA, Frierson HF, Jr., Vessella RL, Dong JT (2003) Deletion, mutation, and loss of expression of KLF6 in human prostate cancer. Am J Pathol 162: 1349-54.

Chen Y, Wang J, Fraig MM, Metcalf J, Turner WR, Bissada NK, Watson DK, Schweinfest CW (2001) Defects of DNA mismatch repair in human prostate cancer. Cancer Res 61: 4112-21.

Cher ML, Bova GS, Moore DH, Small EJ, Carroll PR, Pin SS, Epstein JI, Isaacs WB, Jensen RH (1996) Genetic alterations in untreated metastases and androgen-independent prostate cancer detected by comparative genomic hybridization and allelotyping. Cancer Res 56: 3091-102

Chumakov I, Rigault P, Guillou S, Ougen P, Billaut A, Guasconi G, Gervy P, LeGall I, Soularue P, Grinas L, et al. (1992) Continuum of overlapping clones spanning the entire human chromosome 21g [see comments]. Nature 359: 380-7

Coffey DS (1993) Prostate cancer. An overview of an increasing dilemma. [Review]. Cancer 71: 880-6

Comer FI, Parent CA (2002) PI 3-kinases and PTEN: how opposites chemoattract. Cell 109: 541-4.

- Cook WD, McCaw BJ (2000) Accommodating haploinsufficient tumor suppressor genes in Knudson's model. Oncogene 19: 3434-8.
- Cooney KA, McCarthy JD, Lange E, Huang L, Miesfeldt S, Montie JE, Oesterling JE, Sandler HM, Lange K (1997) Prostate cancer susceptibility locus on chromosome 1q: a confirmatory study [see comments]. J Natl Cancer Inst 89: 955-9

Coughlin SS, Hall IJ (2002) A review of genetic polymorphisms and prostate cancer risk. Ann Epidemiol 12: 182-96.

Chughtai SA, Crundwell MC, Cruickshank NR, Affie E, Armstrong S, Knowles MA, Takle LA, Kuo M, Khan N, Phillips SM, Neoptolemos JP, Morton DG (1999) Two novel regions of interstitial deletion on chromosome 8p in colorectal cancer. Oncogene 18: 657-65.

Cunha GR, Donjacour AA, Cooke PS, Mee S, Bigsby RM, Higgins SJ, Sugimura Y (1987) The endocrinology and developmental biology of the prostate. Endocr Rev 8: 338-62.

Cunha GR (1994) Role of mesenchymal-epithelial interactions in normal and abnormal development of the mammary gland and prostate. Cancer 74: 1030-44.

- Cunningham C, Dunlop MG, Wyllie AH, Bird CC (1993) Deletion mapping in colorectal cancer of a putative tumour suppressor gene in 8p22-p21.3. Oncogene 8: 1391-6
- Cunningham JM, Shan A, Wick MJ, McDonnell SK, Schaid DJ, Tester DJ, Qian J, Takahashi S, Jenkins RB, Bostwick DG, Thibodeau SN (1996) Allelic imbalance and microsatellite instability in prostatic adenocarcinoma. Cancer Res 56: 4475-82

Dahia PL (2000) PTEN, a unique tumor suppressor gene. Endocr Relat Cancer 7: 115-29.

- Davies MA, Kim SJ, Parikh NU, Dong Z, Bucana CD, Gallick GE (2002) Adenoviral-mediated expression of MMAC/PTEN inhibits proliferation and metastasis of human prostate cancer cells. Clin Cancer Res 8: 1904-14.
- De Marzo AM, Nelson WG, Meeker AK, Coffey DS (1998) Stem cell features of benign and malignant prostate epithelial cells. J Urol 160: 2381-92.
- Di Cristofano A, Carpino N, Dunant N, Friedland G, Kobayashi R, Strife A, Wisniewski D, Clarkson B, Pandolfi PP, Resh MD (1998a) Molecular cloning and characterization of p56dok-2 defines a new family of RasGAP-binding proteins. J Biol Chem 273: 4827-30
- Di Cristofano A, Pesce B, Cordon-Cardo C, Pandolfi PP (1998b) Pten is essential for embryonic development and tumour suppression. Nat Genet 19: 348-55.
- Di Cristofano A, De Acetis M, Koff A, Cordon-Cardo C, Pandolfi PP (2001) Pten and p27KIP1 cooperate in prostate cancer tumor suppression in the mouse. Nat Genet 27: 222-4.
- Dib A, Adelaide J, Chaffanet M, Imbert A, Le Paslier D, Jacquemier J, Gaudray P, Theillet C, Birnbaum D, Pebusque MJ (1995) Characterization of the region of the short arm of chromosome 8 amplified in breast carcinoma. Oncogene 10: 995-1001
- Dobashi Y, Shuin T, Tsuruga H, Uemura H, Torigoe S, Kubota Y (1994) DNA polymerase beta gene mutation in human prostate cancer. Cancer Res 54: 2827-9
- Dong JT, Sipe TW, Hyytinen ER, Li CL, Heise C, McClintock DE, Grant CD, Chung LW, Frierson HF, Jr. (1998) PTEN/MMAC1 is infrequently mutated in pT2 and pT3 carcinomas of the prostate. Oncogene 17: 1979-82.
- Donlon T, Wyman AR, Mulholland J, Barker D, Bruns G, Latt S, Botstein D (1986) Alpha satellite-like sequences at the centromere of chromosome 8. Am J Hum Genet 39: 196.
- Eagle LR, Yin X, Brothman AR, Williams BJ, Atkin NB, Prochownik EV (1995) Mutation of the MXI1 gene in prostate cancer. Nat Genet 9: 249-55.
- Edelhoff S, Ayer DE, Zervos AS, Steingrimsson E, Jenkins NA, Copeland NG, Eisenman RN, Brent R, Disteche CM (1994) Mapping of two genes encoding members of a distinct subfamily of MAX

interacting proteins: MAD to human chromosome 2 and mouse chromosome 6, and MXI1 to human chromosome 10 and mouse chromosome 19. Oncogene 9: 665-8.

- Egawa S, Uchida T, Suyama K, Wang C, Ohori M, Irie S, Iwamura M, Koshiba K (1995) Genomic instability of microsatellite repeats in prostate cancer: relationship to clinicopathological variables. Cancer Res 55: 2418-21.
- El Gedaily A, Bubendorf L, Willi N, Fu W, Richter J, Moch H, Mihatsch MJ, Sauter G, Gasser TC (2001) Discovery of new DNA amplification loci in prostate cancer by comparative genomic hybridization. Prostate 46: 184-90.
- El-Naggar AK, Coombes MM, Batsakis JG, Hong WK, Goepfert H, Kagan J (1998) Localization of chromosome 8p regions involved in early tumorigenesis of oral and laryngeal squamous carcinoma. Oncogene 16: 2983-7.
- Emi M, Fujiwara Y, Nakajima T, Tsuchiya E, Tsuda H, Hirohashi S, Maeda Y, Tsuruta K, Miyaki M, Nakamura Y (1992) Frequent loss of heterozygosity for loci on chromosome 8p in hepatocellular carcinoma, colorectal cancer, and lung cancer. Cancer Res 52: 5368-72
- Emi M, Fujiwara Y, Ohata H, Tsuda H, Hirohashi S, Koike M, Miyaki M, Monden M, Nakamura Y (1993) Allelic loss at chromosome band 8p21.3-p22 is associated with progression of hepatocellular carcinoma. Genes Chromosomes Cancer 7: 152-7
- Epstein CJ, Martin GM, Schultz AL, Motulsky AG (1966) Werner's syndrome a review of its symptomatology, natural history, pathologic features, genetics and relationship to the natural aging process. Medicine (Baltimore) 45: 177-221
- Eydmann ME, Knowles MA (1997) Mutation analysis of 8p genes POLB and PPP2CB in bladder cancer. Cancer Genet Cytogenet 93: 167-71.
- Fearon ER, Cho KR, Nigro JM, Kern SE, Simons JW, Ruppert JM, Hamilton SR, Preisinger AC, Thomas G, Kinzler KW, et al. (1990) Identification of a chromosome 18q gene that is altered in colorectal cancers. Science 247: 49-56
- Feilotter HE, Nagai MA, Boag AH, Eng C, Mulligan LM (1998) Analysis of PTEN and the 10q23 region in primary prostate carcinomas. Oncogene 16: 1743-8
- Fero ML, Randel E, Gurley KE, Roberts JM, Kemp CJ (1998) The murine gene p27Kip1 is haploinsufficient for tumour suppression. Nature 396: 177-80.
- Fink TM, Zimmer M, Tschopp J, Etienne J, Jenne DE, Lichter P (1993) Human clusterin (CLI) maps to 8p21 in proximity to the lipoprotein lipase (LPL) gene. Genomics 16: 526-8
- Fodde R, Smits R (2002) Cancer biology. A matter of dosage. Science 298: 761-3.
- Fujimoto M, Fults DW, Thomas GA, Nakamura Y, Heilbrun MP, White R, Story JL, Naylor SL, Kagan-Hallet KS, Sheridan PJ (1989) Loss of heterozygosity on chromosome 10 in human glioblastoma multiforme. Genomics 4: 210-4.
- Fujino T, Risinger JI, Collins NK, Liu FS, Nishii H, Takahashi H, Westphal EM, Barrett JC, Sasaki H, Kohler MF, et al. (1994) Allelotype of endometrial carcinoma. Cancer Res 54: 4294-8
- Fujiwara Y, Emi M, Ohata H, Kato Y, Nakajima T, Mori T, Nakamura Y (1993) Evidence for the presence of two tumor suppressor genes on chromosome 8p for colorectal carcinoma. Cancer Res 53: 1172-4
- Fujiwara Y, Ohata H, Emi M, Okui K, Koyama K, Tsuchiya E, Nakajima T, Monden M, Mori T, Kurimasa A, et a (1994) A 3-Mb physical map of the chromosome region 8p21.3-p22, including a 600-kb region commonly deleted in human hepatocellular carcinoma, colorectal cancer, and non-small cell lung cancer. Genes Chromosom Cancer 10: 7-14
- Fujiwara Y, Ohata H, Kuroki T, Koyama K, Tsuchiya E, Monden M, Nakamura Y (1995) Isolation of a candidate tumor suppressor gene on chromosome 8p21.3-p22 that is homologous to an extracellular domain of the PDGF receptor beta gene. Oncogene 10: 891-5
- Fujiwara Y, Hoon DS, Yamada T, Umeshita K, Gotoh M, Sakon M, Nishisho I, Monden M (2000) PTEN / MMAC1 mutation and frequent loss of heterozygosity identified in chromosome 10q in a subset of hepatocellular carcinomas. Jpn J Cancer Res 91: 287-92.
- Fukuhara H, Maruyama T, Nomura S, Oshimura M, Kitamura T, Sekiya T, Murakami Y (2001) Functional evidence for the presence of tumor suppressor gene on chromosome 10p15 in human prostate cancers. Oncogene 20: 314-9.
- Gao J, Isaacs JT (1998) Development of an androgen receptor-null model for identifying the initiation site for androgen stimulation of proliferation and suppression of programmed (apoptotic) death of PC-82 human prostate cancer cells. Cancer Res 58: 3299-306.
- Gibbs M, Chakrabarti L, Stanford JL, Goode EL, Kolb S, Schuster EF, Buckley VA, Shook M, Hood L, Jarvik GP, Ostrander EA (1999) Analysis of chromosome 1q42.2-43 in 152 families with high risk of prostate cancer. Am J Hum Genet 64: 1087-95.

Gibbs M, Stanford JL, Jarvik GP, Janer M, Badzioch M, Peters MA, Goode EL, Kolb S, Chakrabarti L, Shook M, Basom R, Ostrander EA, Hood L (2000) A genomic scan of families with prostate cancer identifies multiple regions of interest. Am J Hum Genet 67: 100-9.

Gimm O, Chi H, Dahia PL, Perren A, Hinze R, Komminoth P, Dralle H, Reynolds PR, Eng C (2001) Somatic mutation and germline variants of MINPP1, a phosphatase gene located in proximity to PTEN on 10q23.3, in follicular thyroid carcinomas. J Clin Endocrinol Metab 86: 1801-5.

- Gittes RF (1991) Carcinoma of the prostate. N Engl J Med 324: 236-45.
- Goode EL, Stanford JL, Chakrabarti L, Gibbs M, Kolb S, McIndoe RA, Buckley VA, Schuster EF, Neal CL, Miller EL, Brandzel S, Hood L, Ostrander EA, Jarvik GP (2000) Linkage analysis of 150 high-risk prostate cancer families at 1q24-25. Genet Epidemiol 18: 251-75
- Goode EL, Stanford JL, Peters MA, Janer M, Gibbs M, Kolb S, Badzioch MD, Hood L, Ostrander EA, Jarvik GP (2001) Clinical characteristics of prostate cancer in an analysis of linkage to four putative susceptibility loci. Clin Cancer Res 7: 2739-49.
- Goto M, Miller RW, Ishikawa Y, Sugano H (1996) Excess of rare cancers in Werner syndrome (adult progeria). Cancer Epidemiol Biomarkers Prev 5: 239-46
- Goss KH, Risinger MA, Kordich JJ, Sanz MM, Straughen JE, Slovek LE, Capobianco AJ, German J, Boivin GP, Groden J (2002) Enhanced tumor formation in mice heterozygous for Blm mutation. Science 297: 2051-3.
- Gray IC, Phillips SM, Lee SJ, Neoptolemos JP, Weissenbach J, Spurr NK (1995) Loss of the chromosomal region 10g23-25 in prostate cancer. Cancer Res 55: 4800-3.
- Gray IC, Štewart LM, Phillips SM, Hamilton JA, Gray NE, Watson GJ, Spurr NK, Snary D (1998) Mutation and expression analysis of the putative prostate tumour-suppressor gene PTEN. Br J Cancer 78: 1296-300.
- Gray MD, Shen JC, Kamath-Loeb AS, Blank A, Sopher BL, Martin GM, Oshima J, Loeb LA (1997) The Werner syndrome protein is a DNA helicase. Nat Genet 17: 100-3.
- Green ED, Olson MV (1990) Systematic screening of yeast artificial-chromosome libraries by use of the polymerase chain reaction. Proc Natl Acad Sci U S A 87: 1213-7.
- Greenlee RT, Murray T, Bolden S, Wingo PA (2000) Cancer statistics, 2000. CA Cancer J Clin 50: 7-33.
- Greenlee RT, Hill-Harmon MB, Murray T, Thun M (2001) Cancer statistics, 2001. CA Cancer J Clin 51: 15-36.
- Gronberg H, Smith J, Emanuelsson M, Jonsson BA, Bergh A, Carpten J, Isaacs W, Xu J, Meyers D, Trent J, Damber JE (1999) In Swedish families with hereditary prostate cancer, linkage to the HPC1 locus on chromosome 1q24-25 is restricted to families with early-onset prostate cancer. Am J Hum Genet 65: 134-40
- Gronberg H, Bergh A, Damber JE, Emanuelsson M (2000) Cancer risk in families with hereditary prostate carcinoma. Cancer 89: 1315-21.
- Gu J, Tamura M, Yamada KM (1998) Tumor suppressor PTEN inhibits integrin- and growth factormediated mitogen-activated protein (MAP) kinase signaling pathways. J Cell Biol 143: 1375-83.
- Guldberg P, thor Straten P, Birck A, Ahrenkiel V, Kirkin AF, Zeuthen J (1997) Disruption of the MMAC1/PTEN gene by deletion or mutation is a frequent event in malignant melanoma. Cancer Res 57: 3660-3.
- Gustafson CE, Wilson PJ, Lukeis R, Baker E, Woollatt E, Annab L, Hawke L, Barrett JC, Chenevixtrench G (1996) Functional evidence for a colorectal cancer tumor suppressor gene at chromosome 8p22-23 by monochromosome transfer. Cancer Res 56: 5238-5245
- Hahn SA, Schutte M, Hoque AT, Moskaluk CA, da Costa LT, Rozenblum E, Weinstein CL, Fischer A, Yeo CJ, Hruban RH, Kern SE (1996) DPC4, a candidate tumor suppressor gene at human chromosome 18q21.1. Science 271: 350-3
- Hahn WC, Weinberg RA (2002) Rules for making human tumor cells. N Engl J Med 347: 1593-603.

Hanahan D, Weinberg RA (2000) The hallmarks of cancer. Cell 100: 57-70.

- Hanley AJ, Choi BC, Holowaty EJ (1995) Cancer mortality among Chinese migrants: a review. Int J Epidemiol 24: 255-65.
- Hawkins GA, Mychaleckyj JC, Zheng SL, Faith DA, Kelly B, Isaacs SD, Wiley KE, Chang BL, Ewing CM, Bujnovszky P, Bleecker ER, Walsh PC, Meyers DA, Isaacs WB, Xu J (2002) Germline sequence variants of the LZTS1 gene are associated with prostate cancer risk. Cancer Genet Cytogenet 137: 1-7.
- He WW, Sciavolino PJ, Wing J, Augustus M, Hudson P, Meissner PS, Curtis RT, Shell BK, Bostwick DG, Tindall DJ, Gelmann EP, Abate-Shen C, Carter KC (1997) A novel human prostate-specific, androgen-regulated homeobox gene (NKX3.1) that maps to 8p21, a region frequently deleted in prostate cancer. Genomics 43: 69-77

- Herbst RA, Weiss J, Ehnis A, Cavenee WK, Arden KC (1994) Loss of heterozygosity for 10q22-10qter in malignant melanoma progression. Cancer Res 54: 3111-4.
- Hickson ID (2003) RecQ helicases: caretakers of the genome. Nat Rev Cancer 3: 169-78.
- Higuchi M, Masuyama N, Fukui Y, Suzuki A, Gotoh Y (2001) Akt mediates Rac/Cdc42-regulated cell motility in growth factor-stimulated cells and in invasive PTEN knockout cells. Curr Biol 11: 1958-62.
- Hoehn W, Schroeder FH, Reimann JF, Joebsis AC, Hermanek P (1980) Human prostatic adenocarcinoma: some characteristics of a serially transplantable line in nude mice (PC 82). Prostate 1: 95-104.
- Hoehn W, Wagner M, Riemann JF, Hermanek P, Williams E, Walther R, Schrueffer R (1984) Prostatic adenocarcinoma PC EW, a new human tumor line transplantable in nude mice. Prostate 5: 445-52.
- Hsieh CL, Oakley-Girvan I, Gallagher RP, Wu AH, Kolonel LN, Teh CZ, Halpern J, West DW, Paffenbarger RS, Jr., Whittemore AS (1997) Re: prostate cancer susceptibility locus on chromosome 1g: a confirmatory study [letter; comment]. J Natl Cancer Inst 89: 1893-4.
- Huang H, Cheville JC, Pan Y, Roche PĆ, Schmidt LJ, Tindall DJ (2001) PTEN induces chemosensitivity in PTEN-mutated prostate cancer cells by suppression of Bcl-2 expression. J Biol Chem 276: 38830-6.
- Ichikawa T, Ichikawa Y, Isaacs JT (1991) Genetic factors and metastatic potential of prostatic cancer. Cancer Surv 11: 35-42.
- Ichikawa T, Nihei N, Suzuki H, Oshimura M, Emi M, Nakamura Y, Hayata I, Isaacs JT, Shimazaki J (1994) Suppression of metastasis of rat prostatic cancer by introducing human chromosome 8. Cancer Res 54: 2299-302
- Isaacs WB (1995) Molecular genetics of prostate cancer. Cancer Surv 25: 357-79
- Ishii H, Baffa R, Numata SI, Murakumo Y, Rattan S, Inoue H, Mori M, Fidanza V, Alder H, Croce CM (1999) The FEZ1 gene at chromosome 8p22 encodes a leucine-zipper protein, and its expression is altered in multiple human tumors. Proc Natl Acad Sci U S A 96: 3928-33
- Ittmann M (1996) Allelic loss on chromosome 10 in prostate adenocarcinoma. Cancer Res 56: 2143-7
- Ittmann MM (1998) Chromosome 10 alterations in prostate adenocarcinoma (review). Oncol Rep 5: 1329-35.
- Jacobsen SJ, Katusic SK, Bergstralh EJ, Oesterling JE, Ohrt D, Klee GG, Chute CG, Lieber MM (1995) Incidence of prostate cancer diagnosis in the eras before and after serum prostate-specific antigen testing. Jama 274: 1445-9.
- Jacoby RF, Schlack S, Cole CE, Skarbek M, Harris C, Meisner LF (1997) A juvenile polyposis tumor suppressor locus at 10q22 is deleted from nonepithelial cells in the lamina propria. Gastroenterology 112: 1398-403.
- James CD, Carlbom E, Dumanski JP, Hansen M, Nordenskjold M, Collins VP, Cavenee WK (1988) Clonal genomic alterations in glioma malignancy stages. Cancer Res 48: 5546-51.
- Jenkins R, Takahashi S, DeLacey K, Bergstralh E, Lieber M (1998) Prognostic significance of allelic imbalance of chromosome arms 7q, 8p, 16q, and 18q in stage T3N0M0 prostate cancer. Genes Chromosomes Cancer 21: 131-43.
- Jourenkova-Mironova N, Mitrunen K, Bouchardy C, Dayer P, Benhamou S, Hirvonen A (2000) Highactivity microsomal epoxide hydrolase genotypes and the risk of oral, pharynx, and larynx cancers. Cancer Res 60: 534-6.
- Kagan J, Stein J, Babaian RJ, Joe YS, Pisters LL, Glassman AB, von Eschenbach AC, Troncoso P (1995) Homozygous deletions at 8p22 and 8p21 in prostate cancer implicate these regions as the sites for candidate tumor suppressor genes. Oncogene 11: 2121-6
- Kallioniemi A, Kallioniemi OP, Sudar D, Rutovitz D, Gray JW, Waldman F, Pinkel D (1992) Comparative genomic hybridization for molecular cytogenetic analysis of solid tumors. Science 258: 818-21.
- Kallioniemi A, Kallioniemi OP, Citro G, Sauter G, DeVries S, Kerschmann R, Caroll P, Waldman F (1995) Identification of gains and losses of DNA sequences in primary bladder cancer by comparative genomic hybridization. Genes Chromosomes Cancer 12: 213-9.
- Kallioniemi OP, Visakorpi T (1996) Genetic basis and clonal evolution of human prostate cancer. Adv Cancer Res 68: 225-55.
- Kamb A, Gruis NA, Weaver-Feldhaus J, Liu Q, Harshman K, Tavtigian SV, Stockert E, Day RS, 3rd, Johnson BE, Skolnick MH (1994) A cell cycle regulator potentially involved in genesis of many tumor types [see comments]. Science 264: 436-40
- Karayi MK, Neal DE, Markham AF (2000) Current status of linkage studies in hereditary prostate cancer. BJU Int 86: 659-69.

- Karlbom AE, James CD, Boethius J, Cavenee WK, Collins VP, Nordenskjold M, Larsson C (1993) Loss of heterozygosity in malignant gliomas involves at least three distinct regions on chromosome 10. Hum Genet 92: 169-74.
- Kawamata N, Park D, Wilczynski S, Yokota J, Koeffler HP (1996) Point mutations of the Mxil gene are rare in prostate cancers. Prostate 29: 191-3
- Keetch DW, Humphrey PA, Smith DS, Stahl D, Catalona WJ (1996) Clinical and pathological features of hereditary prostate cancer. J Urol 155: 1841-3
- Kerangueven F, Noguchi T, Coulier F, Allione F, Wargniez V, Simony-Lafontaine J, Longy M, Jacquemier J, Sobol H, Eisinger F, Birnbaum D (1997) Genome-wide search for loss of heterozygosity shows extensive genetic diversity of human breast carcinomas. Cancer Res 57: 5469-74
- Kim MJ, Bhatia-Gaur R, Banach-Petrosky WA, Desai N, Wang Y, Hayward SW, Cunha GR, Cardiff RD, Shen MM, Abate-Shen C (2002a) Nkx3.1 mutant mice recapitulate early stages of prostate carcinogenesis. Cancer Res 62: 2999-3004.
- Kim MJ, Cardiff RD, Desai N, Banach-Petrosky WA, Parsons R, Shen MM, Abate-Shen C (2002b) Cooperativity of Nkx3.1 and Pten loss of function in a mouse model of prostate carcinogenesis. Proc Natl Acad Sci U S A 99: 2884-9.
- Kim SK, Ro JY, Kemp BL, Lee JS, Kwon TJ, Hong WK, Mao L (1998) Identification of two distinct tumorsuppressor loci on the long arm of chromosome 10 in small cell lung cancer. Oncogene 17: 1749-53.
- Kinzler KW, Vogelstein B (1997) Cancer-susceptibility genes. Gatekeepers and caretakers. Nature 386: 761, 763.
- Kinzler KW, Vogelstein B (1998) Landscaping the cancer terrain. Science 280: 1036-7.
- Kirchhoff M, Gerdes T, Rose H, Maahr J, Ottesen AM, Lundsteen C (1998) Detection of chromosomal gains and losses in comparative genomic hybridization analysis based on standard reference intervals. Cytometry 31: 163-73
- Knudson AG, Jr. (1971) Mutation and cancer: statistical study of retinoblastoma. Proc Natl Acad Sci U S A 68: 820-3
- Komiya A, Suzuki H, Ueda T, Yatani R, Emi M, Ito H, Shimazaki J (1996) Allelic losses at loci on chromosome 10 are associated with metastasis and progression of human prostate cancer. Genes Chromosomes Cancer 17: 245-53
- Komiya A, Suzuki H, Ueda T, Aida S, Ito N, Shiraishi T, Yatani R, Emi M, Yasuda K, Shimazaki J, Ito H (1997) PRLTS gene alterations in human prostate cancer. Jpn J Cancer Res 88: 389-93
- Kon H, Sonoda Y, Kumabe T, Yoshimoto T, Sekiya T, Murakami Y (1998) Structural and functional evidence for the presence of tumor suppressor genes on the short arm of chromosome 10 in human gliomas. Oncogene 16: 257-63.
- Kotelevets L, van Hengel J, Bruyneel E, Mareel M, van Roy F, Chastre E (2001) The lipid phosphatase activity of PTEN is critical for stabilizing intercellular junctions and reverting invasiveness. J Cell Biol 155: 1129-35.
- Koyama M, Ito M, Nagai H, Emi M, Moriyama Y (1999) Inactivation of both alleles of the DPC4/SMAD4 gene in advanced colorectal cancers: identification of seven novel somatic mutations in tumors from Japanese patients. Mutat Res 406: 71-7.
- Kuczyk MA, Serth J, Bokemeyer C, Schwede J, Herrmann R, Machtens S, Grunewald V, Hofner K, Jonas U (1998) The MXI1 tumor suppressor gene is not mutated in primary prostate cancer. Oncol Rep 5: 213-6
- Kunimi K, Bergerheim US, Larsson IL, Ekman P, Collins VP (1991) Allelotyping of human prostatic adenocarcinoma. Genomics 11: 530-6
- Kurimasa A, Nagata Y, Shimizu M, Emi M, Nakamura Y, Oshimura M (1994) A human gene that restores the DNA-repair defect in SCID mice is located on 8p11.1-->q11.1. Hum Genet 93: 21-6.
- Kwabi-Addo B, Giri D, Schmidt K, Podsypanina K, Parsons R, Greenberg N, Ittmann M (2001) Haploinsufficiency of the Pten tumor suppressor gene promotes prostate cancer progression. Proc Natl Acad Sci U S A 98: 11563-8.
- Lahoz EG, Xu L, Schreiber-Agus N, DePinho RA (1994) Suppression of Myc, but not E1a, transformation activity by Max-associated proteins, Mad and Mxi1. Proc Natl Acad Sci U S A 91: 5503-7.
- Laitinen S, Karhu R, Sawyers CL, Vessella RL, Visakorpi T (2002) Chromosomal aberrations in prostate cancer xenografts detected by comparative genomic hybridization. Genes Chromosomes Cancer 35: 66-73.
- Lander ES, Linton LM, Birren B, Nusbaum C, Zody MC, Baldwin J, Devon K, Dewar K, Doyle M, FitzHugh W, Funke R, Gage D, Harris K, Heaford A, Howland J, Kann L, Lehoczky J, LeVine R,

McEwan P, McKernan K, Meldrim J, Mesirov JP, Miranda C, Morris W, Naylor J, Raymond C, Rosetti M, Santos R, Sheridan A, Sougnez C, Stange-Thomann N, Stojanovic N, Subramanian A, Wyman D, Rogers J, Sulston J, Ainscough R, Beck S, Bentley D, Burton J, Clee C, Carter N, Coulson A, Deadman R, Deloukas P, Dunham A, Dunham I, Durbin R, French L, Grafham D, Gregory S, Hubbard T, Humphray S, Hunt A, Jones M, Lloyd C, McMurray A, Matthews L, Mercer S, Milne S, Mullikin JC, Mungall A, Plumb R, Ross M, Shownkeen R, Sims S, Waterston RH, Wilson RK, Hillier LW, McPherson JD, Marra MA, Mardis ER, Fulton LA, Chinwalla AT, Pepin KH, Gish WR, Chissoe SL, Wendl MC, Delehaunty KD, Miner TL, Delehaunty A, Kramer JB, Cook LL, Fulton RS, Johnson DL, Minx PJ, Clifton SW, Hawkins T, Branscomb E, Predki P, Richardson P, Wenning S, Slezak T, Doggett N, Cheng JF, Olsen A, Lucas S, Elkin C, Uberbacher E, Frazier M, et al. (2001) Initial sequencing and analysis of the human genome. Nature 409: 860-921.

- Larsson C, White I, Johansson C, Stark A, Meijer J (1995) Localization of the human soluble epoxide hydrolase gene (EPHX2) to chromosomal region 8p21-p12. Hum Genet 95: 356-8.
- Leach FS, Velasco A, Hsieh JT, Sagalowsky AI, McConnell JD (2000) The mismatch repair gene hMSH2 is mutated in the prostate cancer cell line LNCaP. J Urol 164: 1830-3.
- Lebel M, Cardiff RD, Leder P (2001) Tumorigenic effect of nonfunctional p53 or p21 in mice mutant in the Werner syndrome helicase. Cancer Res 61: 1816-9.
- Lee C (1997) Cellular interactions in prostate cancer. Br J Urol 79: 21-7.
- Lee SH, Shin MS, Kim HS, Park WS, Kim SY, Jang JJ, Rhim KJ, Jang J, Lee HK, Park JY, Oh RR, Han SY, Lee JH, Lee JY, Yoo NJ (2000) Somatic mutations of Fas (Apo-1/CD95) gene in cutaneous squamous cell carcinoma arising from a burn scar. J Invest Dermatol 114: 122-6.
- Lengauer C, Kinzler KW, Vogelstein B (1998) Genetic instabilities in human cancers. Nature 396: 643-9.
- Lerebours F, Olschwang S, Thuille B, Schmitz A, Fouchet P, Buecher B, Martinet N, Galateau F, Thomas G (1999) Fine deletion mapping of chromosome 8p in non-small-cell lung carcinoma. Int J Cancer 81: 854-8
- Leube B, Drechsler M, Muhlmann K, Schafer R, Schulz WA, Santourlidis S, Anastasiadis A, Ackermann R, Visakorpi T, Muller W, Royer-Pokora B (2002) Refined mapping of allele loss at chromosome 10q23-26 in prostate cancer. Prostate 50: 135-44.
- Levy A, Dang UC, Bookstein R (1999) High-density screen of human tumor cell lines for homozygous deletions of loci on chromosome arm 8p. Genes Chromosomes Cancer 24: 42-7.
- Li DM, Sun H (1997) TEP1, encoded by a candidate tumor suppressor locus, is a novel protein tyrosine phosphatase regulated by transforming growth factor beta. Cancer Res 57: 2124-9.
- Li J, Yen C, Liaw D, Podsypanina K, Bose S, Wang SI, Puc J, Miliaresis C, Rodgers L, McCombie R, Bigner SH, Giovanella BC, Ittmann M, Tycko B, Hibshoosh H, Wigler MH, Parsons R (1997) PTEN, a putative protein tyrosine phosphatase gene mutated in human brain, breast, and prostate cancer. Science 275: 1943-1947.
- Liaw D, Marsh DJ, Li J, Dahia PL, Wang SI, Zheng Z, Bose S, Call KM, Tsou HC, Peacocke M, Eng C, Parsons R (1997) Germline mutations of the PTEN gene in Cowden disease, an inherited breast and thyroid cancer syndrome. Nat Genet 16: 64-7.
- Lilja H (1985) A kallikrein-like serine protease in prostatic fluid cleaves the predominant seminal vesicle protein. J Clin Invest 76: 1899-903.
- Lilja H, Abrahamsson PA (1988) Three predominant proteins secreted by the human prostate gland. Prostate 12: 29-38.
- Lisitsyn N, Wigler M (1993) Cloning the differences between two complex genomes. Science 259: 946-51
- Lombard DB, Beard C, Johnson B, Marciniak RA, Dausman J, Bronson R, Buhlmann JE, Lipman R, Curry R, Sharpe A, Jaenisch R, Guarente L (2000) Mutations in the WRN gene in mice accelerate mortality in a p53-null background. Mol Cell Biol 20: 3286-91.
- Lundwall A, Lilja H (1987) Molecular cloning of human prostate specific antigen cDNA. FEBS Lett 214: 317-22.
- MacGrogan D, Levy A, Bostwick D, Wagner M, Wells D, Bookstein R (1994) Loss of chromosome arm 8p loci in prostate cancer: mapping by quantitative allelic imbalance. Genes Chromosom Cancer 10: 151-9
- Macgrogan D, Levy A, Bova GS, Isaacs WB, Bookstein R (1996) Structure and methylation-associated silencing of a gene within a homozygously deleted region of human chromosome band 8p22. Genomics 35: 55-65
- MacGrogan D, Bookstein R (1997) Tumour suppressor genes in prostate cancer. Semin Cancer Biol 8: 11-9.

- Macoska JA, Trybus TM, Sakr WA, Wolf MC, Benson PD, Powell IJ, Pontes JE (1994) Fluorescence in situ hybridization analysis of 8p allelic loss and chromosome 8 instability in human prostate cancer. Cancer Res 54: 3824-30
- Macoska JA, Trybus TM, Benson PD, Sakr WA, Grignon DJ, Wojno KD, Pietruk T, Powell IJ (1995) Evidence for three tumor suppressor gene loci on chromosome 8p in human prostate cancer. Cancer Res 55: 5390-5
- Macoska JA, Beheshti B, Rhim JS, Hukku B, Lehr J, Pienta KJ, Squire JA (2000) Genetic characterization of immortalized human prostate epithelial cell cultures. Evidence for structural rearrangements of chromosome 8 and i(8q) chromosome formation in primary tumor-derived cells. Cancer Genet Cytogenet 120: 50-7
- Maehama T, Dixon JE (1998) The tumor suppressor, PTEN/MMAC1, dephosphorylates the lipid second messenger, phosphatidylinositol 3,4,5-trisphosphate. J Biol Chem 273: 13375-8.
- Marker PC, Donjacour AA, Dahiya R, Cunha GR (2003) Hormonal, cellular, and molecular control of prostatic development. Dev Biol 253: 165-74.
- Matsuyama H, Pan Y, Skoog L, Tribukait B, Naito K, Ekman P, Lichter P, Bergerheim US (1994) Deletion mapping of chromosome 8p in prostate cancer by fluorescence in situ hybridization. Oncogene 9: 3071-6
- Matsuyama H, Pan Y, Oba K, Yoshihiro S, Matsuda K, Hagarth L, Kudren D, Naito K, Bergerheim US, Ekman P (2001) Deletions on chromosome 8p22 may predict disease progression as well as pathological staging in prostate cancer. Clin Cancer Res 7: 3139-43.
- Matzkin H, Patel JP, Altwein JE, Soloway MS (1994) Stage T1A carcinoma of prostate. Urology 43: 11-21.
- McMenamin ME, Soung P, Perera S, Kaplan I, Loda M, Sellers WR (1999) Loss of PTEN expression in paraffin-embedded primary prostate cancer correlates with high Gleason score and advanced stage. Cancer Res 59: 4291-6.
- McNeal JE (1997) Anatomy and normal histology of the human prostate. In Pathology of the prostate (ed. Foster CS, Bostwick DG): 19-34. W.B. Sanders Company, Philadelphia London Toronto Montreal Sydney Tokyo.
- Meijer J, DePierre JW (1988) Cytosolic epoxide hydrolase. Chem Biol Interact 64: 207-49
- Meisslitzer C, Ruppitsch W, Weirich-Schwaiger H, Weirich HG, Jabkowsky J, Klein G, Schweiger M, Hirsch-Kauffmann M (1997) Werner syndrome: characterization of mutations in the WRN gene in an affected family. Eur J Hum Genet 5: 364-70
- Merrill RM, Weed DL, Feuer EJ (1997) The lifetime risk of developing prostate cancer in white and black men. Cancer Epidemiol Biomarkers Prev 6: 763-8.
- Miyake H, Hara I, Kamidono S, Gleave ME, Eto H (2003) Resistance to cytotoxic chemotherapyinduced apoptosis in human prostate cancer cells is associated with intracellular clusterin expression. Oncol Rep 10: 469-73.
- Mollenhauer J, Wiemann S, Scheurlen W, Korn B, Hayashi Y, Wilgenbus KK, von Deimling A, Poustka A (1997) DMBT1, a new member of the SRCR superfamily, on chromosome 10q25.3-26.1 is deleted in malignant brain tumours. Nat Genet 17: 32-9.
- Mollenhauer J, Holmskov U, Wiemann S, Krebs I, Herbertz S, Madsen J, Kioschis P, Coy JF, Poustka A (1999) The genomic structure of the DMBT1 gene: evidence for a region with susceptibility to genomic instability. Oncogene 18: 6233-40.
- Mollenhauer J, Herbertz S, Holmskov U, Tolnay M, Krebs I, Merlo A, Schroder HD, Maier D, Breitling F, Wiemann S, Grone HJ, Poustka A (2000) DMBT1 encodes a protein involved in the immune defense and in epithelial differentiation and is highly unstable in cancer. Cancer Res 60: 1704-10.
- Mori M, Shiraishi T, Tanaka S, Yamagata M, Mafune K, Tanaka Y, Ueo H, Barnard GF, Sugimachi K (1999) Lack of DMBT1 expression in oesophageal, gastric and colon cancers. Br J Cancer 79: 211-3
- Morita R, Saito S, Ishikawa J, Ogawa O, Yoshida O, Yamakawa K, Nakamura Y (1991) Common regions of deletion on chromosomes 5q, 6q, and 10q in renal cell carcinoma. Cancer Res 51: 5817-20.
- Moser MJ, Bigbee WL, Grant SG, Emond MJ, Langlois RG, Jensen RH, Oshima J, Monnat RJ, Jr. (2000a) Genetic instability and hematologic disease risk in Werner syndrome patients and heterozygotes. Cancer Res 60: 2492-6.
- Moser MJ, Kamath-Loeb AS, Jacob JE, Bennett SE, Oshima J, Monnat RJ, Jr. (2000b) WRN helicase expression in Werner syndrome cell lines. Nucleic Acids Res 28: 648-54.
- Mueller W, Mollenhauer J, Stockhammer F, Poustka A, von Deimling A (2002) Rare mutations of the DMBT1 gene in human astrocytic gliomas. Oncogene 21: 5956-9.

- Murakami YS, Albertsen H, Brothman AR, Leach RJ, White RL (1996) Suppression of the malignant phenotype of human prostate cancer cell line PPC-1 by introduction of normal fragments of human chromosome 10. Cancer Res 56: 2157-60
- Myers MP, Stolarov JP, Eng C, Li J, Wang SI, Wigler MH, Parsons R, Tonks NK (1997) P-TEN, the tumor suppressor from human chromosome 10q23, is a dual- specificity phosphatase. Proc Natl Acad Sci U S A 94: 9052-7.
- Myers MP, Pass I, Batty IH, Van der Kaay J, Stolarov JP, Hemmings BA, Wigler MH, Downes CP, Tonks NK (1998) The lipid phosphatase activity of PTEN is critical for its tumor supressor function. Proc Natl Acad Sci U S A 95: 13513-8.
- Nagase S, Sato S, Tezuka F, Wada Y, Yajima A, Horii A (1996) Deletion mapping on chromosome 10q25-q26 in human endometrial cancer. Br J Cancer 74: 1979-83.
- Nagase S, Yamakawa H, Sato S, Yajima A, Horii A (1997) Identification of a 790-kilobase region of common allelic loss in chromosome 10q25-q26 in human endometrial cancer. Cancer Res 57: 1630-3.
- Narla G, Heath KE, Reeves HL, Li D, Giono LE, Kimmelman AC, Glucksman MJ, Narla J, Eng FJ, Chan AM, Ferrari AC, Martignetti JA, Friedman SL (2001) KLF6, a candidate tumor suppressor gene mutated in prostate cancer. Science 294: 2563-6.
- Narod S (1999) Genetic epidemiology of prostate cancer. Biochim Biophys Acta 1423: F1-13.
- Nelen MR, van Staveren WC, Peeters EA, Hassel MB, Gorlin RJ, Hamm H, Lindboe CF, Fryns JP, Sijmons RH, Woods DG, Mariman EC, Padberg GW, Kremer H (1997) Germline mutations in the PTEN/MMAC1 gene in patients with Cowden disease. Hum Mol Genet 6: 1383-7.
- Neuhausen SL, Farnham JM, Kort E, Tavtigian SV, Skolnick MH, Cannon-Albright LA (1999) Prostate cancer susceptibility locus HPC1 in Utah high-risk pedigrees. Hum Mol Genet 8: 2437-42.
- NIH/CEPH Collaborative Mapping Group (1992) A comprehensive genetic linkage map of the human genome. Science 258: 67-86.
- Nihei N, Ichikawa T, Kawana Y, Kuramochi H, Kugo H, Oshimura M, Killary AM, Rinker-Schaeffer CW, Barrett JC, Isaacs JT, et al. (1995) Localization of metastasis suppressor gene(s) for rat prostatic cancer to the long arm of human chromosome 10. Genes Chromosomes Cancer 14: 112-9
- Nihei N, Ichikawa T, Kawana Y, Kuramochi H, Kugoh H, Oshimura M, Hayata I, Shimazaki J, Ito H (1996) Mapping of metastasis suppressor gene(s) for rat prostate cancer on the short arm of human chromosome 8 by irradiated microcell-mediated chromosome transfer. Gene Chromosome Cancer 17: 260-268
- Nihei N, Kouprina N, Larionov V, Oshima J, Martin GM, Ichikawa T, Barrett JC (2002) Functional evidence for a metastasis suppressor gene for rat prostate cancer within a 60-kilobase region on human chromosome 8p21-p12. Cancer Res 62: 367-70.
- Nishizaki T, DeVries S, Chew K, Goodson WH, 3rd, Ljung BM, Thor A, Waldman FM (1997) Genetic alterations in primary breast cancers and their metastases: direct comparison using modified comparative genomic hybridization. Genes Chromosomes Cancer 19: 267-72
- Noordzij MA, van Weerden WM, de Ridder CM, van der Kwast TH, Schroder FH, van Steenbrugge GJ (1996) Neuroendocrine differentiation in human prostatic tumor models. Am J Pathol 149: 859-71.
- Nupponen NN, Hyytinen ER, Kallioniemi AH, Visakorpi T (1998a) Genetic alterations in prostate cancer cell lines detected by comparative genomic hybridization. Cancer Genet Cytogenet 101: 53-7.
- Nupponen NN, Kakkola L, Koivisto P, Visakorpi T (1998b) Genetic alterations in hormone-refractory recurrent prostate carcinomas. Am J Pathol 153: 141-8
- Nwosu V, Carpten J, Trent JM, Sheridan R (2001) Heterogeneity of genetic alterations in prostate cancer: evidence of the complex nature of the disease. Hum Mol Genet 10: 2313-8.
- Oba K, Matsuyama H, Yoshihiro S, Kishi F, Takahashi M, Tsukamoto M, Kinjo M, Sagiyama K, Naito K (2001) Two putative tumor suppressor genes on chromosome arm 8p may play different roles in prostate cancer. Cancer Genet Cytogenet 124: 20-6.
- Ohata H, Emi M, Fujiwara Y, Higashino K, Nakagawa K, Futagami R, Tsuchiya E, Nakamura Y (1993) Deletion mapping of the short arm of chromosome 8 in non-small cell lung carcinoma. Genes Chromosomes Cancer 7: 85-8
- Ohori M, Scardino PT (1994) Early detection of prostate cancer: the nature of cancers detected with current diagnostic tests. Semin Oncol 21: 522-6.
- Ornstein DK, Cinquanta M, Weiler S, Duray PH, Emmert-Buck MR, Vocke CD, Linehan WM, Ferretti JA (2001) Expression studies and mutational analysis of the androgen regulated homeobox gene NKX3.1 in benign and malignant prostate epithelium. J Urol 165: 1329-34.

- Ozen M, Hopwood VL, Balbay MD, Johnston DA, Babaian RJ, Logothetis CJ, von Eschenbach AC, Pathak S (2000) Correlation of non-random chromosomal aberrations in lymphocytes of prostate cancer patients with specific clinical parameters. Int J Oncol 17: 113-7.
- Pan Y, Kytola S, Farnebo F, Wang N, Lui WO, Nupponen N, Isola J, Visakorpi T, Bergerheim US, Larsson C (1999) Characterization of chromosomal abnormalities in prostate cancer cell lines by spectral karyotyping. Cytogenet Cell Genet 87: 225-32.
- Pan Y, Lui WO, Nupponen N, Larsson C, Isola J, Visakorpi T, Bergerheim US, Kytola S (2001) 5q11, 8p11, and 10q22 are recurrent chromosomal breakpoints in prostate cancer cell lines. Genes Chromosomes Cancer 30: 187-95.
- Paris PL, Witte JS, Kupelian PA, Levin H, Klein EA, Catalona WJ, Casey G (2000) Identification and fine mapping of a region showing a high frequency of allelic imbalance on chromosome 16q23.2 that corresponds to a prostate cancer susceptibility locus. Cancer Res 60: 3645-9
- Parker SL, Tong T, Bolden S, Wingo PA (1997) Cancer statistics, 1997. CA Cancer J Clin 47: 5-27.
- Peiffer SL, Herzog TJ, Tribune DJ, Mutch DG, Gersell DJ, Goodfellow PJ (1995) Allelic loss of sequences from the long arm of chromosome 10 and replication errors in endometrial cancers. Cancer Res 55: 1922-6.
- Persad S, Attwell S, Gray V, Delcommenne M, Troussard A, Sanghera J, Dedhar S (2000) Inhibition of integrin-linked kinase (ILK) suppresses activation of protein kinase B/Akt and induces cell cycle arrest and apoptosis of PTEN-mutant prostate cancer cells. Proc Natl Acad Sci U S A 97: 3207-12.
- Pesche S, Latil A, Muzeau F, Cussenot O, Fournier G, Longy M, Eng C, Lidereau R (1998) PTEN/MMAC1/TEP1 involvement in primary prostate cancers. Oncogene 16: 2879-83.
- Peters MA, Janer M, Kolb S, Jarvik GP, Ostrander EA, Stanford JL (2001a) Germline mutations in the p73 gene do not predispose to familial prostate-brain cancer. Prostate 48: 292-6.
- Peters MA, Jarvik GP, Janer M, Chakrabarti L, Kolb S, Goode EL, Gibbs M, DuBois CC, Schuster EF, Hood L, Ostrander EA, Stanford JL (2001b) Genetic linkage analysis of prostate cancer families to Xq27-28. Hum Hered 51: 107-13.
- Pineau P, Nagai H, Prigent S, Wei Y, Gyapay G, Weissenbach J, Tiollais P, Buendia MA, Dejean A (1999) Identification of three distinct regions of allelic deletions on the short arm of chromosome 8 in hepatocellular carcinoma. Oncogene 18: 3127-34
- Pinkel D, Straume T, Gray JW (1986) Cytogenetic analysis using quantitative, high-sensitivity, fluorescence hybridization. Proc Natl Acad Sci U S A 83: 2934-8.
- Pinkel D, Segraves R, Sudar D, Clark S, Poole I, Kowbel D, Collins C, Kuo WL, Chen C, Zhai Y, Dairkee SH, Ljung BM, Gray JW, Albertson DG (1998) High resolution analysis of DNA copy number variation using comparative genomic hybridization to microarrays. Nat Genet 20: 207-11.
- Podsypanina K, Ellenson LH, Nemes A, Gu J, Tamura M, Yamada KM, Cordon-Cardo C, Catoretti G, Fisher PE, Parsons R (1999) Mutation of Pten/Mmac1 in mice causes neoplasia in multiple organ systems. Proc Natl Acad Sci U S A 96: 1563-8.
- Post PN, Kil PJ, Crommelin MA, Schapers RF, Coebergh JW (1998) Trends in incidence and mortality rates for prostate cancer before and after prostate-specific antigen introduction. A registry-based study in southeastern Netherlands, 1971-1995. Eur J Cancer 34: 705-9.
- Potosky AL, Miller BA, Albertsen PC, Kramer BS (1995) The role of increasing detection in the rising incidence of prostate cancer. Jama 273: 548-52.
- Prasad MA, Trybus TM, Wojno KJ, Macoska JA (1998) Homozygous and frequent deletion of proximal 8p sequences in human prostate cancers: identification of a potential tumor suppressor gene site [In Process Citation]. Genes Chromosomes Cancer 23: 255-62
- Prochownik EV, Eagle Grove L, Deubler D, Zhu XL, Stephenson RA, Rohr LR, Yin X, Brothman AR (1998) Commonly occurring loss and mutation of the MXI1 gene in prostate cancer. Genes Chromosomes Cancer 22: 295-304
- Qian J, Jenkins RB, Bostwick DG (1998) Determination of gene and chromosome dosage in prostatic intraepithelial neoplasia and carcinoma. Anal Quant Cytol Histol 20: 373-80.
- Quon KC, Berns A (2001) Haplo-insufficiency? Let me count the ways. Genes Dev 15: 2917-21.
- Rasheed BK, McLendon RE, Friedman HS, Friedman AH, Fuchs HE, Bigner DD, Bigner SH (1995) Chromosome 10 deletion mapping in human gliomas: a common deletion region in 10q25. Oncogene 10: 2243-6.
- Rasheed BK, Stenzel TT, McLendon RE, Parsons R, Friedman AH, Friedman HS, Bigner DD, Bigner SH (1997) PTEN gene mutations are seen in high-grade but not in low-grade gliomas. Cancer Res 57: 4187-90

- Reifenberger J, Wolter M, Bostrom J, Buschges R, Schulte KW, Megahed M, Ruzicka T, Reifenberger G (2000) Allelic losses on chromosome arm 10q and mutation of the PTEN (MMAC1) tumour suppressor gene in primary and metastatic malignant melanomas. Virchows Arch 436: 487-93.
- Rebbeck TR, Walker AH, Zeigler-Johnson C, Weisburg S, Martin AM, Nathanson KL, Wein AJ, Malkowicz SB (2000) Association of HPC2/ELAC2 genotypes and prostate cancer. Am J Hum Genet 67: 1014-9.
- Reiter RE, Sato I, Thomas G, Qian J, Gu Z, Watabe T, Loda M, Jenkins RB (2000) Coamplification of prostate stem cell antigen (PSCA) and MYC in locally advanced prostate cancer. Genes Chromosomes Cancer 27: 95-103.
- Rempel SA, Schwechheimer K, Davis RL, Cavenee WK, Rosenblum ML (1993) Loss of heterozygosity for loci on chromosome 10 is associated with morphologically malignant meningioma progression. Cancer Res 53: 2386-92.
- Rhei E, Kang L, Bogomolniy F, Federici MG, Borgen PI, Boyd J (1997) Mutation analysis of the putative tumor suppressor gene PTEN/MMAC1 in primary breast carcinomas. Cancer Res 57: 3657-9.
- Risinger JI, Hayes AK, Berchuck A, Barrett JC (1997) PTEN/MMAC1 mutations in endometrial cancers. Cancer Res 57: 4736-8
- Rohrbach H, Haas CJ, Baretton GB, Hirschmann A, Diebold J, Behrendt RP, Lohrs U (1999) Microsatellite instability and loss of heterozygosity in prostatic carcinomas: comparison of primary tumors, and of corresponding recurrences after androgen-deprivation therapy and lymph-node metastases. Prostate 40: 20-7
- Rokman A, Ikonen T, Mononen N, Autio V, Matikainen MP, Koivisto PA, Tammela TL, Kallioniemi OP, Schleutker J (2001) ELAC2/HPC2 involvement in hereditary and sporadic prostate cancer. Cancer Res 61: 6038-41.
- Roylance R, Spurr N, Sheer D (1997) The genetic analysis of prostate carcinoma. Semin Cancer Biol 8: 37-44.
- Saito S, Iida A, Sekine A, Eguchi C, Miura Y, Nakamura Y (2001) Seventy genetic variations in human microsomal and soluble epoxide hydrolase genes (EPHX1 and EPHX2) in the Japanese population. J Hum Genet 46: 325-9.
- Sambrook J, Fritsch EF, Maniatis T (1989) Molecular Cloning: A Laboratory Manual, 2nd ed. Cold Spring Harbor Laboratory Press, Cold Spring Harbor NY.
- Sambrook J and Russell DW (2001) Molecular Cloning: A Laboratory Manual, 3rd ed. Cold Spring Harbor Laboratory Press, Cold Spring Harbor NY.
- Sanchez Y, Lovell M, Marin MC, Wong PE, Wolf-Ledbetter ME, McDonnell TJ, Killary AM (1996) Tumor suppression and apoptosis of human prostate carcinoma mediated by a genetic locus within human chromosome 10pter-q11. Proc Natl Acad Sci U S A 93: 2551-6.
- Sandberg M, Meijer J (1996) Structural characterization of the human soluble epoxide hydrolase gene (EPHX2). Biochem Biophys Res Commun 221: 333-9.
- Sandberg M, Hassett C, Adman ET, Meijer J, Omiecinski CJ (2000) Identification and functional characterization of human soluble epoxide hydrolase genetic polymorphisms. J Biol Chem 275: 28873-81.
- Santourlidis S, Warskulat U, Florl AR, Maas S, Pulte T, Fischer J, Muller W, Schulz WA (2001) Hypermethylation of the tumor necrosis factor receptor superfamily 6 (APT1, Fas, CD95/Apo-1) gene promoter at rel/nuclear factor kappaB sites in prostatic carcinoma. Mol Carcinog 32: 36-43.
- Saramaki O, Willi N, Bratt O, Gasser TC, Koivisto P, Nupponen NN, Bubendorf L, Visakorpi T (2001) Amplification of EIF3S3 gene is associated with advanced stage in prostate cancer. Am J Pathol 159: 2089-94.
- Saric T, Brkanac Z, Troyer DA, Padalecki SS, Sarosdy M, Williams K, Abadesco L, Leach RJ, O'Connell P (1999) Genetic pattern of prostate cancer progression. Int J Cancer 81: 219-24
- Sato K, Qian J, Slezak JM, Lieber MM, Bostwick DG, Bergstralh EJ, Jenkins RB (1999) Clinical significance of alterations of chromosome 8 in high-grade, advanced, nonmetastatic prostate carcinoma. J Natl Cancer Inst 91: 1574-80.
- Scardino PT, Weaver R, Hudson MA (1992) Early detection of prostate cancer. Hum Pathol 23: 211-22.
- Schleutker J, Matikainen M, Smith J, Koivisto P, Baffoe-Bonnie A, Kainu T, Gillanders E, Sankila R, Pukkala E, Carpten J, Stephan D, Tammela T, Brownstein M, Bailey-Wilson J, Trent J, Kallioniemi OP (2000) A genetic epidemiological study of hereditary prostate cancer (HPC) in Finland: frequent HPCX linkage in families with late-onset disease. Clin Cancer Res 6: 4810-5.
- Schoenberg M, Cairns P, Brooks JD, Marshall FF, Epstein JI, Isaacs WB, Sidransky D (1995) Frequent loss of chromosome arms 8p and 13q in collecting duct carcinoma (CDC) of the kidney. Genes Chromosomes Cancer 12: 76-80

- Schrock E, du Manoir S, Veldman T, Schoell B, Wienberg J, Ferguson-Smith MA, Ning Y, Ledbetter DH, Bar-Am I, Soenksen D, Garini Y, Ried T (1996) Multicolor spectral karyotyping of human chromosomes [see comments]. Science 273: 494-7
- Schroder FH, Hermanek P, Denis L, Fair WR, Gospodarowicz MK, Pavone-Macaluso M (1992) The TNM classification of prostate cancer. Prostate Suppl 4: 129-38.
- Schullerus D, von Knobloch R, Chudek J, Herbers J, Kovacs G (1999) Microsatellite analysis reveals deletion of a large region at chromosome 8p in conventional renal cell carcinoma. Int J Cancer 80: 22-4.
- Schutte M, da Costa LT, Hahn SA, Moskaluk C, Hoque AT, Rozenblum E, Weinstein CL, Bittner M, Meltzer PS, Trent JM, et al. (1995) Identification by representational difference analysis of a homozygous deletion in pancreatic carcinoma that lies within the BRCA2 region. Proc Natl Acad Sci U S A 92: 5950-4.
- Sciavolino PJ, Abrams EW, Yang L, Austenberg LP, Shen MM, Abate-Shen C (1997) Tissue-specific expression of murine Nkx3.1 in the male urogenital system. Dev Dyn 209: 127-38.
- Shapiro DN, Valentine V, Eagle L, Yin X, Morris SW, Prochownik EV (1994) Assignment of the human MAD and MXI1 genes to chromosomes 2p12-p13 and 10q24-q25. Genomics 23: 282-5.
- Skobe M, Fusenig NE (1998) Tumorigenic conversion of immortal human keratinocytes through stromal cell activation. Proc Natl Acad Sci U S A 95: 1050-5.
- Smith JR, Freije D, Carpten JD, Gronberg H, Xu JF, Isaacs SD, Brownstein MJ, Bova GS, Guo H, Bujnovszky P, Nusskern DR, Damber JE, Bergh A, Emanuelsson M, Kallioniemi OP, Walkerdaniels J, Baileywilson JE, Beaty TH, Meyers DA, Walsh PC, Collins FS, Trent JM, Isaacs WB (1996) Major susceptibility locus for prostate cancer on chromosome 1 suggested by a genome-wide search. Science 274: 1371-1374.
- Smits R, Ruiz P, Diaz-Cano S, Luz A, Jagmohan-Changur S, Breukel C, Birchmeier C, Birchmeier W, Fodde R (2000) E-cadherin and adenomatous polyposis coli mutations are synergistic in intestinal tumor initiation in mice. Gastroenterology 119: 1045-53.
- Somerville RP, Shoshan Y, Eng C, Barnett G, Miller D, Cowell JK (1998) Molecular analysis of two putative tumour suppressor genes, PTEN and DMBT, which have been implicated in glioblastoma multiforme disease progression. Oncogene 17: 1755-7
- Speaks SL, Sanger WG, Masih AS, Harrington DS, Hess M, Armitage JO (1992) Recurrent abnormalities of chromosome bands 10q23-25 in non-Hodgkin's lymphoma. Genes Chromosomes Cancer 5: 239-43.
- Speicher MR, Gwyn Ballard S, Ward DC (1996) Karyotyping human chromosomes by combinatorial multifluor FISH. Nat Genet 12: 368-75
- Srivastava M, Bubendorf L, Srikantan V, Fossom L, Nolan L, Glasman M, Leighton X, Fehrle W, Pittaluga S, Raffeld M, Koivisto P, Willi N, Gasser TC, Kononen J, Sauter G, Kallioniemi OP, Srivastava S, Pollard HB (2001) ANX7, a candidate tumor suppressor gene for prostate cancer. Proc Natl Acad Sci U S A 98: 4575-80.
- Steck PA, Pershouse MA, Jasser SA, Yung WK, Lin H, Ligon AH, Langford LA, Baumgard ML, Hattier T, Davis T, Frye C, Hu R, Swedlund B, Teng DH, Tavtigian SV (1997) Identification of a candidate tumour suppressor gene, MMAC1, at chromosome 10q23.3 that is mutated in multiple advanced cancers. Nat Genet 15: 356-62
- Straus SE, Jaffe ES, Puck JM, Dale JK, Elkon KB, Rosen-Wolff A, Peters AM, Sneller MC, Hallahan CW, Wang J, Fischer RE, Jackson CM, Lin AY, Baumler C, Siegert E, Marx A, Vaishnaw AK, Grodzicky T, Fleisher TA, Lenardo MJ (2001) The development of lymphomas in families with autoimmune lymphoproliferative syndrome with germline Fas mutations and defective lymphocyte apoptosis. Blood 98: 194-200.
- Suarez BK, Lin J, Witte JS, Conti DV, Resnick MI, Klein EA, Burmester JK, Vaske DA, Banerjee TK, Catalona WJ (2000) Replication linkage study for prostate cancer susceptibility genes. Prostate 45: 106-14.
- Suarez BK, Gerhard DS, Lin J, Haberer B, Nguyen L, Kesterson NK, Catalona WJ (2001) Polymorphisms in the prostate cancer susceptibility gene HPC2/ELAC2 in multiplex families and healthy controls. Cancer Res 61: 4982-4.
- Suzuki A, de la Pompa JL, Stambolic V, Elia AJ, Sasaki T, del Barco Barrantes I, Ho A, Wakeham A, Itie A, Khoo W, Fukumoto M, Mak TW (1998a) High cancer susceptibility and embryonic lethality associated with mutation of the PTEN tumor suppressor gene in mice. Curr Biol 8: 1169-78.
- Suzuki H, Emi M, Komiya A, Fujiwara Y, Yatani R, Nakamura Y, Shimazaki J (1995) Localization of a tumor suppressor gene associated with progression of human prostate cancer within a 1.2 Mb region of 8p22-p21.3. Genes Chromosomes Cancer 13: 168-74

- Suzuki H, Freije D, Nusskern DR, Okami K, Cairns P, Sidransky D, Isaacs WB, Bova GS (1998b) Interfocal heterogeneity of PTEN/MMAC1 gene alterations in multiple metastatic prostate cancer tissues. Cancer Res 58: 204-9
- Suzuki N, Shiratori M, Goto M, Furuichi Y (1999) Werner syndrome helicase contains a 5'-->3' exonuclease activity that digests DNA and RNA strands in DNA/DNA and RNA/DNA duplexes dependent on unwinding. Nucleic Acids Res 27: 2361-8.
- Takayama H, Takakuwa T, Dong Z, Nonomura N, Okuyama A, Nagata S, Aozasa K (2001) Fas gene mutations in prostatic intraepithelial neoplasia and concurrent carcinoma: analysis of laser capture microdissected specimens. Lab Invest 81: 283-8.
- Takeshita H, Sato M, Shiwaku HO, Semba S, Sakurada A, Hoshi M, Hayashi Y, Tagawa Y, Ayabe H, Horii A (1999) Expression of the DMBT1 gene is frequently suppressed in human lung cancer. Jpn J Cancer Res 90: 903-8
- Takle LA, Knowles MA (1996) Deletion mapping implicates two tumor suppressor genes on chromosome 8p in the development of bladder cancer. Oncogene 12: 1083-7
- Tamura M, Gu J, Matsumoto K, Aota S, Parsons R, Yamada KM (1998) Inhibition of cell migration, spreading, and focal adhesions by tumor suppressor PTEN. Science 280: 1614-7.
- Tashiro H, Blazes MS, Wu R, Cho KR, Bose S, Wang SI, Li J, Parsons R, Ellenson LH (1997) Mutations in PTEN are frequent in endometrial carcinoma but rare in other common gynecological malignancies. Cancer Res 57: 3935-40.
- Tanaka K, Kikuchi-Yanoshita R, Muraoka M, Konishi M, Oshimura M, Miyaki M (1996) Suppression of tumorigenicity and invasiveness of colon carcinoma cells by introduction of normal chromosome 8p12-pter. Oncogene 12: 405-10.
- Tavtigian SV, Simard J, Teng DH, Abtin V, Baumgard M, Beck A, Camp NJ, Carillo AR, Chen Y, Dayananth P, Desrochers M, Dumont M, Farnham JM, Frank D, Frye C, Ghaffari S, Gupte JS, Hu R, Iliev D, Janecki T, Kort EN, Laity KE, Leavitt A, Leblanc G, McArthur-Morrison J, Pederson A, Penn B, Peterson KT, Reid JE, Richards S, Schroeder M, Smith R, Snyder SC, Swedlund B, Swensen J, Thomas A, Tranchant M, Woodland AM, Labrie F, Skolnick MH, Neuhausen S, Rommens J, Cannon-Albright LA (2001) A candidate prostate cancer susceptibility gene at chromosome 17p. Nat Genet 27: 172-80.
- Teng DH, Hu R, Lin H, Davis T, Iliev D, Frye C, Swedlund B, Hansen KL, Vinson VL, Gumpper KL, Ellis L, El-Naggar A, Frazier M, Jasser S, Langford LA, Lee J, Mills GB, Pershouse MA, Pollack RE, Tornos C, Troncoso P, Yung WK, Fujii G, Berson A, Steck PA, et al. (1997) MMAC1/PTEN mutations in primary tumor specimens and tumor cell lines. Cancer Res 57: 5221-5.
- Trapman J, Sleddens H, Vanderweiden MM, Dinjens W, Konig JJ, Schroder FH, Faber PW, Bosman FT (1994) Loss of heterozygosity of chromosome 8 microsatellite loci implicates a candidate tumor suppressor gene between the loci d8s87 and d8s133 in human prostate cancer. Cancer Res 54: 6061-6064.
- Trougakos IP, Gonos ES (2002) Clusterin/apolipoprotein J in human aging and cancer. Int J Biochem Cell Biol 34: 1430-48.
- Trybus TM, Burgess AC, Wojno KJ, Glover TW, Macoska JA (1996) Distinct areas of allelic loss on chromosomal regions 10p and 10q in human prostate cancer. Cancer Res 56: 2263-7.
- Uchida T, Wada C, Wang C, Ishida H, Egawa S, Yokoyama E, Ohtani H, Koshiba K (1995) Microsatellite instability in prostate cancer. Oncogene 10: 1019-22.
- ul Haque MF, King LM, Krakow D, Cantor RM, Rusiniak ME, Swank RT, Superti-Furga A, Haque S, Abbas H, Ahmad W, Ahmad M, Cohn DH (1998) Mutations in orthologous genes in human spondyloepimetaphyseal dysplasia and the brachymorphic mouse. Nat Genet 20: 157-62.
- Van Alewijk DC, Van der Weiden MM, Eussen BJ, Van Den Andel-Thijssen LD, Ehren-van Eekelen CC, Konig JJ, van Steenbrugge GJ, Dinjens WN, Trapman J (1999) Identification of a homozygous deletion at 8p12-21 in a human prostate cancer xenograft. Genes Chromosomes Cancer 24: 119-26.
- Van Den Berg C, Guan XY, Von Hoff D, Jenkins R, Bittner, Griffin C, Kallioniemi O, Visakorpi, McGill, Herath J, et al. (1995) DNA sequence amplification in human prostate cancer identified by chromosome microdissection: potential prognostic implications. Clin Cancer Res 1: 11-8.
- Van Weerden WM, de Ridder CM, Verdaasdonk CL, Romijn JC, van der Kwast TH, Schroder FH, van Steenbrugge GJ (1996) Development of seven new human prostate tumor xenograft models and their histopathological characterization. Am J Pathol 149: 1055-62.
- Vecchione A, Ishii H, Shiao YH, Trapasso F, Rugge M, Tamburrino JF, Murakumo Y, Alder H, Croce CM, Baffa R (2001) Fez1/lzts1 alterations in gastric carcinoma. Clin Cancer Res 7: 1546-52.

- Velickovic M, Delahunt B, McIver B, Grebe SK (2002) Intragenic PTEN/MMAC1 loss of heterozygosity in conventional (clear-cell) renal cell carcinoma is associated with poor patient prognosis. Mod Pathol 15: 479-85.
- Venter JC, Adams MD, Myers EW, Li PW, Mural RJ, Sutton GG, Smith HO, Yandell M, Evans CA, Holt RA, Gocayne JD, Amanatides P, Ballew RM, Huson DH, Wortman JR, Zhang Q, Kodira CD, Zheng XH, Chen L, Skupski M, Subramanian G, Thomas PD, Zhang J, Gabor Miklos GL, Nelson C, Broder S, Clark AG, Nadeau J, McKusick VA, Zinder N, Levine AJ, Roberts RJ, Simon M, Slayman C, Hunkapiller M, Bolanos R, Delcher A, Dew I, Fasulo D, Flanigan M, Florea L, Halpern A, Hannenhalli S, Kravitz S, Levy S, Mobarry C, Reinert K, Remington K, Abu-Threideh J, Beasley E, Biddick K, Bonazzi V, Brandon R, Cargill M, Chandramouliswaran I, Charlab R, Chaturvedi K, Deng Z, Di Francesco V, Dunn P, Eilbeck K, Evangelista C, Gabrielian AE, Gan W, Ge W, Gong F, Gu Z, Guan P, Heiman TJ, Higgins ME, Ji RR, Ke Z, Ketchum KA, Lai Z, Lei Y, Li Z, Li J, Liang Y, Lin X, Lu F, Merkulov GV, Milshina N, Moore HM, Naik AK, Narayan VA, Neelam B, Nusskern D, Rusch DB, Salzberg S, Shao W, Shue B, Sun J, Wang Z, Wang A, Wang X, Wang J, Wei M, Wides R, Xiao C, Yan C, et al. (2001) The sequence of the human genome. Science 291: 1304-51.
- Verhagen PC, Hermans KG, Brok MO, van Weerden WM, Tilanus MG, de Weger RA, Boon TA, Trapman J (2002) Deletion of chromosomal region 6q14-16 in prostate cancer. Int J Cancer 102: 142-7.
- Vesprini D, Nam RK, Trachtenberg J, Jewett MA, Tavtigian SV, Emami M, Ho M, Toi A, Narod SA (2001) HPC2 variants and screen-detected prostate cancer. Am J Hum Genet 68: 912-7.
- Virgin JB, Hurley PM, Nahhas FA, Bebchuk KG, Mohamed AN, Sakr WA, Bright RK, Cher ML (1999) Isochromosome 8q formation is associated with 8p loss of heterozygosity in a prostate cancer cell line. Prostate 41: 49-57
- Visakorpi T, Kallioniemi AH, Syvanen AC, Hyytinen ER, Karhu R, Tammela T, Isola JJ, Kallioniemi OP (1995) Genetic changes in primary and recurrent prostate cancer by comparative genomic hybridization. Cancer Res 55: 342-347
- Vlietstra RJ, van Alewijk DC, Hermans KG, van Steenbrugge GJ, Trapman J (1998) Frequent inactivation of PTEN in prostate cancer cell lines and xenografts. Cancer Res 58: 2720-3
- Vocke CD, Pozzatti RO, Bostwick DG, Florence CD, Jennings SB, Strup SE, Duray PH, Liotta LA, Emmert-Buck MR, Linehan WM (1996) Analysis of 99 microdissected prostate carcinomas reveals a high frequency of allelic loss on chromosome 8p12-21. Cancer Res 56: 2411-6
- Voeller HJ, Augustus M, Madike V, Bova GS, Carter KC, Gelmann EP (1997) Coding region of NKX3.1, a prostate-specific homeobox gene on 8p21, is not mutated in human prostate cancers [published erratum appears in Cancer Res 1997 Dec 15;57(24):5613]. Cancer Res 57: 4455-9
- Vogelstein B, Fearon ER, Kern SE, Hamilton SR, Preisinger AC, Nakamura Y, White R (1989) Allelotype of colorectal carcinomas. Science 244: 207-11
- Wagner U, Bubendorf L, Gasser TC, Moch H, Gorog JP, Richter J, Mihatsch MJ, Waldman FM, Sauter G (1997) Chromosome 8p deletions are associated with invasive tumor growth in urinary bladder cancer. Am J Pathol 151: 753-9.
- Wang DS, Rieger-Christ K, Latini JM, Moinzadeh A, Stoffel J, Pezza JA, Saini K, Libertino JA, Summerhayes IC (2000) Molecular analysis of PTEN and MXI1 in primary bladder carcinoma. Int J Cancer 88: 620-5.
- Wang JC, Radford DM, Holt MS, Helms C, Goate A, Brandt W, Parik M, Phillips NJ, DeSchryver K, Schuh ME, Fair KL, Ritter JH, Marshall P, Donis-Keller H (1999) Sequence-ready contig for the 1.4-cM ductal carcinoma in situ loss of heterozygosity region on chromosome 8p22-p23. Genomics 60: 1-11.
- Wang L, Patel U, Ghosh L, Banerjee S (1992) DNA polymerase beta mutations in human colorectal cancer. Cancer Res 52: 4824-7
- Wang L, McDonnell SK, Elkins DA, Slager SL, Christensen E, Marks AF, Cunningham JM, Peterson BJ, Jacobsen SJ, Cerhan JR, Blute ML, Schaid DJ, Thibodeau SN (2001) Role of HPC2/ELAC2 in hereditary prostate cancer. Cancer Res 61: 6494-9.
- Wang SI, Puc J, Li J, Bruce JN, Cairns P, Sidransky D, Parsons R (1997) Somatic mutations of PTEN in glioblastoma multiforme. Cancer Res 57: 4183-6.
- Wang SI, Parsons R, Ittmann M (1998) Homozygous deletion of the PTEN tumor suppressor gene in a subset of prostate adenocarcinomas. Clin Cancer Res 4: 811-5.
- Webb HD, Hawkins AL, Griffin CA (1996) Cytogenetic abnormalities are frequent in uncultured prostate cancer cells. Cancer Genet Cytogenet 88: 126-32.

- Weber JL, May PE (1989) Abundant class of human DNA polymorphisms which can be typed using the polymerase chain reaction. Am J Hum Genet 44: 388-96.
- Weber B, Riess O, Kreklywich CN, Wood S, Hayden MR (1991) A polymorphic DNA marker at the D8S131 locus. Nucleic Acids Res 19: 1725
- Wechsler DS, Shelly CA, Dang CV (1996) Genomic organization of human MXI1, a putative tumor suppressor gene. Genomics 32: 466-70
- Weinberg RA (1989) Oncogenes, antioncogenes, and the molecular bases of multistep carcinogenesis. Cancer Res 49: 3713-21.
- Whang YE, Wu X, Suzuki H, Reiter RE, Tran C, Vessella RL, Said JW, Isaacs WB, Sawyers CL (1998) Inactivation of the tumor suppressor PTEN/MMAC1 in advanced human prostate cancer through loss of expression. Proc Natl Acad Sci U S A 95: 5246-50.
- Wiklund F, Jonsson BA, Goransson I, Bergh A, Gronberg H (2003) Linkage analysis of prostate cancer susceptibility: confirmation of linkage at 8p22-23. Hum Genet 112: 414-8.
- Wilson MJ (1995) Proteases in prostate development, function, and pathology. Microsc Res Tech 30: 305-18.
- Wistuba, II, Behrens C, Virmani AK, Milchgrub S, Syed S, Lam S, Mackay B, Minna JD, Gazdar AF (1999) Allelic losses at chromosome 8p21-23 are early and frequent events in the pathogenesis of lung cancer. Cancer Res 59: 1973-9.
- Wood RD, Mitchell M, Sgouros J, Lindahl T (2001) Human DNA repair genes. Science 291: 1284-9.
- Wood S, Schertzer M (1992) Linkage mapping of the D8S133 locus to chromosome 8p using a highly informative, polymorphic, complex dinucleotide repeat. Genomics 13: 232.
- Wooster R, Bignell G, Lancaster J, Swift S, Seal S, Mangion J, Collins N, Gregory S, Gumbs C, Micklem G (1995) Identification of the breast cancer susceptibility gene BRCA2 [see comments]. Nature 378: 789-92
- Wright K, Wilson PJ, Kerr J, Do K, Hurst T, Khoo SK, Ward B, Chenevix-Trench G (1998) Frequent loss of heterozygosity and three critical regions on the short arm of chromosome 8 in ovarian adenocarcinomas. Oncogene 17: 1185-8
- Wu CL, Roz L, Sloan P, Read AP, Holland S, Porter S, Scully C, Speight PM, Thakker N (1997) Deletion mapping defines three discrete areas of allelic imbalance on chromosome arm 8p in oral and oropharyngeal squamous cell carcinomas. Genes Chromosomes Cancer 20: 347-53
- Wu W, Kemp BL, Proctor ML, Gazdar AF, Minna JD, Hong WK, Mao L (1999) Expression of DMBT1, a candidate tumor suppressor gene, is frequently lost in lung cancer. Cancer Res 59: 1846-51
- Wu X, Senechal K, Neshat MS, Whang YE, Sawyers CL (1998) The PTEN/MMAC1 tumor suppressor phosphatase functions as a negative regulator of the phosphoinositide 3-kinase/Akt pathway. Proc Natl Acad Sci U S A 95: 15587-91.
- Xu J, Meyers D, Freije D, Isaacs S, Wiley K, Nusskern D, Ewing C, Wilkens E, Bujnovszky P, Bova GS, Walsh P, Isaacs W, Schleutker J, Matikainen M, Tammela T, Visakorpi T, Kallioniemi OP, Berry R, Schaid D, French A, McDonnell S, Schroeder J, Blute M, Thibodeau S, Trent J, et al. (1998) Evidence for a prostate cancer susceptibility locus on the X chromosome. Nat Genet 20: 175-9.
- Xu J (2000) Combined analysis of hereditary prostate cancer linkage to 1q24-25: results from 772 hereditary prostate cancer families from the International Consortium for Prostate Cancer Genetics. Am J Hum Genet 66: 945-57.
- Xu J, Zheng SL, Carpten JD, Nupponen NN, Robbins CM, Mestre J, Moses TY, Faith DA, Kelly BD, Isaacs SD, Wiley KE, Ewing CM, Bujnovszky P, Chang B, Bailey-Wilson J, Bleecker ER, Walsh PC, Trent JM, Meyers DA, Isaacs WB (2001a) Evaluation of linkage and association of HPC2/ELAC2 in patients with familial or sporadic prostate cancer. Am J Hum Genet 68: 901-11.
- Xu J, Zheng SL, Chang B, Smith JR, Carpten JD, Stine OC, Isaacs SD, Wiley KE, Henning L, Ewing C, Bujnovszky P, Bleeker ER, Walsh PC, Trent JM, Meyers DA, Isaacs WB (2001b) Linkage of prostate cancer susceptibility loci to chromosome 1. Hum Genet 108: 335-45.
- Xu J, Zheng SL, Hawkins GA, Faith DA, Kelly B, Isaacs SD, Wiley KE, Chang B, Ewing CM, Bujnovszky P, Carpten JD, Bleecker ER, Walsh PC, Trent JM, Meyers DA, Isaacs WB (2001c) Linkage and association studies of prostate cancer susceptibility: evidence for linkage at 8p22-23. Am J Hum Genet 69: 341-50.
- Xu J, Zheng SL, Komiya A, Mychaleckyj JC, Isaacs SD, Hu JJ, Sterling D, Lange EM, Hawkins GA, Turner A, Ewing CM, Faith DA, Johnson JR, Suzuki H, Bujnovszky P, Wiley KE, DeMarzo AM, Bova GS, Chang B, Hall MC, McCullough DL, Partin AW, Kassabian VS, Carpten JD, Bailey-Wilson JE, Trent JM, Ohar J, Bleecker ER, Walsh PC, Isaacs WB, Meyers DA (2002a) Germline mutations and sequence variants of the macrophage scavenger receptor 1 gene are associated with prostate cancer risk. Nat Genet 32: 321-5.

- Xu J, Zheng SL, Komiya A, Mychaleckyj JC, Isaacs SD, Chang B, Turner AR, Ewing CM, Wiley KE, Hawkins GA, Bleecker ER, Walsh PC, Meyers DA, Isaacs WB (2003) Common sequence variants of the macrophage scavenger receptor 1 gene are associated with prostate cancer risk. Am J Hum Genet 72: 208-12.
- Xu LL, Srikantan V, Sesterhenn IA, Augustus M, Dean R, Moul JW, Carter KC, Srivastava S (2000) Expression profile of an androgen regulated prostate specific homeobox gene NKX3.1 in primary prostate cancer. J Urol 163: 972-9.
- Xu ZH, Freimuth RR, Eckloff B, Wieben E, Weinshilboum RM (2002b) Human 3'-phosphoadenosine 5'phosphosulfate synthetase 2 (PAPSS2) pharmacogenetics: gene resequencing, genetic polymorphisms and functional characterization of variant allozymes. Pharmacogenetics 12: 11-21.
- Yang G, Ayala G, Marzo AD, Tian W, Frolov A, Wheeler TM, Thompson TC, Harper JW (2002) Elevated Skp2 protein expression in human prostate cancer: association with loss of the cyclin-dependent kinase inhibitor p27 and PTEN and with reduced recurrence-free survival. Clin Cancer Res 8: 3419-26.
- Yaremko ML, Kutza C, Lyzak J, Mick R, Recant WM, Westbrook CA (1996) Loss of heterozygosity from the short arm of chromosome 8 is associated with invasive behavior in breast cancer. Genes Chromosomes Cancer 16: 189-95.
- Ye L, Miki T, Nakura J, Oshima J, Kamino K, Rakugi H, Ikegami H, Higaki J, Edland SD, Martin GM, Ogihara T (1997) Association of a polymorphic variant of the Werner helicase gene with myocardial infarction in a Japanese population [published erratum appears in Am J Med Genet 1997 May 2;70(1):103]. Am J Med Genet 68: 494-8
- You MJ, Castrillon DH, Bastian BC, O'Hagan RC, Bosenberg MW, Parsons R, Chin L, DePinho RA (2002) Genetic analysis of Pten and Ink4a/Arf interactions in the suppression of tumorigenesis in mice. Proc Natl Acad Sci U S A 99: 1455-60.
- Yu CE, Oshima J, Fu YH, Wijsman EM, Hisama F, Alisch R, Matthews S, Nakura J, Miki T, Ouais S, Martin GM, Mulligan J, Schellenberg GD (1996a) Positional cloning of the Werner's syndrome gene [see comments]. Science 272: 258-62
- Yu CE, Oshima J, Hisama FM, Matthews S, Trask BJ, Schellenberg GD (1996b) A YAC, P1, and cosmid contig and 17 new polymorphic markers for the Werner syndrome region at 8p12-p21. Genomics 35: 431-40
- Yu CE, Oshima J, Wijsman EM, Nakura J, Miki T, Piussan C, Matthews S, Fu YH, Mulligan J, Martin GM, Schellenberg GD (1997) Mutations in the consensus helicase domains of the Werner syndrome gene. Werner's Syndrome Collaborative Group. Am J Hum Genet 60: 330-41
- Yuan XJ, Whang YE (2002) PTEN sensitizes prostate cancer cells to death receptor-mediated and druginduced apoptosis through a FADD-dependent pathway. Oncogene 21: 319-27.
- Yustein AS, Harper JC, Petroni GR, Cummings OW, Moskaluk CA, Powell SM (1999) Allelotype of gastric adenocarcinoma. Cancer Res 59: 1437-41
- Zheng SL, Xu J, Isaacs SD, Wiley K, Chang B, Bleecker ER, Walsh PC, Trent JM, Meyers DA, Isaacs WB (2001) Evidence for a prostate cancer linkage to chromosome 20 in 159 hereditary prostate cancer families. Hum Genet 108: 430-5.
- Zitzelsberger H, Engert D, Walch A, Kulka U, Aubele M, Hofler H, Bauchinger M, Werner M (2001) Chromosomal changes during development and progression of prostate adenocarcinomas. Br J Cancer 84: 202-8.

SUMMARY / SAMENVATTING



SUMMARY

Prostate cancer (PrCa) is the most frequently diagnosed cancer in men, and the second leading cause of male cancer death in Western countries (Chapter 1-1). In order to design novel targeted therapies, knowledge of the molecular mechanisms underlying PrCa growth are urgently needed. Like other cancers, PrCa is believed to result from genomic instability, leading to mutation, deletion, amplification and translocation of a large variety of genes that regulate cell growth and survival. Chromosomal loss is an indication for the localization of a tumour suppressor gene, whereas amplification and translocation are an indication for the localization of an oncogene (Chapter 1-2). One of the most frequent genetic abnormalities in PrCa is loss of the complete, or part of the short arm of chromosome 8. Loss of the long arm of chromosome 10 is another frequently observed genetic defect in PrCa (Chapter 1-3). The aim of this thesis is to further characterise the regions of frequent loss at chromosome 10q and 8p, and to identify the tumour suppressor genes located in these regions (Chapter 1-4).

This study mainly used a set of PrCa derived xenografts (11) and cell lines (4). Human tumours, propagated as xenografts in nude mice, or as *in vitro* growing cell lines are unique sources of tumour DNA that aid the identification of genetic defects in human cancer. They are available in unlimited quantities. Importantly, they lack normal cells of human origin, which increases the detection of chromosomal alterations, homozygous deletions and point mutations in individual genes and their expression level. An additional set of 43 prostate tumour samples was used for chromosome 8p gene mutation analyses.

The set of xenografts and cell lines was searched for chromosome 10 alterations by Comparative Genomic Hybridisation and allelotyping (Chapter 3). CGH revealed a pattern of loss of 10q in six out of fifteen samples, which were confirmed by allelotyping. These were all large deletions. In another seven samples, allelotyping revealed small regions of loss of 10q23.3, in particular. Recently, the *PTEN* tumour suppressor gene was identified and mapped to this chromosomal locus. A *PTEN* structure and expression analysis of the xenografts and cell lines demonstrated five homozygous deletions, two nonsense mutations, and two frame-shift mutations (Chapter 2). Except for the samples with a large homozygous deletion, no aberrant *PTEN* mRNA expression was found. This

indicates that expressional down-regulation is no important mechanism of PTEN inactivation. The high percentage (60%) of deletions and mutations implicates a significant role for PTEN in the pathogenesis of PrCa. Nevertheless, not all losses of 10q23.3 were found in combination with a PTEN inactivation. Four samples with loss of 10q showed an intact PTEN gene (Chapter 3). Apart from haplo-insufficiency, which has been assigned as a mechanism of PTEN inactivation, another 10q tumour suppressor gene may also be involved. We therefore analysed the deletion and down-regulation of expression of genes flanking the *PTEN* locus in both PTEN positive and negative samples (Chapter 3). Candidate gene FLJ11218 appeared particularly interesting because of its differential expression in several samples and a missense mutation in one sample, and because it accompanied all losses and homozygous deletions of PTEN. A search for tumour suppressor genes on distal 10p and 10q did not reveal alteration of expression or structure of the KLF6 and MXI genes. The DMBT1 gene at 10q was deleted in one xenograft and down-regulated in two.

The set of xenografts was also searched for chromosome 8 alterations by Comparative Genomic Hybridisation (CGH) and allelotyping (Chapter 5). CGH showed 8p loss in ten DNAs. In eight of the samples 8p loss included the 8p12-p21 interval. In most DNAs allelotyping matched with the CGH data. Lack of a previously identified tumour suppressor gene on 8p forced us to fine map the position of a tumour suppressor gene. In the past, the search for a homozygous deletion has been pivotal in the identification of a number of tumour suppressor genes, including BRCA2, DPC4, and PTEN. We searched for homozygous deletions in the 8p12-p21 interval by using known and novel polymorphic and nonpolymorphic sequence tagged sites (STSs) between the markers D8S87 and D8S133. The interest for this specific region arose from our previous results, showing allelic loss in 69% of the PrCa samples. In xenograft PC133, the presence of a small homozygous deleted region of 730-1320 Kb was unambiguously established (Chapter 4). At one site, the deletion was found to disrupt the major part of the Werner Syndrome gene (WRN). To facilitate the identification and isolation of candidate tumour suppressor genes in this area, we generated a contiguous PAC/cosmid contig (Chapter 5). Thirty-seven STSs were localized in this map that were subsequently used to screen the panel of xenografts and cell lines for an overlapping homozygous deletion. No additional homozygous deletion was found. Exon trapping, database analysis of sequenced genomic DNA, or expressed sequence tag (EST) characterization did not reveal a novel bona fide gene other than

the WRN gene. A single EST, sts-N22494, was identified as part of the 3'-UTR of a longer WRN transcript. To test WRN as a candidate tumour suppressor gene in PrCa, a mutation and expression analysis was performed. Except for PC133, no alteration in WRN was found. Although absence of WRN may have contributed to PC133 tumour growth, no further evidence was found for a role of WRN in PrCa tumour development. This suggests another tumour suppressor gene on 8p12-p21. To facilitate the identification and isolation of candidate tumour suppressor genes in this area, we isolated the BC028701 - LPL segment, which completely contains the D8S87-D8S133 region, in a single contig of approximately 17 Mbp, composed of 172 YACs (Appendix). We then ordered 303 markers within this contig, including polymorphic sequence tagged sites (STSs), nonpolymorphic STSs, expressed sequence tags (ESTs), genes, and pseudogenes. The availability of this robust physical and transcript map facilitated and justified an extensive search for a homozygous deletion on 8p12-p21 (Chapter 6). Again no homozygous deletion was found. We therefore continued our search for tumour suppressor genes by analysis of positional and functional candidate genes at 8p. An extensive mutation and expression analysis of the LZTS1, PRLTS, NKX3-1, EPHX2, CLU and POLB genes was performed. We did not find convincing evidence that any of these genes are frequently bi-allelic inactivated in human PrCa, although the EPHX2 gene may require further analysis.

In conclusion, except for *PTEN* not a single gene on chromosome 8p or elsewhere on 10q was found to be frequently inactivated due to a double hit deletion and or mutation (Chapter 7). Haplo-insufficiency as an additional mechanism of gene inactivation and its possible implications for PrCa tumorigenesis are discussed in a larger context.

SAMENVATTING

Prostaatkanker is de meest voorkomende vorm van kanker bij mannen in Westerse landen, en de op een na meest frequente oorzaak van overlijden aan kanker bij mannen (Hoofdstuk 1.1). Om nieuwe therapieën te kunnen ontwikkelen is kennis van de moleculaire mechanismen die ten grondslag liggen aan het ontstaan en de groei van de tumor noodzakelijk. Net zoals bij andere tumoren, wordt gedacht dat prostaatkanker het gevolg is van instabiliteit van het genoom. Deze instabiliteit resulteert in mutatie, deletie, amplificatie en translocatie van een grote variëteit aan genen die betrokken zijn bij de regulering van celgroei en celdood. Verlies, amplificatie en translocatie van specifieke delen van chromosomen zijn karakteristieken van tumorcellen. Verlies is een aanwijzing voor de locatie van een gen dat de groei van een tumor remt, oftewel een tumorsuppressorgen. Een amplificatie of translocatie is daarentegen een aanwijzing voor de locatie van een gen dat de groei van een tumor stimuleert, oftewel een oncogen (Hoofdstuk 1.2). Een van de meest voorkomende afwijkingen van het genoom in prostaatkanker is geheel of gedeeltelijk verlies van de korte arm van chromosoom 8. Verlies van de lange arm van chromosoom 10 is eveneens een veel voorkomend defect van het genoom in prostaatkanker (Hoofdstuk 1.3). Doel van dit proefschrift is de karakterisering van de delen van chromosoom 8p en 10q, die het meest frequent verloren gaan, en de identificatie van de tumorsuppressorgenen in deze regio's (Hoofdstuk 1.4).

In deze studie is voornamelijk gebruik gemaakt van een set xenograften (11) en cellijnen (4) die zijn afgeleid van prostaattumoren van de mens. Xenograften zijn tumoren, die als een transplantaat groeien in muizen met een verzwakt immuunsysteem (naakte muizen). Xenograften en *in vitro* groeiende cellijnen zijn unieke bronnen van tumor-DNA, die een belangrijk hulpmiddel zijn bij de identificatie van veranderingen in het genoom van humane tumoren. Ze zijn in principe in onbeperkte hoeveelheden beschikbaar. Nog belangrijker is dat ze, in tegenstelling tot direct van de mens afkomstig tumorweefsel, geheel vrij zijn van normale cellen van humane oorsprong zoals de cellen van steun- en vaatweefsel. Dit vergroot de kans op het vinden van veranderingen in chromosomen en het detecteren van homozygote deleties en puntmutaties in individuele genen. Om dezelfde reden kan genexpressie beter bestudeerd worden. Voor de analyse van mutaties in genen op chromosoom 8p is tevens gebruik gemaakt van weefsel van 43 prostaattumoren.

De xenograften en cellijnen zijn onderzocht op afwijkingen van chromosoom 10 door middel van CGH (Comparative Genomic Hybridization) en allelotypering (Hoofdstuk 3). In zes van de vijftien monsters werd verlies van 10g aangetoond. Dit werd bevestigd met allelotypering. Daarnaast werden met allelotypering in zeven andere monsters verliezen van een klein deel van de chromosomale band 10g23.3 gevonden. Hierin is het PTEN tumorsuppressorgen gelegen. Analyse van de structuur en expressie van PTEN toonde vijf homozygote deleties aan, twee nonsense-mutaties en twee frame-shift-mutaties (Hoofdstuk 2). Behalve in monsters met een grote homozygote deletie werd geen afwijking in de expressie van PTEN gevonden. Dit is een aanwijzing dat vermindering van expressie geen belangrijk mechanisme van inactivering van PTEN is. Het hoge percentage (60%) deleties en mutaties suggereert nochtans een significante rol voor PTEN bij prostaatkanker. Vier DNA's met verlies van 10g bezaten een intact tweede kopie van het PTEN-gen (Hoofdstuk 3). Mogelijk kan verlies van één kopie van PTEN al een rol spelen bij de groei van de tumor. Dit staat bekend als haplo-insufficientie. Daarnaast is het mogelijk dat een ander tumorsuppressorgen op 10g een rol speelt in prostaatkanker. Genen die het PTENgen flankeren zijn daarom onderzocht op deleties en verlaagde expressie. Hiervoor zijn zowel PTEN-positieve als -negatieve samples gebruikt (Hoofdstuk 3). Met name kandidaatgen FLJ11218 lijkt interessant omdat het betrokken is bij homozygote deleties en het differentieel tot expressie komt in normaal en tumorweefsel. Bovendien is een missense-mutatie aangetoond in het DNA van een prostaat carcinoom. Andere kandidaten zijn MINPP1 en PAPSS2. Onderzoek naar de eerder beschreven kandidaat-tumorsuppressorgenen KLF6 en MXI1, gelegen op respectievelijk het uiteinde van chromosoom 10p en 10q, heeft geen afwijkingen laten zien. Het DMBT1gen op 10q bevat in een xenograft een intragenetische homozygote deletie. Daarnaast is de expressie van dit gen laag in twee xenograften.

De xenograften en cellijnen zijn eveneens onderzocht op afwijkingen van chromosoom 8 door middel van CGH en allelotypering (Hoofdstuk 5). In het DNA van tien verschillende tumoren/cellijnen werd verlies van 8p aangetoond met CGH, waarbij in acht gevallen het gebied 8p12-p21 verloren was gegaan. In de meeste gevallen kwamen de resultaten van allelotypering en CGH overeen. Het ontbreken van een eerder geïdentificeerd tumorsuppressorgen op 8p noodzaakte ons de positie van zo'n gen te bepalen. In het verleden zijn homozygote deleties cruciaal gebleken voor de

identificatie van tumorsuppressorgenen zoals BRCA2, DPC4 en PTEN. Daarom werd in het gebied 8p12-p21, met behulp van polymorfe en niet polymorfe markers (STSs). gezocht naar homozygote deleties. Specifiek is gezocht tussen de markers D8S87 en D8S133. De interesse voor dit deel van 8p kwam voort uit eerder onderzoek waarin frequent verlies (69%) van dit gebied werd aangetoond in prostaattumoren. In xenograft PC133 werd een kleine homozygote deletie (730 - 1320 Kb) vastgesteld (Hoofdstuk 4). Deze deletie omvatte een deel van het Werner Syndroom-gen (WRN). Om andere kandidaat-tumorsuppressorgenen in dit gebied te kunnen ontdekken, werd hiervan een PAC/cosmide contig gegenereerd (Hoofdstuk 5). In het gedeleteerde gebied werden 37 markers (STSs) gelokaliseerd. Deze werden vervolgens gebruikt bij het onderzoek op kleine homozygote deleties in de andere xenograften en cellijnen. Dit was zonder succes. Exon trapping en onderzoek van de sequentie van het genomische DNA en analyse van fragmenten van onbekende genen (ESTs), leverde alleen het WRN-gen op. Een EST, sts-N22494, bleek een deel van het 3'UTR van een langer WRN transcript te zijn. Om te testen of het WRN-gen een kandidaat tumorsuppressorgen is in prostaatkanker, werd een analyse van mutatie en expressie van dit gen uitgevoerd. Behalve bij PC133 werd er geen afwijking in WRN gevonden. Hoewel afwezigheid van WRN een bijdrage kan hebben geleverd aan de groei van PC133, zij geen nieuwe argumenten gevonden voor een rol van WRN in prostaatkanker. Dit suggereert dat één of meer andere genen op 8p12p21 hierbij van belang zijn. Om de identificatie en isolatie van kandidaattumorsuppressorgenen te vergemakkelijken is het gebied tussen BC028701 en LPL geïsoleerd in een YAC contig (Appendix). Dit gebied bevat de gehele regio tussen D8S87 en D8S133, in een contig van 172 YACs met een omvang van 17 Mbp (Appendix A6). In dit contig konden 303 markers worden geordend, waaronder polymorfe markers, STSs, ESTs, genen en pseudogenen. Deze gedetailleerde fysische- en transcriptie-kaart maakte het vervolgens mogelijk en verantwoord chromosoom 8p12-p21 in detail te screenen op homozygote deleties (Hoofdstuk 6). Wederom werd geen homozygote deletie gevonden. Het zoeken naar tumorsuppressorgenen is daarom voortgezet middels mutatie- en expressieanalyse van de 8p kandidaatgenen LZTS1, PRLTS, NKX3-1, EPHX2, CLU en POLB. Behalve vanwege hun locatie werden deze genen geselecteerd omdat het biologisch plausibel is dat ze een rol spelen in prostaatkanker. We vonden voor geen van deze genen overtuigend bewijs dat zij frequent bi-allelisch geinactiveerd zijn in prostaatkanker. Alleen aan *EPHX2* is verder onderzoek gerechtvaardigd.

In conclusie, behalve *PTEN*, is op 8p en 10q geen enkel gen gevonden dat frequent is gedeleteerd en / of gemuteerd in beide allelen (Hoofdstuk 7). Gezien deze negatieve bevindingen moet worden overwogen dat in prostaatkanker haploinsufficiency vaker een rol speelt dan tot nu toe werd aangenomen.

DANKWOORD

Het volbrengen van dit promotie-onderzoek is voor mij een uiterst leerzame en meer dan een karaktervormende ervaring geworden. De onvermijdelijke tegenslagen, waarbij je niet alleen met domme pech maar ook nogal eens met je eigen tekortkomingen en zwakheden wordt geconfronteerd, hebben mij er soms bijna toe gebracht definitief een punt achter mijn promotiestudie te zetten. Er is tenslotte meer in het leven. Ik heb echter de afgelopen jaren van een groot aantal mensen bijzondere bijstand mogen ontvangen, die ik hiervoor wil bedanken.

Jan, allereerst ben ik jou dank verschuldigd voor het vertrouwen dat je in me had dat ik dit meest risicovolle (en competitieve) promotieonderzoek dat je ooit gehonoreerd hebt gekregen tot een goed einde zou te brengen. Ons beider streven naar perfectie heeft weliswaar veel tijd gekost maar heeft uiteindelijk wel geresulteerd in dit proefschrift. Je wetenschappelijke kennis en inzicht, je ijzeren geheugen en je bereidheid ondanks een overvolle agenda steeds weer tijd voor ook mij reserveren (zelfs in weekends), hebben hier aan bijgedragen.

Wolter, bedankt voor je luisterend oor in de meest letterlijke en figuurlijke zin. Ons overleg was meestal telefonisch maar ik heb je wel leren kennen als een integer persoon die vaak organisatorisch de juiste oplossing wist aan te dragen.

Hein en Ageeth, ben ik (en dan spreek ik ook namens Lianne) dankbaar voor hun vriendschap. Uit het feit dat jullie vandaag mijn paranimfen zijn, blijkt wel dat jullie je rol als getuige van ons huwelijk meer dan waard zijn geweest. Zeker de laatste paar jaar had ik veel steun aan het urenlang "beppen", filmje pakken, biertje happen en salsa dansen.

De leden van de promotiecommissie: Prof.dr.ir. J. Trapman, Prof.dr. J.W. Oosterhuis, Prof.dr. Th.H. van der Kwast, Prof.dr. C.H. Bangma, Prof.dr. C.J. Cornelisse, Prof.dr. R. Fodde, Prof.dr. B.A. Oostra, Prof.dr. F.T. Bosman en Dr. M. Schutte, dank ik voor hun interesse in het manuscript en snelle beoordeling.

Lydia en Conny, of moet ik zeggen Conny en Lydia? Zonder jullie enorme inzet was ik nooit tot dit resultaat gekomen. De honderden PCRs, gelen, cloneringen, sequentie-reacties, SSCPs, blots, weefselkweken, hybridizaties, en updates van databases, leken ons aanvankelijk niet te deren en hebben vele resultaten opgeleverd. Echter, het uitblijven van nog een (overlappende) homozygote deletie of nieuw kandidaatgen werd ook voor jullie een teleurstelling. Zoals nu blijkt, is er dus een verklaring voor. Ook Phebe, die een jaar lang lief een leed met ons heeft willen delen, wil ik bedanken voor haar bijdrage. Je werk heeft geresulteerd in een interessante stelling en wie weet, krijgt de studie naar de rol van EPHX2 in prostaatkanker nog een doorstart.

Karin, wanneer ga jij een promotieonderzoek doen? Je kunt het, want je expertise spreekt boekdelen (dit proefschrift). Bedankt ook voor je experimenten voor dit proefschrift na mijn vertrek, en het tot vervelens toe mailen van figuren en artikelen. Remco, ook jou wil ik bedanken voor de leuke samenwerking aan het *PTEN*-gen.

De collega's van de afdelingen Pathologie (3^e èn 2^e), Urologie, Endocrinologie & Voortplanting en Klinische Genetica wil ik bedanken voor hun plezierige samenwerking, hulp en goede raad. Jullie verder bedankt voor: de wetenschappelijke visie, het "painten" en "fishen", de assistentie bij CGH analyses, het ontwikkelen van foto's, de figuren en artikelen, het gebruik van m'n PAC filters, het achterhalen van (E-mail)adressen en telefoonnummers, het ontvangstcomité op zondagavonden, de theologische verdieping, het optimisme, de gezelligheid, het overbodig maken van babbelboxen, de pingpong-lessen, het te pas en te onpas extra dimensioneren van teksten en foto's, het leren waarderen van de meest vage doedelzakmuziek en de lessen alpenhoorn blazen op het frame van de kar van het stikstofvat.

De collega's van het NKI, bedank ik voor het delen van hun kennis, ervaring, fanatisme, humor, scherpzinnigheid, het leren maken, hybridiseren en analyseren van micro-arrays en het mee mogen opzetten van de NKI / AvL - KWF micro-array faciliteit.

I would like to thank my colleagues at PamGene for their personal interest and trust. It's great to be part of such a team of pioneers. Tim and Theo, thanks for the opportunity to exploit my creativity and the freedom and extra time to finalise this thesis. The training "context – content" has been of great value.

Vrienden, een promotie gaat ten koste van je sociale leven. Dat hebben jullie moeten ervaren. Excuus daarvoor, en dank voor jullie persoonlijke interesse en de soms pijnlijke vraag, "Hoe gaat het met je promotie?" Er wordt beweerd dat je na een promotie in een gat valt. Kan ik nog op jullie rekenen dit gat te vullen? Hein, Ilonka, en Nicole, hopelijk volgen er nog heel veel culinaire "HLO avondjes" (friet mag ook). Wanneer en bij wie is de volgende avond? Wie mailt mij om te voorkomen dat de kok en zijn/haar eega nog een week hetzelfde eten?

De mensen achter de schermen, waar jullie ook zitten of zaten, bedankt voor alles waar ik geen weet van heb (misschien maar goed ook). Toch wil ik enkele personen graag nog noemen. Mieke, bedankt voor de je verhelderende wetenschappelijke discussies en je trap tegen de parkeermeter. Alle kwartjes zijn gevallen. Michael thanks for your attempt in keeping me from writing English that has gone Dutch.

Mijn familie & schoonfamilie wil ik natuurlijk bedanken voor hun geduld, begrip, opvangen, bijspringen, aandacht, steun en interesse. Lieve papa en mama, bedankt voor jullie liefde en goede zorgen. De afronding van deze fase in mijn leven is ook voor jullie een hele zorg minder. Geniet maar van je (klein)kinderen en vrije tijd.

Anniek en Marlijn, mijn vrouwen, een hele dikke knuffel voor het leren uit- en aankleden van jullie poppen, het kliederen met vingerverf, het springen op de bank en jullie steeds weer verrassende kijk op de wereld om je heen. Hopelijk heeft papa nu veel meer tijd om te spelen, te stoeien en om verhaaltjes voor te lezen. Lianne, mijn liefde, van diegene die je het meest dierbaar is verwacht je, ten onrechte, de grootste flexibiliteit.

Zon op al jullie wegen,

Diek
LIST OF PUBLICATIONS

- Chang GT, Steenbeek M, Schippers E, Blok LJ, van Weerden WM, van Alewijk DC, Eussen BH, van Steenbrugge GJ, Brinkmann AO (2001) A novel gene on human chromosome 2p24 is differentially expressed between androgen-dependent and androgen-independent prostate cancer cells. Eur J Cancer 37: 2129-34.
- Chang GT, Steenbeek M, Schippers E, Blok LJ, van Weerden WM, van Alewijk DC, Eussen BH, van Steenbrugge GJ, Brinkmann AO (2000) Characterization of a zinc-finger protein and its association with apoptosis in prostate cancer cells. J Natl Cancer Inst 92: 1414-21.
- Chang GT, Tapsi N, Steenbeek M, Blok LJ, van Weerden WM, van Alewijk DC, Eussen BH, van Steenbrugge GJ, Brinkmann AO (1999) Identification of a gene on human chromosome 8q11 that is differentially expressed during prostate-cancer progression. Int J Cancer 83: 506-11.
- Van Alewijk DC, Van der Weiden MM, Eussen BJ, Van Den Andel-Thijssen LD, Ehrenvan Eekelen CC, Konig JJ, van Steenbrugge GJ, Dinjens WN, Trapman J (1999)
 Identification of a homozygous deletion at 8p12-21 in a human prostate cancer xenograft. Genes Chromosomes Cancer 24: 119-26.
- Vlietstra RJ, van Alewijk DC, Hermans KG, van Steenbrugge GJ, Trapman J (1998) Frequent inactivation of PTEN in prostate cancer cell lines and xenografts. Cancer Res 58: 2720-3.
- Adelaide J, Chaffanet M, Imbert A, Allione F, Geneix J, Popovici C, van Alewijk D, Trapman J, Zeillinger R, Borresen-Dale AL, Lidereau R, Birnbaum D, Pebusque MJ (1998) Chromosome region 8p11-p21: refined mapping and molecular alterations in breast cancer. Genes Chromosomes Cancer 22: 186-99.

- Steeghs K, Peters W, Bruckwilder M, Croes H, Van Alewijk D, Wieringa B (1995) Mouse ubiquitous mitochondrial creatine kinase: gene organization and consequences from inactivation in mouse embryonic stem cells. DNA Cell Biol 14: 539-53.
- Afink GB, Van Alewijk DC, De Roos AD, Van Zoelen EJ (1994) Lysophosphatidic acid and bradykinin have opposite effects on phenotypic transformation of normal rat kidney cells. J Cell Biochem 56: 480-9.
- Lahaye DH, Afink GB, Bleijs DA, Van Alewijk DC, Van Zoelen EJ (1994) Effect of bradykinin on loss of density-dependent growth inhibition of normal rat kidney cells. Cell Mol Biol (Noisy-le-grand) 40: 717-21.
- Mooney N, Van Alewyk D, Brick-Ghannan C, Charron D (1991) HLA class-II antigenmediated induction of a proliferative response to anti-IgM in human B lymphocytes. Int J Cancer Suppl 6: 30-3.

CURRICULUM VITAE

9 november 1966	Geboren te Tilburg
1984	Eindexamen HAVO aan de Rijks Scholen Gemeenschap Willem II te Tilburg
1984 - 1986	VWO aan de Rijks Scholen Gemeenschap Willem II te Tilburg
1986 - 1990	Dr. Struycken Instituut – Hogeschool West Brabant te Etten-Leur, HLO-Biochemie
	Stage: Laboratoire d'Immunogénetique Moléculaire, Institut des Cordeliers, Université Paris VI, begeleiders Dr. N. Mooney en Prof.dr. D. Charron
1990 - 1991	Katholieke Universiteit Nijmegen, Faculteit der Scheikunde
1991 - 1994	Katholieke Universiteit Nijmegen, Faculteit der Biologie
	1 ^e hoofdvakstage: Afdeling Moleculaire biologie en Celbiologie, Fa- culteit der Natuurwetenschappen, Katholieke Universiteit Nijmegen, begeleiders Dr. G. Afink en Prof.dr. E.J.J. van Zoelen
	2 ^e hoofdvakstage: Afdeling Celbiologie en Histologie, Faculteit der Geneeskunde en Tandheelkunde, Katholieke Universiteit Nijmegen, begeleiders Dr. K. Steegs en Prof.dr. B. Wieringa
1994 - 2000	Promotie onderzoek, Afdeling Pathologie (Josephine Nefkens Instituut), Faculteit der Geneeskunde en Gezondheids- wetenschappen, Erasmus Universiteit Rotterdam. Promotoren: Prof.dr.ir. J. Trapman en Prof.dr. J.W. Oosterhuis
2000 - 2002	DNA microarray project coördinator, Centrale Microarray Faciliteit van het NKI / AvL - KWF te Amsterdam.
2002 - heden	Manufacturing Manager, PamGene International BV. te 's-Hertogenbosch.